# SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

# **SCHEDULE 13D**

# (RULE 13d-101)

# **UNDER THE SECURITIES EXCHANGE ACT OF 1934**

# Western Digital Corporation

(Name of Issuer)

Common Stock, par value \$0.01 per share

(Title of Class of Securities)

958102105

(CUSIP Number)

Taro Kaiho Hitachi, Ltd. 6-6, Marunouchi 1-chome, Chiyoda-ku, Tokyo, 100-8280, Japan Telephone: 81-3-3258-1111

(Name, Address and Telephone Number of Person Authorized to Receive Notices and Communications)

March 8, 2012

(Date of Event Which Requires Filing of This Statement)

If the filing person has previously filed a statement on Schedule 13G to report the acquisition that is the subject of this Schedule 13D, and is filing this schedule because of §§240.13d–1(e), 240.13d–1(f) or 240.13d–1(g), check the following box o

CUSIP No. 958102105

13D

1.	Names of Reporting Pe	orcone		
1.				
2	Hitachi, Ltd.			
2.	Check the Appropriate Box if a Member of a Group ( <i>See</i> Instructions)			
				(a) o (b) o
3.	SEC Use Only			
4.	Source of Funds (See Instructions)			
	00			
5.				0
6.	Citizenship or place of organization			
	Japan			
		7.	Sole Voting Power	
			25,000,000	
		8.	Shared Voting Power	
	Number of Shares Beneficially Owned by		0	
	Each Reporting Person With	9.	Sole Dispositive Power	
			25,000,000	
		10.	Shared Dispositive Power	
			0	
11.	Aggregate Amount Beneficially Owned by Each Reporting Person			
	25,000,000			
12.	Check if the Aggregate	e Amou	nt in Row (11) Excludes Certain Shares ( <i>See</i> Instructions)	0
13.	Percent of Class Repre	sented	by Amount in Row (11)	
	9.6%			
14.	Type of Reporting Pers	son (See	e Instructions)	
	со			

### Item 1. Security and Issuer.

This statement on Schedule 13D (this "<u>Statement</u>") relates to the shares of common stock, par value \$0.01 per share, of Western Digital Corporation, a Delaware corporation (such common stock, the "<u>Shares</u>"; such corporation, the "<u>Issuer</u>"). The Issuer's principal executive offices are located at 3355 Michelson Drive, Suite 100, Irvine, California 92612.

### Item 2. Identity and Background.

This Statement is being filed by Hitachi, Ltd., a joint stock corporation incorporated under the laws of Japan (the "<u>Reporting Person</u>"). The principal business of the Reporting Person is manufacturing and offering electrical and electronic products ranging from electricity generation systems to consumer products and electronic devices.

The address of the principal office and principal business of the Reporting Person is 6-6, Marunouchi 1-chome, Chiyoda-ku, Tokyo, 100-8280, Japan.

<u>Schedule 1</u>, which is attached hereto and incorporated herein by reference, sets forth the following information with respect to each executive officer and director of the Reporting Person: the person's (a) name, (b) business address and (c) present principal occupation or employment and the name, principal business and address of any corporation or other organization in which such employment is conducted.

Neither the Reporting Person nor, to the knowledge of the Reporting Person, any of its executive officers or directors listed in <u>Schedule 1</u> has, during the last five years, been convicted in a criminal proceeding (excluding traffic violations or similar misdemeanors).

Neither the Reporting Person nor, to the knowledge of the Reporting Person, any of its executive officers or directors listed in <u>Schedule 1</u> has, during the last five years, been a party to a civil proceeding of a judicial or administrative body of competent jurisdiction and as a result of such proceeding was or is subject to a judgment, decree or final order enjoining future violations of, or prohibiting or mandating activities subject to, federal or state securities laws or finding any violation with respect to such laws.

Each of the executive officers and directors of the Reporting Person listed in <u>Schedule 1</u> are citizens of Japan, except for Stephen Gomersall, who is a citizen of the United Kingdom.

### Item 3. Source and Amount of Funds or Other Consideration.

Effective March 8, 2012 (the "<u>Closing Date</u>"), the Reporting Person received 25,000,000 Shares together with cash consideration in connection with the Issuer's purchase from the Reporting Person of all of the outstanding stock of Viviti Technologies Ltd., a company incorporated under the laws of the Republic of Singapore (such stock, the "<u>Company Stock</u>"; such company, the "<u>Company</u>") and a wholly owned subsidiary of the Reporting Person prior to the Closing Date, pursuant to a Stock Purchase Agreement made and entered into as of March 7, 2011, by and among the Issuer, Western Digital Ireland, Ltd., a corporation organized under the

laws of the Cayman Islands and an indirect wholly owned subsidiary of the Issuer (the "<u>Buyer</u>"), the Reporting Person, and the Company (as amended through the date hereof, the "<u>Agreement</u>").

The aggregate consideration paid by the Issuer for the Company Stock was approximately \$4.9 billion, which consisted of 25,000,000 Shares and approximately \$3.9 billion in cash.

### Item 4. Purpose of Transaction.

The acquisition of the 25,000,000 Shares referred to in this Statement and that was contemplated and consummated pursuant to the Agreement is for the Reporting Person's investment in the Issuer. In addition, pursuant to an Investor Rights Agreement made as of March 8, 2012, among the Issuer and the Reporting Person (the "<u>IRA</u>"), the Reporting Person, among other rights and obligations:

- has the right, but not the obligation, to designate two nominees to serve as directors of the Issuer;
- agrees not to authorize, permit or direct its subsidiaries or affiliates to, directly or indirectly, sell or otherwise transfer any interest in the Shares acquired pursuant to the Agreement for one year following the Closing Date without the consent of the Issuer;
- agrees not to, directly or indirectly, transfer, sell or otherwise dispose of any of the Shares acquired pursuant to the Agreement to certain persons who are listed on a competitor list letter provided by the Issuer, which list may be amended from time to time by the Issuer; and
- has registration rights with respect to the Shares acquired pursuant to the Agreement, including shelf, demand and piggyback registration rights.

In the event the Reporting Person determines to designate two nominees to serve as directors of the Issuer, the Issuer shall increase the size of its Board of Directors by two, and fill the resulting vacancies with such designees in accordance with the Issuer's Bylaws. The Reporting Person's right to designate nominees shall terminate automatically at the end of the second full calendar year following the Closing Date with respect to one nominee and shall terminate with respect to either or both nominees automatically, or automatically upon written notice from the Issuer, upon the occurrence of certain events prescribed in the IRA.

Other than as described in this Item 4, neither the Reporting Person nor, to the knowledge of the Reporting Person, any of its directors or executive officers listed on <u>Schedule 1</u> to this Statement, has any present plans or intentions which relate to or would result in any of the transactions required to be described in Item 4 of this Statement.

The descriptions of the transactions and agreements set forth herein are qualified in their entirety by the terms and conditions of the complete agreements governing such matters, each of which is incorporated by reference into this Statement as exhibits pursuant to Item 7 of this



Statement. Please refer to the Agreement and the IRA attached to this Statement as Exhibits 99.1 and 99.2 respectively for further details.

### Item 5. Interest in Securities of the Issuer.

(a) The Reporting Person beneficially owns, within the meaning of Rule 13d-3 under the Securities Exchange Act of 1934, an aggregate of 25,000,000 Shares. The Shares owned by the Reporting Person constitute approximately 9.6% of the Shares issued and outstanding (computed on the basis of 234,930,822 Shares issued and outstanding, as reported in the Issuer's current report on Form 10-Q, filed with the Securities and Exchange Commission on January 27, 2012 plus 25,000,000 Shares issued on the Closing Date in connection with the Agreement).

(b) The Reporting Person possesses sole power to vote and to dispose of the 25,000,000 Shares pursuant to the terms and conditions of the IRA as referenced in Item 4 of this Statement.

(c) There are no other transactions in the Shares that were effected during the past sixty days other than the consummation of the acquisition of the Company pursuant to the Agreement as described in Item 3 of this Statement.

- (d) None.
- (e) Not Applicable.

### Item 6. Contracts, Arrangements, Understandings or Relationships With Respect to Securities of the Issuer.

Except as described in this Statement, neither the Reporting Person nor, to the knowledge of the Reporting Person, any of its executive officers or directors listed on <u>Schedule 1</u> to this Statement, has any contract, arrangement, understanding or relationship (legal or otherwise) with any person with respect to the Shares held by the Reporting Person, including, but not limited to, transfer or voting of any of such securities, finder's fees, joint ventures, loan or option arrangements, puts or calls, guarantees of profits, division of profits or loss, or the giving or withholding of proxies.

# Item 7. Material to be Filed as Exhibits.

Exhibit 99.1	Stock Purchase Agreement entered into as of March 7, 2011, by and among the Reporting Person, the Issuer, the Buyer, and the Company
	as amended on May 27, 2011 and November 23, 2011 (incorporated by reference to Exhibit 2.1 to the Issuer's Current Report on Form 10-
	Q, dated May 2, 2011, Exhibit 2.2 to the Issuer's Current Report on Form 10-K, dated August 12, 2011, and Exhibit 2.3 to the Issuer's
	Current Report on Form 10-Q, dated January 27, 2012, respectively), and by the amendments filed as Exhibits 99.2, 99.3, 99.4 and 99.5 to
	this Statement
Exhibit 99.2	Third Amendment to Stock Purchase Agreement, dated January 30, 2012

- Exhibit 99.2 I hird Amendment to Stock Purchase Agreement, dated January 30, 2012
- Exhibit 99.3 Fourth Amendment to Stock Purchase Agreement, dated February 15, 2012
- Exhibit 99.4 Fifth Amendment to Stock Purchase Agreement, dated March 6, 2012

Exhibit 99.5	Sixth Amendment to Stock Purchase Agreement, dated March 6, 2012
Exhibit 99.6	Investor Rights Agreement made as of March 8, 2012, among the Issuer and the Reporting Person

# SIGNATURE

After reasonable inquiry and to the best of my knowledge and belief, I certify that the information set forth in this Statement is true, complete and correct.

Date: March 15, 2012

# Hitachi, Ltd.

By: /s/ Toshiaki Kuzuoka

Name:Toshiaki KuzuokaTitle:Senior Vice President and Executive Officer

# Schedule 1

The following table sets forth (i) the name and present principal occupation or employment, for each of the Reporting Person's executive officers and directors and (ii) the name, principal business and address of any corporation or other organization in which such employment is conducted. Unless otherwise indicated, the business address of each such executive officer and director is 6-6, Marunouchi 1-chome, Chiyoda-ku, Tokyo, 100-8280, Japan. Each of the executive officers and directors of the Reporting Person listed below is a citizen of Japan, except for Stephen Gomersall, who is a citizen of the United Kingdom.

### Directors

Name	Present Principal Occupation or Employment (Principal Position outside Hitachi, if any)	
Takashi Kawamura	Chairman of the Board	
Yoshie Ota	Director	
Mitsuo Ohashi	Director (Advisor, Showa Denko K.K.)	
Nobuo Katsumata	Director (Chairman, Member of the Board, Marubeni Corporation)	
Tohru Motobayashi	Director (Attorney at law)	
Isao Ono	Director (Chairman of the Board, Hitachi Solutions, Ltd.)	
Stephen Gomersall	Director (Chairman of the Board, Hitachi Europe Ltd.)	
Tadamichi Sakiyama	Director	
Masaharu Sumikawa	Director (Chairman of the Board, Hitachi Plant Technologies, Ltd.)	
Hiroaki Nakanishi	Director*	
Takashi Hatchoji	Director (Chairman of the Board, Hitachi America, Ltd.)	
Takashi Miyoshi	Director*	
Note: The Directors marked with * concurrently serve as Executive Officers. See "Executive Officers" below.		

Present Principal Occupation or Employment
(Responsibility of Executive Officer as authorized
by the Board of Directors)

Name	by the Board of Directors)
Hiroaki Nakanishi	Representative Executive Officer President
Koji Tanaka	(Overall management) Representative Executive Officer
	Executive Vice President and Executive Officer (Power systems business, transportation systems business, social infrastructure & industrial systems business, urban planning and development systems business and defense systems business)
Junzo Nakajima	Representative Executive Officer Executive Vice President and Executive Officer (Information & control systems business, information & telecommunication systems business and information technology)
Takashi Miyoshi	Representative Executive Officer Executive Vice President and Executive Officer (Management strategies, finance and corporate pension system)
Nobuo Mochida	Representative Executive Officer Executive Vice President and Executive Officer (High functional materials & components, quality assurance, production engineering, battery systems business and automotive systems business)
Kazuhiro Mori	Representative Executive Officer Executive Vice President and Executive Officer (Sales operations, Hitachi group global business, corporate export regulation, medical systems business and business incubation)
Shigeru Azuhata	Senior Vice President and Executive Officer (Research & development, environmental strategies and medical systems business)
Shinjiro Iwata	Senior Vice President and Executive Officer (Information & telecommunication systems business)

# Present Principal Occupation or Employment (Responsibility of Executive Officer as authorized

Name	by the Board of Directors)
Makoto Ebata	Senior Vice President and Executive Officer (Procurement and consumer business)
Toshiaki Kuzuoka	Senior Vice President and Executive Officer (Human capital, legal and corporate communications, government & external relations, corporate brand and corporate auditing)
Toyoaki Nakamura	Representative Executive Officer Senior Vice President and Executive Officer (Finance and corporate pension system)
Toshio Ikemura	Vice President and Executive Officer (Urban planning and development systems business)
Hitoshi Isa	Vice President and Executive Officer (Power systems business (thermal power systems business promotion))
Tatsuro Ishizuka	Vice President and Executive Officer (Power systems business)
Osamu Ohno	Vice President and Executive Officer (Information technology)
Nobuyuki Ohno	Vice President and Executive Officer (Hitachi group global business (China))
Yoshifumi Kanda	Vice President and Executive Officer (Power systems business (sales operations))
Masahiro Kitano	Vice President and Executive Officer (Environmental strategies, quality assurance and production engineering)
Ryuichi Kitayama	Vice President and Executive Officer (Sales operations)
Kazuhiro Kurihara	Vice President and Executive Officer (Sales operations (Chubu Area))
Yutaka Saito	Vice President and Executive Officer (Information & control systems business)

### Present Principal Occupation or Employment (Responsibility of Executive Officer as authorized by the Board of Directors)

Name	by the Board of Directors)
Kaichiro Sakuma	Vice President and Executive Officer (Information & telecommunication systems business (platform systems business))
Yasuo Tanabe	Vice President and Executive Officer (Government & external relations)
Masahide Tanigaki	Vice President and Executive Officer (Sales operations, Hitachi group global business and corporate export regulation)
Toshikazu Nishino	Vice President and Executive Officer (Management strategies)
Masaharu Hanyu	Vice President and Executive Officer (Power systems business (nuclear power systems business promotion))
Toshiaki Higashihara	Vice President and Executive Officer (Social infrastructure & industrial systems business)
Naoki Mitarai	Vice President and Executive Officer (Human capital)
Yoshihiko Mogami	Vice President and Executive Officer (Information & telecommunication systems business (system solutions business))
	11

### THIRD AMENDMENT TO STOCK PURCHASE AGREEMENT

This Third Amendment to the Stock Purchase Agreement (this "<u>Amendment</u>") is made this 30<sup>th</sup> day of January, 2012, by and among Western Digital Corporation, a Delaware corporation (the "<u>Buyer Parent</u>"), Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect wholly owned subsidiary of the Buyer Parent (the "<u>Buyer</u>"), Hitachi, Ltd., a company incorporated under the laws of Japan (the "<u>Seller</u>"), and Viviti Technologies Ltd., a company incorporated under the laws of the Republic of Singapore and a wholly owned subsidiary of the Seller (the "<u>Company</u>", and collectively with the Seller, Buyer and Buyer Parent, the "<u>Parties</u>", and each, a "<u>Party</u>"). Capitalized terms used herein and not otherwise defined shall have the meanings set forth in the Stock Purchase Agreement (as defined below).

WHEREAS, the Parties entered into a Stock Purchase Agreement on March 7, 2011, as amended by a First Amendment to Stock Purchase Agreement, dated May 27, 2011, and as further amended by a Second Amendment to Stock Purchase Agreement, dated November 23, 2011 (together, the "Stock Purchase Agreement"); and

WHEREAS, the Parties desire to amend the Stock Purchase Agreement as reflected herein.

NOW, therefore, in exchange for good and valuable consideration, the receipt of which is hereby acknowledged, the Parties hereby agree as follows:

1. **Exhibit B of the Stock Purchase Agreement**. Exhibit B of the Stock Purchase Agreement, Form of IP License Agreement, is hereby deleted in its entirety and replaced by the Form of IP License Agreement attached hereto as <u>Exhibit A</u>.

2. Effect on the Stock Purchase Agreement. This Amendment shall not constitute a waiver, amendment or modification of any provision of the Stock Purchase Agreement not expressly referred to herein. Except as expressly amended or modified herein, the provisions of the Stock Purchase Agreement are and shall remain in full force and effect and are hereby ratified and confirmed. On and after the date hereof, each reference in the Stock Purchase Agreement to "this Agreement", "herein", "hereof", "hereunder" or words of similar import shall mean and be a reference to the Stock Purchase Agreement as amended hereby, although it shall not alter the dates as of which any provision of the Stock Purchase Agreement speaks. For example, phrases such as "as of the date hereof" and "as of the date of this Agreement" shall continue to refer to March 7, 2011, the date that the Stock Purchase Agreement was originally executed. To the extent that a provision of this Amendment conflicts with or differs from a provision of the Stock Purchase Agreement, such provision of this Amendment shall prevail and govern for all purposes and in all respects.

3. Miscellaneous. Sections 12.4, 12.5, 12.7 and 12.13 of the Stock Purchase Agreement are incorporated herein by reference.

[Signature page follows]

above.

**"BUYER PARENT"** 

### WESTERN DIGITAL CORPORATION

By: /s/ John F. Coyne

Name: John F. Coyne Title: President & CEO

### **"BUYER"**

### WESTERN DIGITAL IRELAND, LTD.

By: /s/ Michael Ray

Name: Michael Ray Title: Vice President

### "SELLER"

# HITACHI, LTD.

By: /s/ Toyoki Furuta Name: Toyoki Furuta Title: General Manager, Business Development Office

### "COMPANY"

# VIVITI TECHNOLOGIES LTD.

By: /s/ Christopher Dewees Name: Christopher Dewees Title: SVP & General Counsel

Signature Page to Third Amendment to Stock Purchase Agreement

# EXHIBIT A

Amended and Restated Form of IP License Agreement

[See Attached]

#### FORM OF

### LICENSE AGREEMENT

This License Agreement ("**Agreement**"), effective as of the Closing Date (as defined below), is entered into by and between Western Digital Corporation ("**Buyer Parent**"), a Delaware company with its principal place of business at 3355 Michelson Drive, Suite 100 Irvine, California 92612, USA, and Hitachi, Ltd. ("**HITACHI**"), a Japanese company with its principal place of business at 6-6, Marunouchi 1-chome, Chiyoda-ku, Tokyo 100-8280, Japan. Each of the signatories to this Agreement is referred to as a "**Party**," and jointly as the "**Parties**."

### RECITALS

WHEREAS, Buyer Parent, Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect whollyowned subsidiary of the Buyer Parent (the "**Buyer**"), HITACHI, and Hitachi Global Storage Technologies Holdings Pte. Ltd., a company incorporated under the laws of the Republic of Singapore ("**HGST**") and a wholly owned subsidiary of HITACHI, have entered into that certain Stock Purchase Agreement dated as of March 7, 2011 (the "**Purchase Agreement**");

WHEREAS, pursuant to the Purchase Agreement, Buyer Parent desires to enter into a patent cross-license with HITACHI for certain technologies and enter into a mutual release of claims pursuant to the terms and conditions of this Agreement; and

WHEREAS, pursuant to the Purchase Agreement, HITACHI desires to enter into a patent cross-license with Buyer Parent for certain technologies and enter into a mutual release of claims pursuant to the terms and conditions of this Agreement.

NOW, THEREFORE, in consideration of the mutual promises and covenants set out below, the Parties agree as follows:

#### TERMS OF AGREEMENT

### 1. **DEFINITIONS**

- 1.1 **"External Drive**" means an enclosure that encases one or more HDDs or SSDs and related peripheral hardware, or Software or Firmware components (other than RAID).
- 1.2 **"Existing Terms"** means terms and conditions in written agreements existing as of the Closing Date between HGST or any of its Subsidiaries, on the one hand, and HITACHI or any HITACHI Sublicensed Subsidiary, on the other hand, with respect to license limitations, restrictions on use (with respect to any intellectual property rights, Works or materials not owned by HITACHI or a HITACHI Sublicensed Subsidiary), restrictions on disclosure, royalties or other consideration, and/or license duration.

- 1.3 **"Have Made**" means to have any products or other items subject to the licenses granted herein made by a third-party manufacturer for the use, lease, sale or other transfer by a Party or its Subsidiaries; provided, however, that (i) the specifications for such products or other items were solely or jointly created by or at the direction of such Party or its Subsidiaries, and (ii) such product or other item is not an off-the-shelf product of a third-party manufacturer.
- 1.4 **"HDD**" means any device designed to magnetically record and/or read digital information on or from a rotating disk ("**Rotating Magnetic Storage**") that contains one or more spindle motors, one or more magnetic heads, and one or more actuators, all of which are incorporated in a single enclosure, and related peripheral hardware, or Software or Firmware components. For the avoidance of doubt "HDD" (a) includes any device that incorporates both Rotating Magnetic Storage and other digital storage media, such as semiconductor-based memory, but only in the case that the memory capacity of Rotating Magnetic Storage supersedes that of other digital storage media, and (b) excludes External Drive, RAID (as defined below), Information Versatile Disk for Removable Usage (iVDR), and optical disk products.
- 1.5 **"Buyer Parent SSD Patents"** means all Patents related to SSD (i) issued or issuing on patent applications entitled to a first effective filing date on or before the end of the Term, (ii) under which Buyer Parent or any of its Subsidiaries has now or during the Term obtains the right to grant licenses within the scope granted herein without the payment of royalties or other consideration to any third party, and (iii) that would be, but for this Agreement, directly or indirectly infringed by HITACHI or HITACHI Sublicensed Subsidiaries making, using, importing, exporting, leasing, selling, offering to sell, or otherwise disposing of SSD.
- 1.6 **"Storage Software Products**" means the Hitachi Backup<sup>TM</sup> software product and the Lifestudio<sup>TM</sup> software product.
- 1.7 **"Storage Software Patents**" means all Patents related to the Storage Software Products (i) issued or issuing on patent applications entitled to a first effective filing date on or before the end of the Term, (ii) under which Hitachi or any HITACHI Sublicensed Subsidiary has now or during the Term obtains the right to grant licenses within the scope granted herein without the payment of royalties or other consideration to any third party, and (iii) that would be, but for this Agreement, directly or indirectly infringed by Buyer Parent or its Subsidiaries making, using, importing, exporting, selling, leasing, offering to sell, or otherwise disposing of the version of either Storage Software Product that is generally commercially available as of the Closing Date.
- 1.8 **"HGST RAID Patents**" means all Patents related to RAID (i) issued or issuing on patent applications entitled to a first effective filing date on or before the first anniversary of the Closing Date (ii) under which Buyer Parent or any of its Subsidiaries has now or hereafter obtains the right to grant licenses within the

scope granted herein without the payment of royalties or other consideration to any third party, (iii) whose named inventors are or were employees or contractors of HGST or any of its Subsidiaries on or prior to the Closing Date or that were assigned or exclusively licensed to HGST or any of its Subsidiaries on or prior to the Closing Date, and (iv) that would be, but for this Agreement, directly or indirectly infringed by HITACHI or HITACHI Sublicensed Subsidiaries making, using, importing, exporting, leasing, selling, offering to sell, or otherwise disposing of RAID.

- 1.9 **"Buyer Parent RAID Patents**" means all Patents related to RAID (i) issued or issuing on patent applications entitled to a first effective filing date on or before the end of the Term (ii) under which Buyer Parent or any of its Subsidiaries has now or hereafter obtains the right to grant licenses within the scope granted herein without the payment of royalties or other consideration to any third party, and (iii) that would be, but for this Agreement, directly or indirectly infringed by HITACHI or HITACHI Sublicensed Subsidiaries making, using, importing, exporting, leasing, selling, offering to sell, or otherwise disposing of RAID. For clarity, the Buyer Patent RAID Patents include, but are not limited to, the HGST RAID Patents.
- 1.10 "Closing Date" has the meaning set forth in the Purchase Agreement.
- 1.11 **"HITACHI External Drive Patents**" means all Patents related to External Drive (i) issued or issuing on patent applications entitled to a first effective filing date on or before the end of the Term, (ii) under which HITACHI or any HITACHI Sublicensed Subsidiary has now or during the Term obtains the right to grant licenses within the scope granted herein without the payment of royalties or other consideration to any third party, and (iii) that would be, but for this Agreement, directly or indirectly infringed by Buyer Parent or its Subsidiaries making, using, importing, exporting, selling, leasing, offering to sell, or otherwise disposing of External Drive.
- 1.12 **"HITACHI HDD Patents**" means all Patents related to HDD (i) issued or issuing on patent applications entitled to a first effective filing date on or before the end of the Term, (ii) under which HITACHI or any HITACHI Sublicensed Subsidiary has now or hereafter obtains the right to grant licenses within the scope granted herein without the payment of royalties or other consideration to any third party, and (iii) that would be, but for this Agreement, directly or indirectly infringed by Buyer Parent or its Subsidiaries making, using, importing, exporting, leasing, selling, offering to sell, or otherwise disposing of HDD.
- 1.13 **"HITACHI Exclusions**" means (i) any Works (*e.g.*, Software) that as of the Closing Date are made generally commercially available by HITACHI or any of the HITACHI Sublicensed Subsidiaries, and (ii) any proprietary information of HITACHI or any of the HITACHI Sublicensed Subsidiaries that has been disclosed to HGST or any of its Subsidiaries in its role as a supplier or potential supplier of products to HITACHI or a HITACHI Sublicensed Subsidiary.

- 1.14 **"HITACHI Licensed Copyrights**" means copyrights owned by HITACHI or any of the HITACHI Sublicensed Subsidiaries as of the Closing Date in any Works that were provided to HGST by HITACHI or any HITACHI Sublicensed Subsidiary prior to, and are used by HGST or any of its Subsidiaries as of, the Closing Date; <u>provided</u>, <u>however</u>, that "HITACHI Licensed Copyrights" do not include any copyrights embodied or used in any HITACHI Exclusions.
- 1.15 **"HITACHI Licensed Trade Secrets**" means trade secrets owned by HITACHI or any of the HITACHI Licensed Subsidiaries as of the Closing Date that were provided to HGST by HITACHI or any HITACHI Sublicensed Subsidiary prior to, and are used by HGST or any of its Subsidiaries as of, the Closing Date; <u>provided</u>, <u>however</u>, that "HITACHI Licensed Trade Secrets" do not include any trade secrets embodied or used in HITACHI Exclusions.
- 1.16 "HITACHI RAID Patents" means all Patents related to RAID (i) issued or issuing on patent applications entitled to a first effective filing date on or before the end of the Term, (ii) under which HITACHI or any HITACHI Sublicensed Subsidiary has now or during the Term obtains the right to grant licenses within the scope granted herein without the payment of royalties or other consideration to any third party, and (iii) that would be, but for this Agreement, directly or indirectly infringed by Buyer Parent or its Subsidiaries making, using, importing, exporting, selling, leasing, offering to sell, or otherwise disposing of RAID.
- 1.17 "HITACHI SSD Patents" means all Patents related to SSD (i) issued or issuing on patent applications entitled to a first effective filing date on or before the end of the Term, and (ii) under which HITACHI or any HITACHI Sublicensed Subsidiary has now or during the Term obtains the right to grant licenses within the scope granted herein without the payment of royalties or other consideration to any third party, and (iii) that would be, but for this Agreement, directly or indirectly infringed by Buyer Parent or its Subsidiaries making, using, importing, exporting, selling, leasing, offering to sell, or otherwise disposing of SSD.
- 1.18 **"HITACHI Sublicensed Subsidiary**" means any HITACHI Subsidiary to which the rights, licenses, releases, covenants and privileges are extended in accordance with Section 2.4.
- 1.19 **"Non-HDD Business**" means any entity (or internal business division or organization of an entity) that is principally engaged in a business other than the design, sale, or manufacture of HDDs.
- 1.20 "**Patents**" means all classes and types of patents, including utility patents, utility models, design patents, invention certificates, including divisionals, continuations, continuations-in-part, reexaminations, reissues, extensions and renewals, in all jurisdictions of the world.
- 1.21 "RAID" means a device comprised of an assembly of storage media components and related peripheral hardware, or Software or Firmware components contained

within one or more enclosures, combined with associated Software or Firmware, for the purpose of data storage and retrieval, designed to protect digital information from a single component failure or to enhance performance by dividing data among multiple storage media components. For the avoidance of doubt, "RAID" excludes storage media components incorporated therein.

- 1.22 **"Software or Firmware**" means a set of instructions, that either (i) directly provides instructions to the computer hardware, or, (ii) indirectly serves as an input to another piece of software.
- 1.23 "SSD" means a product, component, subcomponent or subassembly thereof, and related peripheral hardware, or Software or Firmware components, designed, in whole or in part, to electrically write, read and/or erase digital information from storage media or memory (including, without limitation, NOR-flash, NAND-flash, or MRAM semiconductor microchips), which utilizes an HDD-type interface (or an interface also employed for HDDs), including without limitation, PATA, SATA, SAS, PCI Express, SCI, Fibre Channel, USB 3.0 and InfiniBand. For the avoidance of doubt, "SSD" excludes HDD and RAID.
- 1.24 **"Subsidiary**" of either party shall mean a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities or other interests having ordinary voting power for the election of directors or other governing body are at the time beneficially owned, directly, or indirectly, by such party. For the purposes of this Agreement, HGST and its Subsidiaries will be Subsidiaries of Buyer Parent (subject to meeting the requirements of the preceding sentence) and are not Subsidiaries of HITACHI.
- 1.25 **"Term**" means the period from the Closing Date through the fifth anniversary of the Closing Date.
- 1.26 **"HDD Component**" means any component, subcomponent, subassembly, Software or Firmware which is specifically designed for HDD, whether or not sold as a stand-alone unit or integrated within an HDD.
- 1.27 **"Works**" means works of authorship and other copyrightable materials and subject matter, including Software to the extent within the subject matter of copyright.
- 1.28 **"3.5" HDD**" means an HDD in the 3.5" form factor having an enclosure with a width of 101.6mm and a depth of 146mm that is designed and sold for use in, and used in, consumer electronics and desktop personal computer products only. For clarity, the term "3.5" HDD" does not include (a) HDDs in any form factor designed for use in, sold for use in or used in servers or other applications other than consumer electronics or desktop personal computers or (b) HDD Components.

### 2. LICENSES, RELEASES, AND COVENANTS GRANTED TO HITACHI

- 2.1 <u>SSD License to HITACHI</u>. Buyer Parent, on behalf of itself and its Subsidiaries, hereby grants to HITACHI a non-exclusive, irrevocable, fully paid-up, world-wide term license, without the right to grant a sublicense (other than pursuant to Section 2.4), under the Buyer Parent SSD Patents, to make, Have Made, use, import, export, sell, lease, offer to sell, or otherwise dispose of SSD.
- 2.2 <u>RAID License to HITACHI</u>. Buyer Parent, on behalf of itself and its Subsidiaries, hereby grants to HITACHI a non-exclusive, irrevocable, fully paid-up, world-wide, license, without the right to grant a sublicense (other than pursuant to Section 2.4), under the Buyer Parent RAID Patents, to make, Have Made, use, import, export, sell, lease, offer to sell, or otherwise dispose of RAID.
- 2.3 <u>Past Release to HITACHI</u>. Buyer Parent, on behalf of itself and its Subsidiaries, hereby irrevocably releases and forever discharges HITACHI and the HITACHI Sublicensed Subsidiaries from any and all claims, losses, demands, causes of action, liabilities, and rights of action of any kind, at law or in equity, whether known or unknown, suspected and unsuspected, disclosed and undisclosed, liquidated or unliquidated, for any act of direct or indirect infringement, before the Closing Date, of any Buyer Parent SSD Patent or Buyer Parent RAID Patent by HITACHI or any HITACHI Sublicensed Subsidiary making, using, importing, exporting, selling, leasing, offering to sell, or otherwise disposing of SSD (in the case of Buyer Parent SSD Patents) or RAID (in the case of Buyer Parent RAID Patents). For the avoidance of doubt, the foregoing release is intended to and does apply to all activities of HITACHI and the HITACHI Sublicensed Subsidiaries that fall within the scope of the licenses granted by Buyer Parent set forth in this Agreement that were carried out before the Closing Date.
- 2.4 <u>Extension to HITACHI's Subsidiaries</u>. HITACHI shall have a right to extend to its Subsidiaries the rights, releases, licenses, covenants and any other privileges granted to HITACHI hereunder, if such Subsidiary agrees to abide by the terms and conditions of this Agreement and include its Patents in HITACHI HDD Patents, HITACHI External Drive Patents, HITACHI RAID Patents, and HITACHI SSD Patents as if it were named in the place of HITACHI hereunder, subject to the termination provision in Section 4.2.
- 2.5 <u>Covenant Not To Sue HITACHI Regarding RAID</u>. Buyer Parent and its Subsidiaries shall reserve the right to sue the sellers and/or manufacturers of HDD incorporated into RAID sold to HITACHI or HITACHI Sublicensed Subsidiaries for infringement of Buyer Parent Patents related to HDD by making, using, selling, offering to sell, leasing, importing or otherwise disposing of such HDD incorporated into RAID; provided however, that Buyer Parent, on behalf of itself and its Subsidiaries, covenants not to sue HITACHI, HITACHI Sublicensed Subsidiaries, or their customers (to the extent the RAID was purchased from HITACHI or HITACHI Sublicensed Subsidiaries) for infringement of Buyer

Parent Patents related to HDD by using, leasing, importing, exporting, selling, offering to sell such HDD incorporated into RAID.

### 3. LICENSES, RELEASE AND COVENANTS GRANTED TO BUYER PARENT

- 3.1 <u>HDD/HDD Component License to Buyer Parent</u>. HITACHI, on behalf of itself and HITACHI Sublicensed Subsidiaries, hereby grants to Buyer Parent and its Subsidiaries a non-exclusive, irrevocable, fully paid-up, world-wide, license, without the right to grant a sublicense (other than pursuant to Section 3.11), under the HITACHI HDD Patents, to make, Have Made, use, import, export, sell, lease, offer to sell, or otherwise dispose of HDD and HDD Components.
- 3.2 <u>SSD License to Buyer Parent</u>. HITACHI, on behalf of itself and HITACHI Sublicensed Subsidiaries, hereby grants to Buyer Parent and its Subsidiaries a non-exclusive, irrevocable, fully paid-up, world-wide term license, without the right to grant a sublicense under HITACHI SSD Patents, to make, Have Made, use, import, export, sell, lease, offer to sell, or otherwise dispose of SSD.
- 3.3 <u>RAID License to Buyer Parent</u>. HITACHI, on behalf of itself and HITACHI Sublicensed Subsidiaries, hereby grants to Buyer Parent and its Subsidiaries a non-exclusive, irrevocable, fully paid-up, world-wide term license, without the right to grant a sublicense , under HITACHI RAID Patents, to make, Have Made, use, import, export, sell, lease, offer to sell, or otherwise dispose of RAID.
- 3.4 <u>External Drive License to Buyer Parent</u>. HITACHI, on behalf of itself and HITACHI Sublicensed Subsidiaries, hereby grants to Buyer Parent and its Subsidiaries a non-exclusive, irrevocable, fully paid-up, world-wide term license, without the right to grant a sublicense , under HITACHI External Drive Patents, to make, Have Made, use, import, export, sell, lease, offer to sell, or otherwise dispose of External Drive.
- 3.5 <u>Storage Software License</u>. HITACHI, on behalf of itself and HITACHI Sublicensed Subsidiaries, hereby grants to Buyer Parent and its Subsidiaries a non-exclusive, irrevocable, fully paid-up, world-wide, term license, without the right to grant a sublicense , under the Storage Software Patents, to make, Have Made, use, import, export, sell, lease, offer to sell, or otherwise dispose of the Storage Software Products, including future versions thereof.
- 3.6 <u>HITACHI Licensed Copyrights License</u>. Subject, in the case of any particular HITACHI Licensed Work, to any applicable Existing Terms, HITACHI, on behalf of itself and the HITACHI Sublicensed Subsidiaries, hereby grants to Buyer Parent and its Subsidiaries a nonexclusive, irrevocable, fully paid-up, world-wide, perpetual license under the HITACHI Licensed Copyrights to reproduce, distribute, display, perform, and make derivative works of any Works covered by the HITACHI Licensed Copyrights ("**HITACHI Licensed Works**"), in each case, in the ordinary course of the business of Buyer Parent and its Subsidiaries (such license, the "**HITACHI Licensed Copyrights License**").

Buyer Parent and its Subsidiaries may grant sublicenses under the HITACHI Licensed Copyrights License to third parties (a) as provided in Section 3.11 and (b) in connection with the development, manufacture, distribution, provision, sale and use of any products or services of Buyer Parent and its Subsidiaries, in each case, solely in such fields, provided that any disclosure to such third parties of HITACHI Licensed Trade Secrets embodied in HITACHI Licensed Works (and any use of such HITACHI Licensed Trade Secrets by such third parties) shall be only as permitted by Section 3.8.

- 3.7 <u>HITACHI Licensed Trade Secrets License</u>. Subject to the terms and conditions of this Agreement and, in the case of any particular HITACHI Licensed Trade Secret, to any applicable Existing Terms, HITACHI, on behalf of itself and the HITACHI Sublicensed Subsidiaries, hereby grants to Buyer Parent and its Subsidiaries, a nonexclusive, irrevocable, fully paid-up, worldwide, perpetual license under the HITACHI Licensed Trade Secrets to use the HITACHI Licensed Trade Secrets, in each case, in the ordinary course of the business of Buyer Parent and its Subsidiaries (the "HITACHI Licensed Trade Secrets License"). The HITACHI Licensed Trade Secrets shall be deemed Confidential Information of HITACHI and the HITACHI Trade Secret License shall be subject to the terms and conditions of Section 3.8. Notwithstanding the preceding sentence, Buyer Parent and its Subsidiaries are busined in Section 3.11 and (b) reasonably necessary in connection with the development, manufacture, distribution, provision, sale, and use of the products and services of Buyer Parent and its Subsidiaries, in each case, solely in such fields, provided that such third parties are bound in writing by confidentiality terms at least as protective as those set forth in Section 3.8. Except as expressly set forth in the preceding sentence or as expressly permitted by Section 3.8, Buyer Parent and its Subsidiaries shall not (and shall have no right to) disclose or allow the use of any HITACHI Licensed Trade Secret.
- 3.8 <u>Confidentiality</u>. Buyer Parent and its Subsidiaries shall maintain the HITACHI Licensed Trade Secrets with at least the same degree of care it uses to protect its own proprietary information of a similar nature or sensitivity, but no less than reasonable care under the circumstances. Buyer Parent and its Subsidiaries shall not disclose any HITACHI Licensed Trade Secrets to third parties except where (i) such disclosure is required by law or order of a court of competent jurisdiction, provided that, in such event, Buyer Parent shall provide HITACHI prompt, advance notice of such requirement to allow intervention (and shall cooperate with HITACHI) to contest or minimize the scope of the disclosure (including through application for a protective order), (ii) the HITACHI Licensed Trade Secret disclosed was in the public domain prior to the disclosure to Buyer Parent or its Subsidiaries, (iii) the HITACHI Licensed Trade Secret disclosed becomes part of the public domain by publication or otherwise except by breach of this Agreement, or (iv) Buyer Parent can establish by competent proof the HITACHI Licensed Trade Secret disclosed was received from a third party without restrictions on confidentiality. Buyer Parent shall be responsible to HITACHI for

the acts and omissions of any Subsidiary, employee or agent with respect to such confidentiality obligations. To the extent a particular HITACHI Licensed Trade Secret is subject to confidentiality obligations under a written agreement existing as of the Closing Date between HGST or any of its Subsidiaries, on the one hand, and HITACHI or any HITACHI Sublicensed Subsidiary, on the other hand, the obligations set forth in this Section 3.8 will not extend in duration beyond the expiration of the confidentiality obligations set forth in such written agreement.

- 3.9 Past Release to Buyer Parent. HITACHI, on behalf of itself and HITACHI Sublicensed Subsidiaries, hereby irrevocably releases and forever discharges Buyer Parent and its Subsidiaries from any and all claims, losses, demands, causes of action, liabilities, and rights of action of any kind, at law or in equity, whether known or unknown, suspected and unsuspected, disclosed and undisclosed, liquidated or unliquidated, for any act of direct or indirect infringement, before the Closing Date, of any HITACHI HDD Patent, HITACHI SSD Patent, HITACHI External Drive Patent, HITACHI RAID Patent or Storage Software Patent by Buyer Parent or its Subsidiaries making, using, importing, exporting, selling, leasing, offering to sell, or otherwise disposing of HDD (in the case of HITACHI HDD Patents), External Drive (in the case of HITACHI External Drive Patents), SSD (in the case of HITACHI SSD Patents), RAID (in the case of HITACHI RAID patents) or Storage Software Products (in the case of Storage Software Patents). For the avoidance of doubt, the foregoing release is intended to and does apply to all activities of Buyer Parent and its Subsidiaries that fall within the scope of the licenses granted by HITACHI set forth in this Agreement that were carried out before the Closing Date.
- 3.10 <u>No Licenses to Third Parties</u>. Nothing in this Agreement shall be deemed to grant any license, immunity, release, or other right to any person or entity, other than the Parties and their respective Subsidiaries, as provided herein.
- 3.11 <u>Divestment Sublicense</u>. Buyer Parent shall have a right to grant to Toshiba Corporation ("**Toshiba**") a sublicense under the rights granted to Buyer Parent pursuant to Sections 3.1, 3.6 and 3.7, effective upon the closing of Toshiba's purchase from Buyer Parent of assets relating to the 3.5" HDD business owned by Viviti Technologies Limited pursuant to the Purchase Agreement in connection with any commitment to, order by, regulation of, or other obligation to a regulatory body, competition authority or similar entity including the commitments under the Case No. Comp M.6203 Western Digital / Viviti Technologies Commitments to the European Commission (collectively, such assets purchased by Toshiba constituting the "**Divested Business**"), subject to the following conditions:
  - (a) the sublicense granted to Toshiba under the rights granted to Buyer Parent pursuant to Sections 3.1, 3.6 and 3.7 will apply only with respect to developing, making, Having Made, using, importing, exporting, selling, leasing, offering to sell, distributing and otherwise disposing of 3.5" HDDs in the ordinary course of Toshiba's 3.5" HDD business;

- (b) the sublicense granted to Toshiba under the rights granted to Buyer Parent pursuant to Sections 3.1, 3.6 and 3.7 will apply only to the operation of the Divested Business, and not to the operation of any businesses or assets (whether or not related to 3.5" HDDs) possessed by Toshiba prior to Toshiba's acquisition of the Divested Business, or acquired or developed by Toshiba independently of Toshiba's acquisition of the Divested Business; and
- (c) the sublicense granted to Toshiba will be pursuant to a written agreement, satisfactory in form to Hitachi, obligating Toshiba to observe all applicable terms and conditions of this Agreement with respect to rights granted to Toshiba, and containing confidentiality terms at least as protective of the HITACHI Licensed Trade Secrets, and HITACHI's rights therein, as the terms of Section 3.8 of this Agreement, and further containing terms expressly designating Hitachi as an intended third party beneficiary and providing that Hitachi shall have the right to exercise the same rights and remedies against Toshiba with respect to the sublicense as HITACHI may exercise against Buyer Parent with respect to the applicable provisions of this Agreement. For the avoidance of doubt, the form set forth as Exhibit A is satisfactory to Hitachi as relates to the sublicense pursuant to this Section 3.11.

Notwithstanding anything to the contrary, HITACHI may enforce any rights or seek any remedies arising from any act or omission by Toshiba directly against Toshiba or Buyer Parent; provided, however, that HITACHI will not seek injunctive relief against Buyer Parent, or terminate any rights of Buyer Parent pursuant to this Agreement, due to a breach of the sublicense by Toshiba. If HITACHI is unable to enforce any rights or remedies against Toshiba because of Toshiba's status as a sublicensee under the sublicense agreement (as opposed to a direct licensee of HITACHI), then, Buyer Parent and its Subsidiaries shall reasonably cooperate with HITACHI to enforce such rights or obtain remedies (including, to the extent necessary, Buyer Parent's or its Subsidiaries' joining as a party to any enforcement action or terminating rights granted to Toshiba pursuant to the sublicense agreement). Buyer Parent shall provide a copy of the executed sublicense agreement to HITACHI, and, with respect to those terms relating to the sublicense from HITACHI, shall (i) not amend the sublicense agreement without HITACHI's prior consent, (ii) provide HITACHI with prompt notice of any breach of the sublicense agreement by Toshiba of which Buyer Parent becomes aware, and (iii) not waive any such breach of the sublicense agreement by Toshiba without HITACHI's prior consent.

# 4. TERMS AND TERMINATIONS

4.1 <u>Termination and Survival</u>. The licenses granted in Sections 2.1, 3.2, 3.3, 3.4 and shall terminate at the end of Term. Further, the license grant in Sections 2.2 and 3.1 shall terminate in part: with respect to Section 2.2, only the license under Buyer Parent RAID Patents that are not HGST RAID Patents shall terminate at the end of the Term; with respect to Section 3.1, only the license under Patents

issued or issuing on patent applications entitled to a first effective filing date after the first anniversary of the Closing Date shall terminate at the end of the Term. The remainder of the licenses set forth in Sections 2.2 and 3.1, including their transferability, shall continue and remain in full force and effect until the expiration of the life of the last Patent, as defined herein. In addition, the covenant set forth in Section 2.5 shall terminate (a) at the end of the Term, with respect to Patents issued or issuing on patent applications entitled to a first effective filing date after the first anniversary of the Closing Date and (b) upon the expiration of the life of the last Patent subject to the covenant, as to all other Patents subject to the covenant. The remainder of this Agreement, including the licenses set forth in Sections 3.6 and 3.7 and the provisions of Section 3.8, shall continue and remain in full force for so long as any HITACHI License Copyright or HITACHI Licensed Trade Secret continues to exist and be enforceable.

- 4.2 <u>Termination of the Rights Granted to Subsidiaries</u>. All rights, licenses and covenants granted to any Subsidiaries of either Party shall automatically terminate in the event that such Subsidiaries no longer fall under the definition of "Subsidiary," as defined herein.
- 4.3 <u>Termination For Material Breach</u>. If Buyer Parent or any of its Subsidiaries breaches Section 2.5 of this Agreement, and if such material breach is not corrected within sixty (60) days after written notice by HITACHI, the continuing covenants, rights, licenses, and any other privileges granted to Buyer Parent and its Subsidiaries in this Agreement may be terminated by HITACHI forthwith by written notice to Buyer Parent; provided that no such termination shall terminate, modify or limit, in any manner, any of the continuing covenants, rights, licenses, and any other privileges granted to HITACHI and the HITACHI Sublicensed Subsidiaries in this Agreement, all of which shall continue unaffected.
- 4.4 <u>Termination For Bankruptcy</u>. A Party may terminate this Agreement by giving written notice of termination to the other Party at any time upon or after:
  - (a) the filing by the other Party of an petition in bankruptcy or insolvency which petition or proceeding is not dismissed within sixty (60) days;
  - (b) any adjudication that the other Party is bankrupt or insolvent;
  - (c) the filing by the other Party of any petition or answer seeking reorganization, readjustment or arrangement of its business under any law relating to bankruptcy or insolvency which petition or proceeding is not dismissed within sixty (60) days;
  - (d) the appointment of a receiver for all or substantially all of the property of the other Party;
  - (e) the institution of any proceeding for the liquidation winding up of the other Party's business which petition or proceeding is not dismissed within sixty (60) days.

Notwithstanding the foregoing, the occurrence of any of the events provided above shall not be deemed to terminate, modify or limit, in any manner (or permit a Party to terminate, modify or limit, in any manner), any of the continuing covenants, rights, licenses and any other privileges granted to the terminating Party and its Subsidiaries in this Agreement, all of which shall continue unaffected by such event.

4.5 <u>No Release on Termination</u>. No termination in this Agreement pursuant to Section 4.2, 4.3, or 4.4 above shall relieve either Party of any obligation or liability accrued hereunder prior to such termination.

### 5. REPRESENTATIONS AND WARRANTIES

- 5.1 <u>Authority</u>. Each Party represents and warrants that it has the corporate power and authority to enter this Agreement, and to carry out the terms and obligations set forth in this Agreement, and that the persons executing this Agreement on its behalf have the authority to act for and bind the Party and/or, in case of Buyer Parent, its Subsidiaries, and in case of HITACHI, HITACHI Sublicensed Subsidiaries.
- 5.2 <u>Buyer Parent's Right To Grant</u>. Buyer Parent, on behalf of itself and its Subsidiaries, represents and warrants that Buyer Parent (or a Subsidiary of Buyer Parent) owns, and that Buyer Parent has the right to license, the Patents licensed by Buyer Parent; and that it is not a party to any agreements or obligations inconsistent with this Agreement.
- 5.3 <u>HITACHI's Right To Grant</u>. HITACHI, on behalf of itself and HITACHI Sublicensed Subsidiaries, represents and warrants that HITACHI (or a HITACHI Sublicensed Subsidiary) owns, and that HITACHI has the right to license, the Patents licensed by HITACHI; and that it is not a party to any agreements or obligations inconsistent with this Agreement.
- 5.4 <u>No Warranty by Buyer Parent</u>. This Agreement does not and shall not be interpreted or construed to include: (a) any warranty or representation on the part of Buyer Parent as to the validity, enforceability, or scope of any Patent licensed by Buyer Parent; (b) any warranty or representation that any specific apparatus or method used by HITACHI in connection with any Patents licensed by Buyer Parent is or will be free from infringement of patents of others or other intangible rights of third parties; (c) any requirement to file any patent application, or secure or maintain any patent; (d) any obligation to bring or prosecute any action for infringement of any Patent licensed by Buyer Parent; (e) any obligation to furnish any technical or support information; (f) any license or right by implication or estoppel; or (g) any warranty of MERCHANTABILITY, NON-INFRINGEMENT OR FITNESS FOR ANY PARTICULAR PURPOSE OR USE for implementations of Patents licensed by Buyer Parent.

5.5 <u>No Warranty by HITACHI</u>. This Agreement does not and shall not be interpreted or construed to include: (a) any warranty or representation on the part of HITACHI as to the validity, enforceability, or scope of any Patent licensed by HITACHI; (b) any warranty or representation that any specific apparatus or method used by Buyer Parent or its Subsidiaries in connection with any Patent licensed by HITACHI is or will be free from infringement of patents of others or other intangible rights of third parties; (c) any requirement to file any patent application, or secure or maintain any patent; (d) any obligation to bring or prosecute any action for infringement of any Patent licensed by HITACHI; (e) any obligation to furnish any technical or support information; (f) any license or right by implication or estoppel; or (g) any warranty of MERCHANTABILITY, NON-INFRINGEMENT OR FITNESS FOR ANY PARTICULAR PURPOSE OR USE for implementations of the Patents licensed by HITACHI.

### 6. NOTICES

All notices and other communications required or permitted hereunder shall be in writing and shall be mailed by Federal Express or other internationally recognized express carrier, registered or certified mail, postage prepaid, or otherwise delivered by hand or by messenger, addressed to a Party at the addresses set forth below, or at such other address as a Party may substitute by written notice provided to the other Party in such manner. Such notices shall be deemed to have been served when delivered.

If to HITACHI:

IP Business Division Intellectual Property Group Hitachi, Ltd. Atten: General Manager 6-1 Marunouchi 1-chome, Chiyoda-ku Tokyo 100-8220 Japan

If to Buyer Parent:

### 7. BUSINESS ACQUISITION, TRANSFER, CHANGE OF CONTROL, ETC.

- 7.1 <u>Assignment and Change of Control of Buyer Parent HDD Business</u>. Buyer Parent may assign or transfer the rights, licenses, and covenants under Section 3.1 of this Agreement in connection with any sale or other transfer of (i) all or substantially all of Buyer Parent's HDD business, (ii) all or substantially all of Buyer Parent's HDD-related assets in a single transaction or a series of transactions, or (iii) any transaction under which Buyer Parent undergoes a change in control.
- 7.2 <u>Assignment of Buyer Parent Non-HDD Business</u>. In connection with any sale or other transfer of (a) all or substantially all of Buyer Parent's RAID, External Drive, and/or SSD business, or (b) all or substantially all of Buyer Parent's RAID-, External Drive-and/or SSD-related assets in a single transaction or a series of transactions, Buyer Parent may assign or transfer the rights, licenses and

covenants under Sections 3.2, 3.3, 3.4, 3.5 and/or 3.6 corresponding to the sold or transferred business, subject to the following limitations:

- (a) The rights, licenses, and covenants under Sections 3.2, 3.3, 3.4, 3.5 and/or 3.6 shall be assignable provided that: the acquiring third party ("Acquiring 3rd Party") agrees to grant a royalty-free license to HITACHI and the HITACHI Sublicensed Subsidiaries under the Patents which would fall within the scope of Buyer Parent SSD Patents or Buyer Parent RAID Patents if the Acquiring 3rd Party were named in the place of Buyer Parent in Sections 1.4 and 1.5 hereunder, in each case for the applicable term set forth in Section 4.1;
- (b) The rights, licenses, and covenants under Sections 3.2, 3.3, 3.4, 3.5 and/or 3.6 shall not be extended to the business originally run by the Acquiring 3<sup>rd</sup> Party or any of its Subsidiaries. To the extent that such rights, licenses and covenants continue to extend to the acquired portion of the business, the extension shall be limited to a revenue of such business equal to no more than the revenue of the acquired business in the twelve (12) months preceding such sale or transfer plus twenty percent (20%); and shall also be limited, in each of the successive twelve-month periods following such transfer or sale, to a revenue equal to no more than the immediately preceding twelve-month period plus twenty percent (20%). To the extent that the business exceeds this cap, and the Acquiring 3rd Party desires to obtain further rights and licenses, the Acquiring 3<sup>rd</sup> Party and HITACHI shall discuss a mutually acceptable resolution; and,
- (c) In no event shall the rights, licenses, and covenants under Sections 3.2, 3.3, 3.4, 3.5 and/or 3.6 supersede or relieve any obligation by any party to pay on-going royalties under any existing patent license agreement with respect to business operations of the Acquiring 3<sup>rd</sup> Party.
- (d) The rights, licenses, and covenants under Sections 3.2, 3.3, 3.4, 3.5 and/or 3.6 may not be assigned or transferred by Buyer Parent to any Acquiring 3<sup>rd</sup> Party if such Acquiring 3<sup>rd</sup> Party or any of its Subsidiaries has made, in court or otherwise, a written allegation of patent infringement against HITACHI that existed prior to the completion of the assignment or transfer and has not been resolved prior to the completion of the assignment or transfer.
- 7.3 Impact of Change of Control of Buyer Parent or HGST on Sections 3.2, 3.3, 3.4, 3.5 and 3.6 Licenses. In the event that in a single transaction or in a series of transactions Buyer Parent is acquired by a third party, through a merger, stock sale, asset purchase, or otherwise, HITACHI may, at its option, terminate the rights, licenses, and covenants under Section 3.2, 3.3, 3.4, 3.5 or 3.6 of this Agreement (but not the license under Section 3.1), if and only if the Acquiring 3rd Party or any of its Subsidiaries is a competitor in the RAID business. Buyer Parent shall timely notify HITACHI in writing of the acquisition with the

information reasonably required for HITACHI to determine whether or not to terminate those rights, including the framework and schedule of the acquisition. HITACHI shall notify, in turn, Buyer Parent in writing of its election whether or not to terminate the license within ninety (90) days after receipt of the notice. In the event that HITACHI chooses to terminate the license pursuant to the terms of this provision, HITACHI and Buyer Parent shall have a good faith basis upon which to renegotiate the license. Notwithstanding the foregoing, the rights, licenses and covenants in Sections 3.1 and 3.8 shall not be terminated in any event.

- 7.4 Acquisition of Non-HDD Business by Buyer Parent. In the event that Buyer Parent directly or indirectly acquires a Non-HDD Business pursuant to a merger, asset or stock purchase, or otherwise in a single transaction or a series of transactions during the Term, HITACHI shall have an option not to extend the rights, licenses, releases, and covenants under Section 3 to such Non-HDD Business; provided, however, if HITACHI exercises the option, the Patents owned or controlled by the acquired Non-HDD Business shall not be included in the definitions of Buyer Parent SSD Patents or Buyer Parent RAID Patents. Buyer Parent shall notify HITACHI in writing of the acquisition with information of the acquired Non-HDD Business reasonably required for HITACHI to determine the exercise of the option. HITACHI shall notify, in turn, Buyer Parent in writing of its exercise of the option or not within ninety (90) days after HITACHI received such notice without delay.
- 7.5 <u>Assignment of RAID Business By HITACHI</u>. HITACHI may assign or transfer the rights, licenses and covenants under Section 2.2 in connection with any sale or other transfer of (i) all or substantially all of HITACHI's RAID business or (ii) substantially all of HITACHI's related RAID assets in a single transaction or a series of transactions.
- 7.6 <u>Assignment of SSD Business By HITACHI</u>. HITACHI may assign or transfer the rights, licenses and covenants under Section 2.1 in connection with any sale or other transfer of (i) all or substantially all of HITACHI's SSD business or (ii) substantially all of HITACHI's related SSD assets in a single transaction or a series of transactions during the Term.
- 7.7 <u>Change of Control of HITACHI</u>. In the event that in a single transaction or in a series of transactions HITACHI is acquired by a third party, through a merger, stock sale, asset purchase, or otherwise, this Agreement shall remain effective and shall not be terminated.
- 7.8 <u>Notice of Assignment/Transfer</u>. In the event of a valid assignment or transfer (*i.e.*, one made pursuant to the terms of Section 7.1, 7.2, 7.5 or 7.6 above) by either Party of this Agreement, the transferring or assigning Party shall, within ninety (90) calendar days, provide the other Party notice, in writing, of such assignment or transfer and shall identify the person to whom it has made such assignment or transfer.

- 7.9 <u>Effect of Assignment</u>. Any license, covenants and/or any other rights assigned by either Party pursuant to any provision of this Section 7 shall be effective with respect to the assignee (and its business and operations) on a prospective basis only. Accordingly, no such assignment shall relieve, release or discharge any assignee from any liability for damages accruing prior to the effective date of the applicable assignment due to the infringement of any Patent licensed under this Agreement. Should either Party assign or otherwise transfer this Agreement or any of its respective Patents, such transfer shall not operate to terminate, impair or in any way modify the licenses and releases provided by the Parties herein, except as expressly provided for under the terms of this Agreement.
- 7.10 <u>Assignment of Patents</u>. From and after any assignment or transfer of a Patent by either Party or its Subsidiaries, the assignee shall be bound by the terms and conditions of this Agreement with respect to such Patents.
- 7.11 <u>Sublicense</u>. Neither party, nor their Subsidiaries, may sublicense any rights under this Agreement unless otherwise provided hereunder.

# 8. MISCELLANEOUS

- 8.1 <u>Entire Agreement</u>. This Agreement contains the complete agreement between, and contains all of the promises and undertakings made to each other by, the Parties regarding the subject matter of this Agreement. Any and all prior agreements, representations, negotiations, and undertakings between the Parties, oral or written, express or implied, with respect to the subject matter hereof are hereby superseded and merged by and into this Agreement. This Agreement may not be revised or modified without the mutual written consent of the Parties.
- 8.2 <u>Severability</u>. If any paragraph, provision or clause of this Agreement shall be found or held to be invalid or unenforceable by a court or other decision-making body of competent jurisdiction, in a judgment from which no further appeal can be taken, the remainder of this Agreement shall remain valid and enforceable, and to the extent required in the pursuit of this Agreement, the Parties shall negotiate in good faith a substitute, valid and enforceable provision that reflects the Parties' intent in entering this Agreement.
- 8.3 <u>No Other Rights</u>. Nothing contained in this Agreement shall be construed as conferring any rights by implication, estoppel or otherwise, under any non-patent intellectual property right, or any patent, other than the Patents licensed to Buyer Parent and HITACHI. Neither Party is required hereunder to furnish or disclose to the other any technical or other information (including copies of Patents licensed to Buyer Parent and HITACHI.)
- 8.4 <u>ADR Provision</u>. The Parties agree that, in the event of a dispute arising under this Agreement, the Parties shall agree to attend a nonbinding mediation prior to bringing any action in a court of law. The mediation shall be facilitated by a neutral third-party, selected by the parties. A representative from each Party with

authority to settle on behalf of that party shall attend the mediation in person. The purpose of the mediation is to resolve the disputed issue(s) without the need for formal legal proceedings.

- 8.5 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware as applied to contracts entered into and fully performed in such state.
- 8.6 <u>Jurisdiction</u>. The Parties agree that the Delaware Court of Chancery (and if jurisdiction in the Delaware Court of Chancery shall be unavailable, any Delaware State court and the Federal court of the United States of America sitting in the State of Delaware) shall have exclusive jurisdiction of interpreting or enforcing this Agreement. The parties expressly waive their right to a trial by jury.
- 8.7 <u>Counterparts</u>. This Agreement may be executed in identical counterparts, each of which shall constitute an original and all of which shall constitute one and the same agreement.
- 8.8 <u>Bankruptcy Code</u>. The Parties acknowledge and agree that the rights and licenses granted in this Agreement, including but not limited to Section 2.0, are "intellectual property" as defined in Section 101 (35) of the United States Bankruptcy Code, as the same may be amended from time to time (the "Code"), for purposes of Section 365(n) of the Code, and have been licensed in this Agreement in a contemporaneous exchange for value.
- 8.9 <u>Further Assurances</u>. Each Party hereto shall execute and cause to be delivered to each other Party hereto such instruments and other documents, and shall take such other actions, as such other Party may reasonably request, to effect, evidence, record or perfect any of the licenses and other rights set forth in this Agreement. The requesting Party shall reimburse the other Party for its reasonable out-of-pocket costs in responding to a request under this Section.
- 8.10 <u>Disclosure</u>. The terms of this Agreement shall remain confidential and neither Party shall disclose this Agreement or its terms to any individual or entity, except (a) with the prior written consent of the other Party, (b) to the extent disclosure by a Party is required by law or order of a court of competent jurisdiction, provided that, in such event, such Party shall provide the other Party prompt, advance notice of such requirement to allow intervention (and shall cooperate with the other Party) to contest or minimize the scope of the disclosure (including through application for a protective order) or (c) in confidence by a Party to its auditors or legal counsel in connection with the provision of audit or legal services (respectively) to such Party, to the extent required for the provision of such services.

[Remainder of this page is intentionally left blank]

IN WITNESS WHEREOF, the Parties have caused their duly authorized officers to execute this Agreement on the dates indicated below.

# For and on behalf of

Hitachi Ltd., and HITACHI Sublicensed Subsidiaries,

By:	
Name:	Naoya Takahashi
Title:	Executive Vice President and Executive Officer

For and on behalf of

Western Digital Corporation, a Delaware corporation, and its Subsidiaries,

By:	
Name:	
Title:	

Date:

Date:

### FOURTH AMENDMENT TO STOCK PURCHASE AGREEMENT

This Fourth Amendment to the Stock Purchase Agreement (this "<u>Amendment</u>") is made this 15th day of February, 2012, by and among Western Digital Corporation, a Delaware corporation (the "<u>Buyer Parent</u>"), Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect wholly owned subsidiary of the Buyer Parent (the "<u>Buyer</u>"), Hitachi, Ltd., a company incorporated under the laws of Japan (the "<u>Seller</u>"), and Viviti Technologies Ltd., a company incorporated under the laws of the Republic of Singapore and a wholly owned subsidiary of the Seller (the "<u>Company</u>", and collectively with the Seller, Buyer and Buyer Parent, the "<u>Parties</u>", and each, a "<u>Party</u>"). Capitalized terms used herein and not otherwise defined shall have the meanings set forth in the Stock Purchase Agreement (as defined below).

WHEREAS, the Parties entered into a Stock Purchase Agreement on March 7, 2011, as amended by a First Amendment to Stock Purchase Agreement, dated May 27, 2011, as further amended by a Second Amendment to Stock Purchase Agreement, dated November 23, 2011, and as further amended by a Third Amendment to Stock Purchase Agreement, dated January 30, 2012 (together, the "Stock Purchase Agreement"); and

WHEREAS, the Parties desire to amend the Stock Purchase Agreement as reflected herein.

NOW, therefore, in exchange for good and valuable consideration, the receipt of which is hereby acknowledged, the Parties hereby agree as follows:

1. **Schedule 7.8 of the Stock Purchase Agreement**. Schedule 7.8 of the Stock Purchase Agreement, Retained Land, is hereby deleted in its entirety and replaced by the Schedule 7.8 attached hereto as <u>Exhibit A</u>.

2. **Effect on the Stock Purchase Agreement.** This Amendment shall not constitute a waiver, amendment or modification of any provision of the Stock Purchase Agreement not expressly referred to herein. Except as expressly amended or modified herein, the provisions of the Stock Purchase Agreement are and shall remain in full force and effect and are hereby ratified and confirmed. On and after the date hereof, each reference in the Stock Purchase Agreement to "this Agreement", "herein", "hereof", "hereunder" or words of similar import shall mean and be a reference to the Stock Purchase Agreement as amended hereby, although it shall not alter the dates as of which any provision of the Stock Purchase Agreement speaks. For example, phrases such as "as of the date hereof" and "as of the date of this Agreement" shall continue to refer to March 7, 2011, the date that the Stock Purchase Agreement, such provision of this Amendment conflicts with or differs from a provision of the Stock Purchase Agreement, such provision of this Amendment shall prevail and govern for all purposes and in all respects.

3. Miscellaneous. Sections 12.4, 12.5, 12.7 and 12.13 of the Stock Purchase Agreement are incorporated herein by reference.

**"BUYER PARENT"** 

### WESTERN DIGITAL CORPORATION

By: /s/ Michael C. Ray

Name:Michael C. RayTitle:SVP, GC & Secretary

#### **"BUYER"**

### WESTERN DIGITAL IRELAND, LTD.

By: /s/ Michael C. Ray Name: Michael C. Ray Title: VP

### "SELLER"

# HITACHI, LTD.

By: /s/ Toyoki Furuta Name: Toyoki Furuta Title: General Manager, Business Development Office

### "COMPANY"

# VIVITI TECHNOLOGIES LTD.

By: /s/ Christopher Dewees Name: Christopher Dewees Title: SVP & General Counsel

Signature Page to Fourth Amendment to Stock Purchase Agreement

### FIFTH AMENDMENT TO STOCK PURCHASE AGREEMENT

This Fifth Amendment to the Stock Purchase Agreement (this "<u>Amendment</u>") is made this 6<sup>th</sup> day of March, 2012, by and among Western Digital Corporation, a Delaware corporation (the "<u>Buyer Parent</u>"), Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect wholly owned subsidiary of the Buyer Parent (the "<u>Buyer</u>"), Hitachi, Ltd., a company incorporated under the laws of Japan (the "<u>Seller</u>"), and Viviti Technologies Ltd., a company incorporated under the laws of the Seller (the "Company", and collectively with the Seller, the Buyer and the Buyer Parent, the "<u>Parties</u>", and each, a "<u>Party</u>"). Capitalized terms used herein and not otherwise defined shall have the meanings set forth in the Stock Purchase Agreement (as defined below).

WHEREAS, the Parties entered into a Stock Purchase Agreement on March 7, 2011, as amended by a First Amendment to Stock Purchase Agreement, dated May 27, 2011, as further amended by a Second Amendment to Stock Purchase Agreement, dated November 23, 2011, as further amended by a Third Amendment to Stock Purchase Agreement, dated January 30, 2012, and as further amended by a Fourth Amendment to Stock Purchase Agreement, dated February 15, 2012 (together, the "Stock Purchase Agreement");

WHEREAS, concurrent with the execution of this Amendment, the Buyer Parent, the Seller and the Company have executed and delivered Amendment No. 2 to the Transition Services Agreement; and

WHEREAS, the Parties desire to amend the Stock Purchase Agreement and certain other Transaction Documents as reflected herein.

NOW, therefore, in exchange for good and valuable consideration, the receipt of which is hereby acknowledged, the Parties hereby agree as follows:

**1. Section 1.1 of the Stock Purchase Agreement**. The definition of "Outstanding Company Debt" set forth in Section 1.1 of the Stock Purchase Agreement is hereby amended and restated in its entirety to read as follows:

"<u>Outstanding Company Debt</u>" means all short and long-term debt, accounts receivable related borrowings and lines of credit of the Company or any of its Subsidiaries outstanding as of the Closing Date, other than intercompany indebtedness between the Company and any of its Subsidiaries or between any of its Subsidiaries, and other than any debt outstanding under (i) that certain Credit Line Agreement between Hitachi Global Storage Technologies (Shenzhen) Co., Ltd. and Agricultural Bank of China, dated December 22, 2011, and/or (ii) that certain Credit Line Agreement between Hitachi Global Storage Products (Shenzhen) Co., Ltd. and Agricultural Bank of China, dated December 22, 2011."

**2. Section 2.3(a) of the Stock Purchase Agreement**. Section 2.3(a) of the Stock Purchase Agreement is hereby amended and restated in its entirety to read as follows:

"At the Closing, the Buyer shall make the following payments by wire transfer of immediately available funds for the benefit of the Persons and to the accounts and

in the amounts specified on a spreadsheet containing the items listed below in this <u>Section 2.3(a)(</u>the "<u>Consideration Spreadsheet</u>"), to be prepared by the Company not less than three (3) Business Days prior to the Closing:

(i) <u>Unpaid Transaction Expenses</u>. The Buyer shall pay, from the Cash Portion of the Purchase Price, the Transaction Expenses not paid by the Company or any of the Subsidiaries prior to the Closing;

(ii) <u>Unpaid Change of Control Payments</u>. The Buyer shall pay, from the Cash Portion of the Purchase Price, an aggregate amount equal to the Change of Control Payments not paid by the Company or any of the Subsidiaries prior to the Closing to Hitachi Global Storage Technologies Netherlands B.V. ("<u>HGST BV</u>") for further distribution to the applicable Subsidiaries, who will then make subsequent distributions to the applicable employees in accordance with such Subsidiary's normal payroll processes;

(iii) <u>Unpaid Employee Payments</u>. The Buyer shall pay, from the Cash Portion of the Purchase Price, an aggregate amount equal to the Employee Payments not paid by the Company or any of the Subsidiaries prior to the Closing to (A) HGST BV for further distribution to the applicable Subsidiaries, who will then make subsequent distributions to the applicable employees in accordance with such Subsidiary's normal payroll processes and/or (B) the Company in accordance with the Company's normal payroll processes;

(iv) <u>Unpaid Outstanding Company Debt</u>. The Buyer shall pay, from the Cash Portion of the Purchase Price, all Outstanding Company Debt, to the extent not paid by the Company or any of the Subsidiaries prior to the Closing;

(v) <u>Equity Award Payments</u>. The Buyer shall pay, from the Cash Portion of the Purchase Price, an aggregate amount equal to the Equity Award Payments not paid by the Company or any of the Subsidiaries prior to the Closing to (A) HGST BV for further distribution to the applicable Subsidiaries, who will then subsequently make such Equity Award Payments, less applicable withholding Taxes, to the respective Vested Equity Holders in accordance with such Subsidiary's normal payroll process and/or (B) the Company, who will then subsequently make such Equity Award Payments, less applicable withholding Taxes, to the respective Vested Equity Holders in accordance with the Company's normal payroll process;

(vi) <u>Seller</u>. The Buyer shall pay to the Seller from the Cash Portion of the Purchase Price, an amount equal to the sum of (A) the product of (x) the total number of issued and paid-up shares of the Stock owned by the Seller at the Closing and (y) the Per Share Closing Payment less (B) the Stock Portion of the Purchase Price;

provided, however, that neither the Company nor any of the Subsidiaries will distribute any unpaid Change of Control Payments, unpaid Employee Payments or Equity Award

Payments to the individuals listed on <u>Schedule 1.1(a)</u> prior to the eighth (8<sup>th</sup>) day following such individual's execution of a waiver and release of claims against the Company, substantially in the form prescribed by the Company attached hereto as Exhibit H (together with the related clarification included in <u>Exhibit H</u>)."

## 3. Amendments to the Stock Purchase Agreement Related to Secondment.

## (a) Section 6.12(e)(i) of the Stock Purchase Agreement.

(1) The number of R&D Secondees defined as "up to 113 employees of the Seller" in Section 6.12(e)(i) of the Stock Purchase Agreement is hereby amended to state "up to 102 employees of the Seller".

(2) <u>Schedule 6.12(e)(i)</u> of the Stock Purchase Agreement (Seller Seconded R&D Employees) is hereby deleted in its entirety and replaced by the <u>Schedule 6.12(e)(i)</u> attached hereto as <u>Exhibit A</u>.

## (b) Section 6.12(e)(ii) of the Stock Purchase Agreement.

(1) The Non-R&D/R&D Related Secondees defined as "up to 26 employees of the Seller" in Section 6.12(e)(ii) of the Stock Purchase Agreement is hereby amended to state "up to 31 employees of the Seller and its Related Subsidiaries".

(2) <u>Schedule 6.12(e)(ii)</u> of the Stock Purchase Agreement (Seller Seconded Non-R&D Employees) is hereby deleted in its entirety and replaced by the <u>Schedule 6.12(e)(ii)</u> attached hereto as <u>Exhibit B</u>.

(3) The reference to "either the Seller or HGST Japan" in the proviso of Section 6.12(e)(ii) of the Stock Purchase Agreement is hereby amended to state "either the Seller or its Related Subsidiary, as applicable, or HGST Japan".

(c) Section 6.12(e)(iii) of the Stock Purchase Agreement. Each of the two references to "the Seller" in the second sentence of Section 6.12(e) (iii) is hereby amended to state "the Seller or its Related Subsidiary, as applicable,".

#### (d) Section 6.12(f) of the Stock Purchase Agreement.

(1) The HGST Japan Secondees defined as "up to 82 employees of HGST Japan to Seller" in Section 6.12(f) of the Stock Purchase Agreement is hereby amended to state "up to 60 employees of HGST Japan to Seller and its Related Subsidiaries, as applicable".

(2) <u>Schedule 6.12(f)</u> of the Stock Purchase Agreement (HGST Japan Secondees) is hereby deleted in its entirety and replaced by the <u>Schedule 6.12(f)</u> attached hereto as <u>Exhibit C</u>.

(3) Each of the two references to "the Seller" in the third sentence of Section 6.12(f) is hereby amended to state "the Seller or its Related Subsidiary, as applicable,".

**4. Amendment to Secondment Periods in the Stock Purchase Agreement**. Each of the R&D Secondment Period, the Non-R&D/R&D Related Secondment Period and the HGST Japan Secondment Period as defined in Sections 6.12(e)(i), 6.12(e)(ii) and 6.12(f), respectively,

of the Stock Purchase Agreement is hereby amended such that the initial period of secondment shall commence on the Closing Date (unless otherwise indicated on <u>Schedule 6.12(e)(i)</u>, <u>Schedule 6.12(e)(ii)</u> or <u>Schedule 6.12(f)</u> and terminate as of the close of business on March 31, 2014.

**5. Section 6.12(h)(ii) of the Stock Purchase Agreement**. Section 6.12(h)(ii) of the Stock Purchase Agreement is hereby amended to delete the last sentence of such section, which reads: "The Buyer shall use commercially reasonable efforts to provide for remittance of continued payroll deductions to service any participant loan balance maintained by a Continuing Employee under the Seller 401(k) Plan through the end of the calendar quarter following the calendar quarter in which the Closing occurs (or through the date of distribution of such Continuing Employee's account under the Seller 401(k) Plan, if earlier)."

#### 6. Schedule 1.1(a) to the Stock Purchase Agreement.

(a) Item 3 of Schedule 1.1(a) is hereby amended and restated in its entirety to read as follows:

"3. Severance and Change in Control Agreement, dated September 27, 2010, by and between the Company and Steven G. Campbell, as contemplated to be amended pursuant to <u>Schedule 6.1</u>.\* \*\*"

(b) Item 4 of Schedule 1.1(a) is hereby amended and restated in its entirety to read as follows:

"4. Severance and Change in Control Agreement, dated September 27, 2010, by and between the Company and Michael D. Cordano, as contemplated to be amended pursuant to <u>Schedule 6.1</u>.\* \*\*\*?

**7. Exhibit A to the Stock Purchase Agreement**. Exhibit B (Key HGST Employees) to Exhibit A to the Stock Purchase Agreement, Form of Agreement Not to Compete, is hereby amended and restated in its entirety to read as set forth on <u>Exhibit D</u> hereto.

**8.** Exhibit B to the Stock Purchase Agreement. Section 1.1 of Exhibit B to the Stock Purchase Agreement, Form of License Agreement, is hereby amended and restated in its entirety to read as follows:

"External Drive" means an enclosure that encases one or more HDDs or SSDs and related peripheral hardware, or Software or Firmware components (other than RAID). For the avoidance of doubt, "External Drive" includes Information Versatile Disk for Removable Usage (iVDR) enclosures that encase one or more HDDs or SSDs and related peripheral hardware, or Software or Firmware components (other than RAID)."

## 9. Exhibit C to the Stock Purchase Agreement.

(a) Section 2.01 of Exhibit C to the Stock Purchase Agreement, Form of Investor Rights Agreement, is hereby amended and restated in its entirety to read as follows:

"**Investor Designee Appointment and Nomination Right**. The Investor shall have the right, but not the obligation, to designate two nominees to serve as

directors of the Company (each, an "Investor Designee" and, together, the "Investor Designees"). In the event the Investor determines to designate the initial Investor Designees, the Investor shall notify the Company in writing of the names of the initial Investor Designees. Promptly following receipt by the Company of all documentation reasonably requested by the Company in connection with the appointment of the initial Investor Designees, the Company shall increase the size of the Board by two, and fill the resulting vacancies with the initial Investor Designees in accordance with the Company's Bylaws. Thereafter, the Company shall (a) include the Investor Designees in its slate of nominees for election to the Board of Directors at each annual or special meeting of stockholders of the Company following the Closing at which directors are to be elected and at which the seats held by the Investor Designees are subject to election (such annual or special meetings, the "Election Meetings") and (b) recommend that the Company's stockholders vote in favor of the election of the Investor Designees, support the Investor Designees for election in a manner no less favorable than the manner in which the Company supports its other nominees, and otherwise use commercially reasonable efforts to cause the election of the Investor Designees to the Board of Directors at each of the Election Meetings. The foregoing appointment and nomination rights will be subject to the Investor Designees satisfying the Company's Board Qualifications (as defined in Section 2.03); provided that, if an Investor Designee does not meet the Board Qualifications, (i) the Company will not nominate a replacement candidate in place of the rejected Investor Designee (unless the Investor does not nominate a replacement candidate pursuant to its rights in the following clause (ii) within the time period stated in such clause), and (ii) the Investor shall have the right (if exercised as promptly as reasonably practicable and in any event within 30 days) to nominate a replacement candidate in place of the rejected Investor Designee until such time as an Investor Designee that meets the Board Qualifications is put forward by the Investor."

(b) The list attached as Exhibit E hereto is the "Competitor List Letter" as such term is defined in the Form of Investor Rights Agreement attached as Exhibit C to the Stock Purchase Agreement.

**10. Exhibit E to the Stock Purchase Agreement**. Exhibit E to the Stock Purchase Agreement, Form of R&D Services Agreement, is hereby amended and restated in its entirety to read as set forth on <u>Exhibit F</u> hereto.

**11. Exhibit F to the Stock Purchase Agreement**. Exhibit F to the Stock Purchase Agreement, Form of Branding Agreement, is hereby amended and restated in its entirety to read as set forth on <u>Exhibit G</u> hereto.

**12.** Exhibit G to the Stock Purchase Agreement. Exhibit G to the Stock Purchase Agreement, Form of Secondment Agreement, is hereby amended and restated in its entirety to read as set forth on Exhibit H hereto.

**13.** Exhibit H to the Stock Purchase Agreement. The Stock Purchase Agreement is hereby amended by adding Exhibit H as set forth on Exhibit I hereto.

14. Exhibit I to the Stock Purchase Agreement.

(a) The Stock Purchase Agreement is hereby amended by adding <u>Exhibit I</u>, <u>Form of Reverse Services Agreement</u>, as set forth on <u>Exhibit J</u> hereto.

(b) Section 8.8 of the Stock Purchase Agreement. A new Section 8.8 is hereby added to the Stock Purchase Agreement as follows:

"Section 8.8 Reverse Services Agreement. The Seller shall have received the reverse services agreement, in the form attached as <u>Exhibit I</u>, executed by the Company."

**15. Effect on the Stock Purchase Agreement.** This Amendment shall not constitute a waiver, amendment or modification of any provision of the Stock Purchase Agreement not expressly referred to herein. Except as expressly amended or modified herein, the provisions of the Stock Purchase Agreement are and shall remain in full force and effect and are hereby ratified and confirmed. On and after the date hereof, each reference in the Stock Purchase Agreement as amended hereby, although it shall not alter the dates as of which any provision of the Stock Purchase Agreement speaks. For example, phrases such as "as of the date hereof" and "as of the date of this Agreement" shall continue to refer to March 7, 2011, the date that the Stock Purchase Agreement was originally executed. To the extent that a provision of this Amendment conflicts with or differs from a provision of the Stock Purchase Agreement, such provision of this Amendment shall prevail and govern for all purposes and in all respects.

16. Miscellaneous. Sections 12.4, 12.5, 12.7 and 12.13 of the Stock Purchase Agreement are incorporated herein by reference.

[Signature page follows]

IN WITNESS WHEREOF, the Parties hereto have executed this Fifth Amendment to Stock Purchase Agreement as of the date first written above.

## **"BUYER PARENT"**

## WESTERN DIGITAL CORPORATION

By: /s/ Michael Ray Michael Ray

Senior Vice President, General Counsel and Secretary

## **"BUYER"**

#### WESTERN DIGITAL IRELAND, LTD.

By: /s/ Michael Ray

Michael Ray Vice President

## "SELLER"

## HITACHI, LTD.

By: /s/ Toyoki Furuta Name: Toyoki Furuta Title: General Manager Business Development Office

#### "COMPANY"

#### VIVITI TECHNOLOGIES LTD.

By: /s/ Christopher S. Dewees

Name:Christopher S. DeweesTitle:SVP

Signature Page to Fifth Amendment to Stock Purchase Agreement

# EXHIBIT F

# EXHIBIT E TO THE STOCK PURCHASE AGREEMENT Form of R&D Services Agreement

[see attached]

Form of R&D Services Agreement

#### FORM OF

#### **R&D SERVICES AGREEMENT**

This R&D Services Agreement ("**Agreement**") effective as of the Closing Date (as defined below), is entered into by and between Western Digital Corporation, a Delaware company with its principal place of business at 3355 Michelson Drive, Suite 100 Irvine, California 92612 ("**Company**") and Hitachi, Ltd. ("**Hitachi**"), a Japanese company with its principal place of business at 6-6, Marunouchi 1- chome, Chiyoda-ku, Tokyo 100-8280, Japan. Each of the signatories to this Agreement is referred to as a "**Party**," and jointly as the "**Parties**."

#### **RECITALS**

WHEREAS, Company, Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect wholly-owned subsidiary of the Company ("**Buyer** "), Hitachi and Viviti Technology Ltd., a company incorporated under the laws of the Republic of Singapore ("**Viviti**") have entered into that certain Stock Purchase Agreement dated as of March 7, 2011 (the "**Purchase Agreement**");

WHEREAS, prior to the Closing Date, Hitachi has performed, from time to time, certain research and development services for the Viviti; and

WHEREAS, Company and Hitachi desire that following the Closing Date Hitachi will continue to provide certain research and development services to Company and its Subsidiaries relating to HDD (as defined below), of a nature similar to those provided to Viviti prior to the Closing Date, pursuant to the terms and conditions set forth herein;

NOW, THEREFORE, in consideration of the mutual promises and covenants set out herein, the Parties agree as follows:

#### **TERMS OF AGREEMENT**

#### 1. DEFINITIONS

- 1.1 **"Background Intellectual Property Rights"** means Intellectual Property Rights, if any, owned by Hitachi that are (a) embodied in Background Technology, and (b) not Developed Intellectual Property Rights; *provided, however*, that a Patent will be a Background Intellectual Property Right only if Hitachi has the right to grant to Company a license of the scope set forth in Section 4.2 under such Patent without incurring any obligation to pay any royalty or other consideration to any third party.
- 1.2 **"Background Technology**" means Technology, if any, owned by Hitachi or its Subsidiaries that is (a) embodied in any Deliverable, and (b) not Developed Technology.

- 1.3 "Closing Date" has the meaning set forth in the Purchase Agreement.
- 1.4 **"Deliverables**" means the deliverables, if any, to be provided by Hitachi as part of the Services performed hereunder, as specifically described in an applicable Project Agreement.
- 1.5 **"Developed Intellectual Property Rights**" means Intellectual Property Rights to the extent such Intellectual Property Rights (i) are first created by Hitachi in the course of its performance of the Services pursuant to the applicable Project Agreement and within the scope and during the term of such applicable Project Agreement, and (ii) are embodied in Developed Technology. For the avoidance of doubt, "Developed Intellectual Property Rights" includes the right to seek Patent protection for inventions that constitute Developed Intellectual Property, if any, but does not include any Patents or Patent applications of Hitachi or its Subsidiaries.
- 1.6 "Developed IP" means, collectively, Developed Technology and Developed Intellectual Property Rights.
- 1.7 **"Developed Technology**" means Technology embodied in any Deliverable to the extent such Technology is first developed or created by Hitachi in the course of its performance of the Services pursuant to the applicable Project Agreement and within the scope and during the term of such applicable Project Agreement.
- 1.8 **"Fully Burdened Cost"** means all of Hitachi's costs and expenses incurred in connection with the performance of the Services, including labor and material costs, expenses, and general and administrative expenses and overhead, all as determined in accordance with generally accepted accounting principles as consistently applied by Hitachi.
- 1.9 **"HDD**" has the meaning set forth in the License Agreement.
- 1.10 **"Intellectual Property Rights**" means Patents, copyrights, and rights with respect to trade secrets, whether arising under the laws of the United States, Japan or any other jurisdiction, including, in each case, any rights apply for, register, and enforce any of the foregoing. Notwithstanding the foregoing, "Intellectual Property Rights" does not include any trademark rights or similar rights with respect to indicia of source or origin.
- 1.11 "License Agreement" means that certain License Agreement entered into by and between Company and Hitachi as of the Closing Date.
- 1.12 **"Patents**" means all classes and types of patents, including utility patents, utility models, design patents, invention certificates, including divisionals, continuations, continuations-in-part, reexaminations, reissues, extensions and renewals, in all jurisdictions of the world.

- 1.13 "**Project Coordinator(s)**" means the Hitachi representative(s) and Company representative(s) identified as project coordinator(s) in the applicable Project Agreement.
- 1.14 **"Services**" means the research and development services to be performed by Hitachi hereunder, as specifically described in an applicable Project Agreement.
- 1.15 **"Software or Firmware**" means a set of instructions, that either (i) directly provides instructions to the computer hardware, or, (ii) indirectly serves as an input to another piece of software.
- 1.16 **"Subsidiary**" of either Party shall mean a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities or other interests having ordinary voting power for the election of directors or other governing body are at the time beneficially owned, directly, or indirectly, by such Party. For the purposes of this Agreement, Viviti and its Subsidiaries will be Subsidiaries of Company (subject to meeting the requirements of the preceding sentence) and are not Subsidiaries of Hitachi.
- 1.17 **"Technology**" means inventions, know-how, designs, specifications, Software or Firmware and other copyrightable material, technical information, devices, and other developments and technology.
- 1.18 "**Term**" means the period beginning on the Closing Date and ending eighteen (18) months from the Closing Date.

## 2. SCOPE OF SERVICES

2.1 **Project Agreements.** Any services and deliverables to be performed or provided by Hitachi hereunder with respect to any Project (as defined below) shall be mutually agreed and described in a written Project Agreement entered into by the Parties hereunder and specifically referencing this Agreement ("**Project Agreements**"). As soon as practicable following the Closing Date, the Parties shall use commercially reasonable efforts to enter into a Project Agreement for each research and development project listed in <u>Schedule 1</u> ("**Project**"), which such Project Agreements shall describe the applicable Project in reasonable detail, contain terms and conditions applicable to such Project that are materially consistent with the terms and conditions for each such Project as set forth in <u>Schedule 2</u> hereto, and set forth in sufficient detail (i) the type, scope, and nature of the Services and Deliverables to be performed or provided by Hitachi with respect to such Project, (ii) specific staffing requirements and schedule, if applicable, for such Project, (iii) the Project Coordinators for such Project, and (iv) such other matters as the Parties may mutually agree. In addition, Hitachi will use commercially reasonable efforts to identify in the applicable Project Agreement any Patents owned by Hitachi and claiming Background Technology Hitachi expects to be embodied in Deliverables to be delivered to Company in the course of a Project under which Hitachi does not have the right to grant to

company a license of the scope set forth in Section 4.2 without incurring an obligation to pay a royalty or other consideration to a third party. All Project Agreements shall be an integral part of this Agreement and be subject to the terms and conditions set forth herein.

- 2.2 **Projects.** The Projects currently contemplated by the Parties are described in <u>Schedule 1</u> hereto. The Parties may mutually agree to add additional Projects\_directly related to the business of Company during the Term. Unless otherwise expressly agreed in writing by the Parties, all Projects will be directed to HDDs. If the Parties agree in writing to undertake a Project directed in whole or in part to any product or technology other than HDDs, the Parties will discuss and agree in the Project Agreement for such Project any additional or revised terms that will apply to such Project, including without limitation with respect to ownership and licenses of associated Developed IP and/or Background IP.
- 2.3 **Services**. Hitachi shall use commercially reasonable efforts to perform or cause to be performed for Company the Services, and to deliver to Company the Deliverables, as set forth in the applicable Project Agreements. Hitachi shall perform such Services in a manner that is consistent with its past practice in regards to its performance of similar services for Viviti prior to the Closing Date (**"Past Practice"**).
- 2.4 **Personnel and Project Coordinators.** Hitachi will assign employees and subcontractors with suitable qualifications to perform the Services consistent with Past Practice. Hitachi may replace or change employees and subcontractors as required. The Parties' Project Coordinators will be responsible for exchanging information, coordinating meetings, and arranging all other matters pertinent to the applicable Project. Each Party may change its Project Coordinator by giving written notice to the other Party. The Project Coordinators are not authorized to modify or change any term or condition of this Agreement, including any Project Agreement.
- 2.5 **Company's Cooperation**. Company acknowledges that its timely provision of reasonable assistance, cooperation, and complete and accurate information and data ("**Cooperation**") is essential to the performance of the Services, and that Hitachi shall not be responsible for any deficiency in performing the Services if such deficiency results from Company's failure to provide such Cooperation as required hereunder.
- 2.6 **Quarterly Project Review**. Company shall have the right to review and assess the continuing need for all Projects on a quarterly basis and may terminate one or more active Projects, and the Project Agreements associated therewith, effective upon the beginning of any calendar quarter (i.e., January 1, April 1, July 1, or October 1), provided that Company has provided Hitachi at least forty-five (45) days' advance notice of such termination.

## 3. PAYMENTS

- 3.1 **Fees**. Unless otherwise specified in the applicable Project Agreement, all Services shall be provided on a time-and-materials basis at Hitachi's Fully Burdened Cost (such Fully Burdened Costs and any other fees and payments specified in an applicable Project Agreement, the "**Fees**").
- 3.2 **Expenses**. Company shall reimburse Hitachi for all reasonable travel, lodging, communications, shipping, and other out-of-pocket expenses incurred by Hitachi in connection with providing the Services and Deliverables ("**Expenses**").
- 3.3 **Payment Terms**. Hitachi will invoice Company on a monthly basis for all Fees and Expenses and other payments due under this Agreement and any Project Agreement and, unless otherwise specified in the applicable Project Agreement, Company shall pay such invoiced amounts within thirty (30) days of the date of the invoice. Customer agrees to pay interest at the rate of one and one-half percent (1.5%) per month (or the maximum rate permitted by applicable law, whichever is less) for all amounts not paid within thirty (30) days from the date of the invoice therefor. All payments shall be made in Japanese Yen by wire transfer in immediately available funds to an account designated by Hitachi.
- 3.4 **Taxes**. In addition to all Fees, Expenses and other amounts payable under this Agreement and the Project Agreements, Company shall pay or reimburse Hitachi for all federal, state, local or other taxes, including, without limitation, sales, use, excise, withholding, and property taxes, or amounts levied in lieu thereof, based on the Services provided or amounts payable under this Agreement or any Project Agreement. Company shall have no responsibility for taxes imposed on Hitachi's net income.

#### 4. INTELLECTUAL PROPERTY

4.1 **Developed IP.** Subject to Company's compliance with the terms and conditions of this Agreement, including payment of all Fees, Expenses, and other amounts payable hereunder, Hitachi hereby assigns and agrees to assign all of its right, title and interest in and to the Developed IP to Company. Company shall have the sole right to apply for, file, register, or otherwise seek Intellectual Property Rights with respect to Developed IP, including the right to seek Patent protection for inventions that constitute Developed IP, if any. Hitachi will provide (and will use commercially reasonable efforts to cause any inventors of Developed IP in Hitachi's or its Subsidiaries' employ to provide) reasonable information and assistance, at Company's cost and expense, to effect the assignment of rights pursuant to this Section 4.1. With respect to any Developed Intellectual Property Rights, a worldwide, non-exclusive, perpetual and irrevocable license to use such Developed Technology in the ordinary course of its business. With respect to any Patents within the Developed Intellectual Property Rights, Hitachi reserves and Company hereby

grants and agrees to grant to Hitachi and its Subsidiaries a worldwide, non-exclusive, perpetual and irrevocable license to make, have made, use, sell, offer for sale, and import any article of manufacture or composition of matter and practice any method or process.

- 4.2 **Background IP**. To the extent, if any, that any Background Technology is embodied in any Deliverables provided to Company under this Agreement, subject to Company's compliance with the terms and conditions of this Agreement, including payment of all Fees, Expenses, and other amounts payable hereunder, Hitachi hereby grants and agrees to grant to Company and its Subsidiaries, under Hitachi's Background Intellectual Property Rights, a worldwide, non-exclusive, perpetual, and irrevocable license to use such Background Technology, solely as embodied in such Deliverables, in the ordinary course of its business.
- 4.3 **No Other Rights**. Except as expressly set forth in this Agreement, neither Party grants any rights in or to its Technology or Intellectual Property Rights pursuant to this Agreement. As between the Parties, each Party shall be solely responsible to prepare, file, prosecute, maintain, and enforce its Intellectual Property Rights in its discretion and at its own cost. Except as expressly set forth in this Agreement, there shall be no right, license, authority, covenant not to sue, immunity from suit, or other defense, whether by implication, by reason of exhaustion, estoppel, or otherwise pursuant to or as a result of this Agreement or the activities of the Parties under this Agreement.

## 5. CONFIDENTIALITY

- 5.1 **"Confidential Information**" of a Party shall mean Technology of such Party that is (a) identified as Confidential Information in an applicable Project Agreement, and (b) marked as "proprietary" or "confidential" at the time of disclosure, or, if disclosed in a form not susceptible to marking, described and designated as "proprietary" or "confidential" in a writing provided to the Recipient within thirty (30) days of such disclosure.
- 5.2 **Non-Disclosure**. For the term of each Project Agreement, and for a period of five (5) years from the end of such Project Agreement, the receiving Party agrees to limit disclosure of the disclosing Party's Confidential Information disclosed in connection with such Project Agreement to those of the receiving Party's employees who have a need to know it, and the receiving Party agrees to use the same care and discretion to avoid disclosure, publication or dissemination outside of those employees as the receiving Party does with similar information of its own which it does not desire to publish, disclose or disseminate. In addition, the receiving Party shall not use, and shall not permit its employees to use, any Confidential Information of the disclosing Party except in connection with the performance of its obligations pursuant to this Agreement and the applicable Project Agreement, except as otherwise permitted hereunder (including pursuant to any license grant set forth herein) or in such Project Agreement.

Notwithstanding the foregoing, the receiving Party's use of Residuals of the disclosing Party's confidential information for any purpose shall not constitute a breach of this Section 5.2. As used herein, "**Residuals**" shall mean that portion of any trade secret or other Technology subject to any obligation of confidentiality between the Parties, which is in intangible form and which is retained in the unaided memory of any of a Party's employees who have had authorized access to such trade secrets or other Technology pursuant to this Agreement.

5.3 **Exceptions**. The receiving Party may disclose Confidential Information if the disclosure is required by law, but the receiving Party must give the disclosing Party reasonable prior notice to allow the disclosing Party an opportunity to obtain a protective order. The obligations of Section 5.2 above will not apply to information to the extent it is: (a) already rightfully in the possession of the receiving Party or its Subsidiaries without an obligation of confidence; (b) independently developed by the receiving Party or its Subsidiaries; (c) publicly available when received by the receiving Party, or becomes publicly available through no fault of the receiving Party or its Subsidiaries; or (d) disclosed by the disclosing Party without obligation of confidence.

## TERM AND TERMINATION

6.

6.1 **Term**. This Agreement, including any Project Agreement, shall be effective during the Term, unless terminated earlier in accordance with the provisions hereof and except to the extent any Project Agreement expressly sets forth a different term.

## 6.2 **Termination for Breach**.

- (a) If any material breach of this Agreement occurs, and such breach is not cured within thirty (30) days after written notice from the non-breaching Party, the non-breaching Party shall have the right to terminate this Agreement by giving the breaching Party five (5) days' written notice.
- (b) If any material breach of any Project Agreement occurs, and such breach is not cured within thirty (30) days after written notice from the non-breaching Party, the non-breaching Party shall have the right to terminate that Project Agreement by giving five (5) days' written notice.
- 6.3 **Termination for Bankruptcy**. A Party may terminate this Agreement by giving written notice of termination to the other Party at any time upon or after:
  - (a) the filing by the other Party of an petition in bankruptcy or insolvency which petition or proceeding is not dismissed within sixty (60) days;
  - (b) any adjudication that the other Party is bankrupt or insolvent;
  - (c) the filing by the other Party of any petition or answer seeking reorganization, readjustment or arrangement of its business under any law

relating to bankruptcy or insolvency which petition or proceeding is not dismissed within sixty (60) days;

- (d) the appointment of a receiver for all or substantially all of the property of the other Party;
- (e) the institution of any proceeding for the liquidation winding up of the other Party's business which petition or proceeding is not dismissed within sixty (60) days.
- 6.4 **Effect of Termination of Agreement**. Upon any termination of this Agreement all Project Agreements shall simultaneously terminate. No termination in this Agreement shall relieve either Party of any obligation or liability accrued hereunder prior to such termination. Upon the expiration or earlier termination of this Agreement, the Parties will immediately cease performing work under all Project Agreements. Upon the expiration or earlier termination of any Project Agreement, the Parties will immediately cease performing work under the expired or terminated Project Agreement.
- 6.5 **Survival**. Notwithstanding anything to the contrary in this Agreement, Sections 1, 3, 4, 5 (as provided therein), 6.4, 6.5, 7.2, 7.3, and 10 shall, to the extent applicable, survive any expiration or termination of this Agreement.

## 7. REPRESENTATIONS AND WARRANTIES; LIABILITY LIMITATION

- 7.1 **Authority**. Each Party represents and warrants that it has the corporate power and authority to enter this Agreement, and to carry out the terms and obligations set forth in this Agreement, and that the persons executing this Agreement on its behalf have the authority to act for and bind the Party.
- 7.2 **No Other Warranty.** EXCEPT AS SPECIFICALLY PROVIDED IN THIS SECTION 7, NEITHER PARTY MAKES ANY OTHER WARRANTIES, EITHER EXPRESS OR IMPLIED, AS TO ANY OTHER MATTER WHATSOEVER, AND EACH PARTY HEREBY EXPRESSLY DISCLAIMS ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR ANY PARTICULAR PURPOSE, ACCURACY, NON-INFRINGEMENT, AND TITLE, AND ANY WARRANTIES THAT MAY ARISE FROM COURSE OF DEALING, COURSE OF PERFORMANCE OR USAGE OF TRADE.
- 7.3 **Limitation of Liability**. Except as expressly set forth herein, neither Party shall be liable for any indirect, incidental, consequential, special or punitive damages arising out of or relating to this Agreement, irrespective of whether such Party has been advised of the possibility of any such damages. The foregoing limitation of liability shall not apply with respect to any liability arising out of or relating to any breach of Section 5 or any infringement, misappropriation or other violation of any Intellectual Property Rights of a Party.

#### 8. NOTICES

All notices and other communications required or permitted hereunder shall be in writing and shall be mailed by Federal Express or other internationally recognized express carrier, registered or certified mail, postage prepaid, or otherwise delivered by hand or by messenger, addressed to a Party at the addresses set forth below, or at such other address as a Party may substitute by written notice provided to the other Party in such manner. Such notices shall be deemed to have been served when delivered.

## If to Hitachi:

Hitachi, Ltd. Business Development Office 6-6 Marunouchi 1-chome Chiyoda-ku, Tokyo 100-8280, Japan Attention: General Manager Facsimile: 81-3-4564-6260

#### If to Company:

Western Digital Corporation 3355 Michelson Drive, Suite 100 Irvine, California 92612 Attention: General Counsel Facsimile: (949)672-9612 E-mail: <u>Michael.Ray@wdc.com</u>

#### 9. ASSIGNMENT

This Agreement shall be binding upon and inure to the benefit of and be enforceable by the respective successors and permitted assigns of the Parties. Nothing in this Agreement shall confer any rights upon any person other than the Parties and their respective successors and permitted assigns. Neither Party may assign this Agreement or its rights hereunder to any third party without the written consent of the other Party; provided, that each Party may assign this Agreement and its rights hereunder to one or more of its direct or indirect Subsidiaries without the consent of the other Party so long as the assignee assumes the obligations hereunder of the assigning party in a writing reasonably acceptable to the other Party. Any attempted assignment of this Agreement in violation of this Section 9 shall be void and of no effect.

## 10. MISCELLANEOUS

10.1 **Entire Agreement**. Except for the Purchase Agreement, the License Agreement, other agreements entered into pursuant to the Purchase Agreement, and, where applicable, the Inventor Award Integration Agreement dated December 10, 2010

by and between Hitachi and Hitachi Global Storage Technologies Netherlands B.V. (the "IAIA") (a) this Agreement, together with any Project Agreements, contains the complete agreement between, and contains all of the promises and undertakings made to each other by, the Parties regarding the subject matter of this Agreement, and (b) except as set forth in the Transition Services Agreement (as defined in the Purchase Agreement), any and all prior agreements, representations, negotiations, and undertakings between the Parties, oral or written, express or implied, with respect to the subject matter hereof, are hereby superseded and merged by and into this Agreement, effective as of the Closing Date. This Agreement may not be revised or modified without the mutual written consent of the Parties.

- 10.2 **Application of IAIA**. Notwithstanding anything to the contrary set forth in the IAIA:
  - (a) all Patents claiming inventions that constitute Developed IP will be deemed "New ERA Patents" for purposes of the IAIA (despite the fact that such Patents do not arise out of research pursuant to the ERA);
  - (b) any employee or agent of Hitachi who is named as an inventor in any such Patent shall be deemed a "Hitachi Inventor" with respect to such Patent for purposes of the IAIA;
  - (c) Section 4 of the IAIA shall apply to (i) except as set forth in Section 10.2(d), the incentive payments made by Hitachi to any such employee or agent of Hitachi with respect to such Patents (with such payments deemed "HGST Base Awards" (as defined in the IAIA), despite the fact that such payments are not being made to an employee of HGST), and (ii) Company's reimbursement of Hitachi for all such payments shall be made as if all applicable references in the IAIA to "HGST" were references to Company; *provided, however*, that the Parties agree that, in consideration of the larger size of Company's business after the Closing Date, the incentive payments made in each case to any such employee and agent of Hitachi pursuant to this Agreement and in accordance with the IAIA during the Term shall be double (i.e., two-hundred percent (200%) of) the relevant HGST Base Awards set forth in Section 4 of the IAIA and Appendix C thereof. The Parties acknowledge and agree that the foregoing obligations to make incentive payments to employees and agents of Hitachi and to be reimbursed by Company shall survive any expiration, termination or renegotiation of the IAIA;
  - (d) Without limitation to the incentive payments reimbursable by Company pursuant to Section 10.2(c), the Parties acknowledge and agree that, for all Patents claiming inventions that constitute Developed IP during the Term of this Agreement, (i) such Patents shall be treated as "New ERA Patents" (as defined in the IAIA) (despite the fact that such Patents do not arise out of research pursuant to the Entrusted Research Agreement dated

September 25, 2003 (the "<u>ERA</u>")) and such inventions shall be treated as if they resulted from entrusted R&D activities based on the ERA for which licensing/assignment awards are payable by Hitachi or Hitachi Global Storage Technologies, Inc. ("<u>HGST</u>"), as applicable, to Hitachi's inventors pursuant to (x) the Memorandum, dated as of May 24, 2005, between Seller and HGST and (y) the Memorandum, dated as of March 16, 2005, between Seller and HGST ((x) and (y), collectively, the "<u>Memoranda</u>"), including all appendices thereto and related materials, and (ii) Company shall reimburse Hitachi for any such licensing/assignment awards paid by Hitachi to any employee or agent of Hitachi who is named as an inventor in any such Patent pursuant to the Memoranda (including all appendices thereto and related materials). The Parties acknowledge and agree that the foregoing obligations to make incentive payments to employees and agents of Hitachi and reimbursed by Company shall survive any expiration, termination or renegotiation of the Memoranda and/or IAIA unless otherwise agreed to by the Parties pursuant to Section 10.2(e);

(e) At any time prior to March 31, 2013, upon the request of either Party, Hitachi and Company shall discuss in good faith the treatment of inventor awards, licensing/assignment awards and other compensation payable to any employee or agent of Hitachi who is named as an inventor in any Patent that constitutes Developed IP under this Agreement, including, if proposed by either Party, the payment by the Company of a one-time lump-sum payment to Hitachi in consideration for the termination of Company's further obligation to Hitachi to pay (i) inventor awards or other compensation to inventors employed by Hitachi (including patent filings, patent issues, internal uses of patents and patent licensing and assignment) under Section 10.2(c), and (ii) licensing/assignment awards to inventors employed by Hitachi under Section 10.2(d), in each case for Patents with an effective filing date prior to March 31, 2013; provided that (x) Company shall continue to make the escalated incentive payments set forth in Section 10.2(c) to applicable employees and agents of Hitachi who are named as an inventor in any Patents with an effective filing date on or after April 1, 2013 unless otherwise expressly agreed to by the Parties, and (y) Company shall continue to make the licensing/assignment awards to inventors employed by Hitachi under Section 10.2(d) for Patents with an effective filing date on or after April 1, 2013 unless, on a case-by-case basis with respect to each such Patent, the Parties agree to a lump-sum payment in consideration for the termination of Company's obligation to reimburse Hitachi for such licensing/assignment awards under Section 10.2(d) for Patents with an effective filing date on or after April 1, 2013, in which case such lump-sum payment shall be paid by Hitachi and reimbursed by Company to the applicable employee or agent of Hitachi at the time the Patent File Award pursuant to the IAIA and this Agreement is made for such Patent.

- 10.3 **Severability**. If any paragraph, provision or clause of this Agreement shall be found or held to be invalid or unenforceable by a court or other decision-making body of competent jurisdiction, in a judgment from which no further appeal can be taken, the remainder of this Agreement shall remain valid and enforceable, and to the extent required in the pursuit of this Agreement, the Parties shall negotiate in good faith a substitute, valid and enforceable provision that reflects the Parties' intent in entering this Agreement.
- 10.4 **ADR Provision**. The Parties agree that, in the event of a dispute arising under this Agreement, the Parties shall agree to attend a nonbinding mediation prior to bringing any action in a court of law. The mediation shall be facilitated by a neutral third-party, selected by the Parties. A representative from each Party with authority to settle on behalf of that party shall attend the mediation in person. The purpose of the mediation is to resolve the disputed issue(s) without the need for formal legal proceedings.
- 10.5 **Governing Law**. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware as applied to contracts entered into and fully performed in such state.
- 10.6 **Jurisdiction**. The Parties agree that the Delaware Court of Chancery (and if jurisdiction in the Delaware Court of Chancery shall be unavailable, any Delaware State court and the Federal court of the United States of America sitting in the State of Delaware) shall have exclusive jurisdiction of interpreting or enforcing this Agreement, provided, however, that, notwithstanding the foregoing, either Party may (i) seek injunctive or other preliminary relief, (ii) enforce its Intellectual Property Rights, and (iii) seek recognition and enforcement of any order or judgment, in any court of competent jurisdiction. The parties expressly waive their right to a trial by jury.
- 10.7 **Counterparts.** This Agreement may be executed in identical counterparts, each of which shall constitute an original and all of which shall constitute one and the same agreement.
- 10.8 **Further Assurances**. Each Party hereto shall execute and cause to be delivered to each other Party hereto such instruments and other documents, and shall take such other actions, as such other Party may reasonably request, to effect, evidence, record or perfect any of the licenses and other rights set forth in this Agreement. The requesting Party shall reimburse the other Party for its reasonable out-of-pocket costs in responding to a request under this Section.
- 10.9 **Export Compliance**. Company understands and acknowledges that Hitachi is subject to regulation by agencies of the U.S. government, including the U.S. Department of Commerce, and similar regulations in other jurisdictions, which prohibit export or diversion of certain products and technology to certain countries. Company understands and acknowledges that the Deliverables and other Developed IP are subject in all respects to such United States and other

jurisdictions' laws and regulations as shall from time to time govern the license and delivery of technology and products abroad by persons subject to the jurisdiction of the United States, including the Export Administration Act of 1979, as amended, any successor legislation, and the Export Administration Regulations ("**EAR**") issued by the Department of Commerce, International Trade Administration, Bureau of Industry and Security ("**BIS**"). Company warrants that it will comply in all respects with the export and re-export restrictions applicable to the Deliverables and other Developed IP and further agrees that it will not export, re-export or transship, directly or indirectly, any Deliverables and other Developed IP without the proper authorization from BIS under the EAR, or proper authorization from competent authorities in other jurisdictions, explicitly permitting the export, re-export or transshipment.

[REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

## IN WITNESS WHEREOF, the Parties have caused their duly authorized officers to execute this Agreement as of the Closing Date.

For and on behalf of Hitachi, Ltd.

By:

Title:

For and on behalf of Western Digital Corporation

By:

Name:

Name: Michael C. Ray

Title: Senior Vice President, General Counsel and Secretary

S-1

**R&D** Services Agreement

# EXHIBIT G

# EXHIBIT F TO THE STOCK PURCHASE AGREEMENT Form of Branding Agreement

[see attached]

Form of Branding Agreement

#### Form of

## **BRANDING AGREEMENT**

This Branding Agreement ("**Branding Agreement**") effective as of the Closing Date (as defined below), is entered into by and between Viviti Technologies Ltd., a company incorporated under the laws of the Republic of Singapore with its principal place of business at 4 Kaki Bukit Avenue 1, #03-08, Singapore 417939 ("**Company**") and Hitachi, Ltd., a Japanese company with its principal place of business at 6-6, Marunouchi 1-chome, Chiyoda-ku, Tokyo 100-8280, Japan ("**Hitachi**"). Each of the signatories to this Branding Agreement is referred to as a "**Party**," and jointly as the "**Parties**."

#### RECITALS

WHEREAS, Western Digital Corporation, a Delaware company with its principal place of business at 3355 Michelson Drive, Suite 100, Irvine, California 92612, USA ("**Buyer Parent**"), Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect wholly-owned subsidiary of the Buyer Parent ("**Buyer**"), Hitachi, and Company have entered into that certain Stock Purchase Agreement dated as of March 7, 2011 (the "**Purchase Agreement**");

WHEREAS, Company and Hitachi desire that Company may, on a transitional basis and under Hitachi's authorization, continue to use certain branding of Hitachi with respect to Company's products in accordance with the terms and conditions set forth below in order to provide for an orderly transition to new branding;

NOW, THEREFORE, in consideration of the mutual promises and covenants set out herein, the Parties agree as follows:

#### **TERMS OF AGREEMENT**

#### 1. <u>DEFINITIONS</u>

1.1 Any capitalized term used but not defined herein, including without limitation the following terms, has the meaning set forth in the Transaction Documents:

- (a) Closing Date
- (b) External Drive
- (c) HDD
- (d) HDD Components
- (e) HDD Products
- (f) Losses

- (g) RAID
- (h) SSD
- (i) Storage Software Products.

1.2 "Company Current Products" means those Products of Company or its Subsidiaries in active production as of the Closing Date.

1.3 **"Company Inventory Units**" means any units of Products produced, packaged, or in inventory as of the Closing Date (regardless of whether such Products may be Company Current Products as of the Closing Date) provided that any such units of Products are branded with the Licensed Hitachi Marks as of the Closing Date.

1.4 **"Company Non-Current Products**" means Products previously manufactured and sold by Company or its Subsidiaries under the Licensed Hitachi Marks and that are not in active production by Company or its Subsidiaries at the Closing Date.

1.5 **"Company Product(s)**" means Company Current Products, Company Non-Current Products, and Remanufactured Products and Repair Parts.

1.6 **"Company Reseller**" means an organization or person that is authorized by Company directly, or indirectly through another Company Reseller, to sell or lease Company Products.

1.7 **"Electronic Label**" means any label, sticker, decal or the like affixed to an Internal Product bearing a machine-readable code which identifies with specificity an Internal Product, and which may include other information regarding the Internal Product.

1.8 **"EOL**" or "**End-of-Life**" means, with respect to a particular Product, the date on which Company, in its sole discretion, chooses to no longer offer such Product for sale on a general commercial basis.

1.9 "**Hitachi Marks**" has the meaning set forth in Section 5.1.

1.10 "**Internal Product**" means an HDD Product sold for use in an HDD Included Products or SSD product sold for use in an SSD Included Product.

1.11 **"In-Warranty Products**" has the meaning set forth in Section 2.1(b).

1.12 "Licensed Articles" has the meaning set forth in Section 2.1(a).

1.13 "Licensed Hitachi Marks" shall mean the Hitachi trademarks and/or service marks set forth in Schedule 2.1

1.14 "Licensed Hitachi Trade Dress" has the meaning set forth in Section 2.2.

1.15 "Licensed Marks" means the Licensed Hitachi Marks and Licensed Hitachi Trade Dress.

1.16 **"Licensed Materials**" means packaging; labels; sales, instruction and service materials; marketing materials; product documentation regardless of form or media in which presented, and installation software.

1.17 **"Licensed Products**" means the Licensed Current Products, Warranty Licensed Products, Remanufactured Products, and internal components as more fully described in Sections 2.1(a) through (e), and the Licensed Trade Dress Products.

1.18 "Licensed Trade Dress Products" has the meaning set forth in Section 2.2.

1.19 **"Material Quality Breach**" has the meaning set forth in Section 4.4.

1.20 **"New Trademark**" means any mark developed by Company for any product developed, manufactured, or produced by Company subsequent to the Closing Date that is not part of or encompassed by Company Products.

1.21 "Products" means HDD Products, SSD products, RAID products, External Drive products, and Storage Software Products.

1.22 **"Qualification**" means technical evaluation or testing by an OEM customer which must be passed by a supplier with respect to a particular product in order to qualify the product for purchasing by such customer for use in an HDD Included Product or SSD Included Product. As used in this Branding Agreement, the term "Qualification" includes evaluation and testing with respect to technical and engineering requirements and excludes any marketing, promotion, advertising or retail display requirements.

1.23 **"Remanufactured Products**" means Company Non-Current Products and Company Current Products sold under or that bear the Licensed Hitachi Marks, whether produced or sold before or after Closing, that are remanufactured or repaired by Company after the Closing Date.

1.24 **"Repair Parts**" means (a) replacement products for (i) Company Non-Current Products or (ii) Company Current Products; and (b) replacement parts which, to the extent available prior to the Closing Date, bore the Licensed Hitachi Marks prior to the Closing Date and which are used for the repair or remanufacture of Company Non-Current Products or Company Current Products.

1.25 **"SSD Included Product**" means products that include an SSD among their components but are a combination of such SSD with at least one other device distinct from such SSD and provide a material function that is not provided by a single SSD. Examples of SSD Included Products are storage systems (including RAID), External Drives, JBODs, enterprise systems, servers, server blades, desktop computers, portable computers (including notebook and handheld computers), personal digital assistants, digital video recorders, digital cameras, game consoles, mobile phones and global positioning systems that include an SSD among their components.

1.26 **"Subsidiary**" of either Party means a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities

or other interests having ordinary voting power for the election of directors or other governing body are at the time beneficially owned, directly, or indirectly, by such party. For the purposes of this Agreement, Company and its Subsidiaries are not Subsidiaries of Hitachi.

#### 1.27 **"Terminated Products**" has the meaning set forth in Section 8.2(b).

## 2. GRANT OF LICENSES AND CONSENTS

2.1 Hitachi hereby grants to Company and Company hereby accepts non-exclusive, royalty free licenses to use the Licensed Hitachi Marks under the following terms and conditions:

(a) on or in connection with Company Current Products, Company Inventory Units, and Licensed Materials relating thereto ("**Licensed Articles**"), Company may use the Licensed Hitachi Marks for a period of one hundred eighty (180) days after the Closing Date ("**Cut-Over Date**") (subject to Section 6.2), provided the Licensed Articles, subsequent to the Closing Date, are manufactured by or for Company pursuant to the quality control requirements set forth in Section 4 and are marketed by Company itself or through Company Resellers;

(b) on or in connection with Repair Parts for Company Current Products or Company Non-Current Products under warranty after the Closing Date ("**In-Warranty Products**") and Licensed Materials relating thereto (collectively, "**Warranty Licensed Products**"), Company may use the Licensed Hitachi Marks for a period of time equal to and coterminous with the warranty period for such In-Warranty Products provided that the Licensed Hitachi Marks were used in connection with such Repair Parts prior to the Closing Date (to the extent such Repair Parts were available before the Closing Date);

(c) on or in connection with Remanufactured Products used to replace particular Products and Licensed Materials relating thereto, Company may use the Licensed Hitachi Marks until the seven (7) year anniversary of the EOL of the particular Product or until such Product is no longer supported by Company or until the seven (7) year anniversary of the Cut-Over Date, whichever occurs first, provided such products are manufactured by or for Company pursuant to the quality control requirements set forth in Article 4 and are marketed by Company itself or through Company Resellers;

(d) notwithstanding Section 2.1(a) above, in any Electronic Label used to identify any Company Current Product that is an Internal Product, Remanufactured Product that is an Internal Product or Repair Part for an Internal Product for so long as continued use of the applicable Licensed Hitachi Mark(s) in such Electronic Label is necessary in order to avoid triggering a Qualification requirement (as dictated by the applicable agreement with the customer of such Company Product) that would not arise if the applicable Hitachi Mark(s) were retained in the Electronic Label, but only until the Company alters such Company Current Product, Remanufactured Product or Repair Part identified by the Label in a manner that triggers a Qualification requirement, provided that the Licensed Hitachi Marks are not more visible to individuals than are the Labels used in units of Company Current Products manufactured prior to the Closing Date;

(e) on or in connection with internal components of any Company Current Product that is an Internal Product, including but not limited to HDD heads and mask works used

in the making of HDD Components, for so long as continued use of the applicable Licensed Hitachi Mark(s) on or in connection with such internal components is necessary in order to avoid triggering a Qualification requirement (as dictated by the applicable agreement with the customer of such Company Current Product) that would not arise if use of the applicable Hitachi Mark(s) were retained on or in connection with such internal components, but only until the Company alters the internal components, or the Company Current Product in which such internal components are used, in a manner that triggers a Qualification requirement, provided that the Licensed Hitachi Marks are not generally visible other than to those individuals installing, administering, or maintaining the any Company Current Product.

2.2 Hitachi hereby grants to Company and Company hereby accepts a non-exclusive, royalty free license to use the Hitachi trade dress as depicted in <u>Schedule 2.2</u> (the "Licensed Hitachi Trade Dress") on or in connection with consumer boxes, commercial packaging, labeling, product information provided on diskettes and/or CDs, and installation software (collectively "Packaging") for any Company Products sold with Packaging bearing the Licensed Hitachi Trade Dress as of the Closing Date (collectively, the "Licensed Trade Dress Products") for one hundred eighty (180) days after the Closing Date, provided that the Licensed Trade Dress Products are manufactured by or for Company, or, with respect to such software, Company has a right to distribute such software, and are marketed by Company itself or through Company Resellers.

2.3 The licenses granted in this Article 2 shall not apply to any services (including but not limited to repair or maintenance services) offered by or under authorization of Company.

## 3. COMPANY'S USE OF THE LICENSED MARKS

3.1 The grant of licenses set forth in Article 2 hereof shall include and incorporate as if fully set forth therein the following terms and conditions which shall control and define the use of the Licensed Marks on or in connection with the Licensed Products:

(a) All use of the Licensed Marks by Company shall be limited to the specific Licensed Products set forth in the license grant and shall meet all quality control requirements of Hitachi as specified in Article 4.

(b) Company shall use the Licensed Marks only in the form and manner and with such proprietary legends and notices as used by Company or its Subsidiaries prior to the Closing Date for the Licensed Products.

(c) Subject to Section 2.1(d) and (e), Company shall place the symbols "™", "tm", or ®, as is appropriate according to the laws in each applicable country or territory next to the Licensed Hitachi Marks. In the case of Licensed Materials, such symbols must appear in connection with the Licensed Hitachi Marks at least once on or in each separate piece of Licensed Material.

(d) Subject to Section 2.1(d) and (e), Company shall use the following legend at least once in each separate piece of Licensed Material on which any of the Licensed Hitachi

Marks appear: "[Licensed Hitachi Mark] is a [registered] trademark of Hitachi, Ltd. and is used under license."

(e) Subject to Section 2.1(d) and (e), on all Licensed Trade Dress Products bearing the Licensed Hitachi Trade Dress, Company shall include the © symbol and the legend "© Hitachi, Ltd."

(f) Except as authorized pursuant to this Branding Agreement, Company shall not at any time use, or authorize the use of, any Hitachi Mark, or any trademark, trade name, service mark, domain name or corporate name consisting of or incorporating any Hitachi Mark, or any mark phonetically equivalent or confusingly similar to any Hitachi Mark.

(g) Company shall not adopt or use on or in connection with any product or service it produces, manufactures, advertises, distributes, sells, or offers for sale, a New Trademark, or any service mark, trade name, domain name or other designation, that is a phonetic equivalent or a colorable imitation of any of the Licensed Hitachi Marks or that is otherwise confusingly similar to any of the Licensed Hitachi Marks, nor shall Company adopt or use any trade dress that copies or is a colorable imitation of the Licensed Hitachi Trade Dress or that is otherwise confusingly similar to the Licensed Hitachi Trade Dress. If any application for registration of a mark or other designation is filed by or on behalf of Company after the Closing Date and such mark is confusingly similar or disparaging to or dilutes any of the Licensed Hitachi Marks anywhere in the world, Company shall abandon all use of such mark and any application to register or registration thereof.

(h) Without limitation to Section 3.1(g), Company shall not at any time use or display (including, without limitation, on or in connection with any product or service it produces, manufactures, advertises, distributes, sells, or offers for sale, or as a corporate name) any Licensed Hitachi Mark contiguous to, connected to, or otherwise arranged in combination with any HGST Mark (or any mark confusing similar to or including any HGST Mark), such that the Licensed Hitachi Mark is not visually distinct from, or is grouped with or part of a composite mark including, any HGST Mark (or any mark confusing similar to or including any HGST Mark), or in any other manner that suggests or implies an association between Hitachi, any of Hitachi's Subsidiaries and/or any Licensed Hitachi Mark, on the one hand, and any HGST Mark (or any mark confusing similar to or including any HGST Mark), on the other hand.

(i) With respect to its use of the Licensed Marks, Company shall comply with all applicable laws and regulations pertaining to the use and designation of trademarks and trade dress in the each country or territory in which the Licensed Marks may be used by Company.

3.2 Company shall not pledge or otherwise encumber any of the Licensed Marks and shall not use any of the Licensed Marks as security or collateral for any loans or indebtedness.

#### 4. <u>QUALITY CONTROL</u>

4.1 Company agrees that the nature and quality of the Licensed Products when such products are sold or distributed in connection with the Licensed Marks shall be commensurate with standards previously achieved and maintained by Company for the identical goods or for comparable products manufactured or distributed by Company during calendar year 2010.

Company shall provide to Hitachi monthly consolidated quality reports with respect to each Licensed Product marked with the Licensed Marks and for which shipments are being made during the term of the applicable license granted pursuant to Article 2.

4.2 Company agrees that all uses of the Licensed Hitachi Trade Dress as permitted by this Branding Agreement shall conform to and are subject to the guidelines established by Hitachi for such trade dress, which guidelines are attached as <u>Schedule 4.2</u>.

4.3 Except for units of Products produced, packaged, or in inventory as of the Closing Date, and Licensed Materials developed by Company or its Subsidiaries prior to Closing or varying in no significant respect from Licensed Materials developed by Company or its Subsidiaries prior to Closing, Company shall, upon Hitachi's request, (a) provide to Hitachi or, at Hitachi's request, demonstrate to Hitachi, samples of Licensed Products that bear, or are marketed or sold using any of the Licensed Marks, and (b) provide to Hitachi samples of Licensed Materials that bear any of the Licensed Marks. Any such requests shall be with reasonable notice and at reasonable intervals.

4.4 Any modifications by Company after the Closing to a Licensed Product marked with any Licensed Mark, including, but not limited to changes in manufacturing processes thereof or parts vendor selection by Company, that result in adverse changes of the annual quality return rate for such product in each of two (2) consecutive thirty (30) day periods greater than three percent (3%) shall be noticed to Hitachi within five (5) Business Days. Any such modifications shall, upon resulting in such changes, be deemed "**Material Quality Breaches**." Company shall provide to Hitachi samples of such Licensed Products.

4.5 Except for Licensed Materials developed by Hitachi or its Subsidiaries prior to Closing or varying in no significant respect from Licensed Materials developed by Hitachi or its Subsidiaries prior to Closing, Company shall submit to Hitachi all new Licensed Materials developed by Company after Closing on and in which the Licensed Marks appear prior to their commercial use. Within fifteen (15) Business Days of receipt of any submission by Company, Hitachi shall inform Company in writing whether the submission meets the applicable standards for quality or usage. If Hitachi does not respond to Company within fifteen (15) Business Days, Company may assume without penalty or prejudice that its submission has been approved, provided that Company produces a written receipt evidencing that the submission was sent to and received by Hitachi.

4.6 If Hitachi determines that any Licensed Products or Licensed Materials using the Licensed Marks materially do not meet the requirements set forth in this Branding Agreement, Hitachi may notify Company, providing Company with a description of the deficiencies. Company shall submit to Hitachi a report describing Company's plan for investigation of the causes of and for remediation of such deficiencies and the time frame for such actions within thirty (30) Business Days after receipt of the notice. If Hitachi considers the actions or time frame described in such report to be inconsistent with applicable industry standards, the Parties shall promptly meet and confer with respect to the report. If the Parties are unable through such meeting to mutually agree on appropriate actions and time frames, Hitachi shall have the right, effective on notice to Company, to suspend use of the Licensed Marks on or in connection with the materially deficient Licensed Products or Licensed Materials until such deficiencies are

cured. To the extent Company has in inventory during a proper suspension of use of the Licensed Marks pursuant to this Section 4.6 any Licensed Products or Licensed Materials that: (i) were produced, manufactured, or remanufactured after the Closing Date, (ii) materially do not meet the requirements set forth in this Branding Agreement, and (iii) are subject to such proper suspension ("**Suspended Products**"); then Company shall not be authorized under the suspended license or consent to ship any such Suspended Products until the Licensed Marks have been removed or obliterated (for example, by covering such Licensed Marks with a sticker) or such material deficiencies have been cured.

4.7 For as long as Company is licensed to use the Licensed Marks for any of the Licensed Products, Company shall provide to Hitachi on an as-needed basis but no less than once during the term of the license grants set forth in Sections 2.1(a) and 2.2 and once per year during the term of the license grants set forth in Sections 2.1(b), 2.1(c), 2.1(d) and 2.1(e) the following information: (a) a list of all recalls of any Licensed Products on which the Licensed Marks appear, the reason for each recall and the action taken to remedy the problem; and (b) the number of warranty claims submitted for each Licensed Product on which the Licensed Marks appear, broken down for each product. Hitachi shall have the right to require that Company produce to Hitachi any written complaints or actual documents from which the information provided by Company under the terms of this provision were created or obtained.

## 5. <u>DOMAIN NAME TRANSITION</u>

5.1 Within ninety (90) days after the Closing Date, Company shall assign to Hitachi all registrations anywhere in the world of any domain names held by Company that include any trademark or service mark owned by Hitachi or any of its Subsidiaries (such trademarks and service marks, the "**Hitachi Marks**" and such domain names, the "**Transferred Domain Names**"). Company and Hitachi hereby agree that, without limitation to the generality of the foregoing, (a) the Hitachi Marks include, but are not limited to, the terms "HGST" and "Hitachi GST" (in any combination of upper and/or lower case letters), any derivations of the foregoing terms, and any logos based on or containing any of the foregoing (including, but not limited to, stylized or graphical versions or renditions thereof) and (b) the Transferred Domain Names include, but are not limited to, any domain names containing the term "hgst" or "hitachigst" (in any combination of upper and/or lower case letters) or any derivation of the foregoing terms.

5.2 During the twelve (12)-month period following the Closing Date, Hitachi shall maintain at the landing page for each Transferred Domain Name a hyperlink that connects the user to a designated Company URL, as notified by Company to Hitachi.

## 6. <u>CESSATION OF USE OF SELLER MARKS</u>

6.1 To the extent that as of the Closing Date Company or any of its Subsidiaries own any registered trademarks or service marks anywhere in the world that contain any Hitachi Mark, Company shall, and shall cause its Subsidiaries to: (a) not renew or maintain (including not filing any affidavits of use or paying any registration fees) such registrations, in order to allow such registrations to expire upon the expiration of their then-current terms and (b) use such trademarks or service marks only to the extent expressly permitted by this Branding Agreement.

6.2 Company agrees that, notwithstanding anything to the contrary set forth herein, but subject to Section 6.3, it shall, and shall cause each of its Subsidiaries to:

(a) as soon as practicable after the Closing Date and in any event within thirty (30) days following the Closing Date, change its and their corporate names to names that do not include any Hitachi Mark (or any word confusingly similar thereto) and make any necessary legal filings with the appropriate Governmental Entities to effect such change;

(b) as soon as practicable after the Closing Date and in any event within forty-five (45) days following the Closing Date, cease use of all stationary, business forms and business cards bearing any Hitachi Marks;

(c) as soon as practicable after the Closing Date and in any event within sixty (60) days following the Closing Date (or such longer period as may be required to obtain any needed approvals of any Governmental Entity), remove, strike over or otherwise obliterate all Hitachi Marks from all facilities, vehicles, physical signage, badges and uniforms of the Company and each of its Subsidiaries; and

(d) as soon as practicable after the Closing Date and in any event within

ninety (90) days following the Closing Date, cease (i) reproducing any product manuals, collateral materials, promotional materials and packaging materials bearing any of the Hitachi Marks, except for those that are Licensed Articles (which shall be subject to Section 2.1(a)), (ii) replicating any software displaying any of the Hitachi Marks, except for software which is a Licensed Article (which shall be subject to Section 2.1(a)), and (iii) publishing any websites bearing any of the Hitachi Marks.

6.3 Notwithstanding any provision of this Agreement, this Agreement shall not prohibit, and Hitachi shall take no action to prevent, any of Company or any Subsidiary of Company during the hold separate period required by the Ministry of Commerce of the People's Republic of China ("**MOFCOM**") from using: (a) the name "HGST" (the four-letter name) and (b) the domain name "hgst.com" in connection with operating the hold separate business during the hold separate period, in each case to the extent necessary, and only for as long as necessary (and in any event for no longer than such hold separate period), to comply with the MOFCOM order dated March 2, 2012 China Time (the "**MOFCOM Order**"). For clarity, the foregoing shall not affect Company's obligations pursuant to Section 5.1 and Section 6.1 except with respect to such Subsidiary's use of the "HGST" name and the "hgst.com" domain name (collectively, such name and domain name constituting the "**HGST Marks**") as set forth (and during the time period set forth) above. Such Subsidiary shall treat the HGST Marks as if they were Licensed Hitachi Marks during the time period set forth above, and all uses by such Subsidiary of the HGST Marks shall be subject to, and such Subsidiary shall, in its use of the HGST Marks, comply with, all terms and conditions of this Branding Agreement applicable to the Licensed Hitachi Marks, the protection thereof, and/or the use thereof by Company, including without limitation the terms and conditions set forth in Sections 3 and 4 (but excluding Sections 3.1(b) and 3.1(d)), in each case solely to the extent such terms and conditions are consistent with the use of the HGST Marks required by the MOFCOM Order. Notwithstanding the foregoing, upon termination of the hold separate period, Company will, and shall be permitted to, promptly transition away from use of the HGST Marks in the same manner as is

provided for the Transferred Domain Names (in the case of the "hgst.com" domain name) and the Licensed Hitachi Marks (in the case of the "HGST" name) in this Agreement, but with the final date of the hold separate period taking the place of the "Closing Date" for all purposes of such transition away from the use of the HGST Marks.

## 7. RIGHT TO SUBLICENSE AND ASSIGN AGREEMENT

7.1 None of the rights provided under this Branding Agreement relating to use of the Licensed Marks may be assigned, pledged, encumbered, or otherwise transferred by Company and shall not inure to the benefit of any trustee in bankruptcy, receiver or other successor of Company, whether by operation of law or otherwise, without the prior written consent of Hitachi, which shall not be unreasonably withheld, and any attempted or actual assignment or transfer without such consent shall be null and void.

7.2 Other than to its Subsidiaries, Company shall not sublicense this Branding Agreement or any rights of Company hereunder to any third party including to any affiliate of Company or, other than Subsidiaries and as provided in Section 7.4, otherwise authorize any third party, including any affiliate, to use any of the Licensed Marks.

7.3 Company shall not have the right to transfer or assign the rights granted herein to any third party including any party that has any ownership interest in or that exercises any control over Company, even as part of a transfer of all of the business of Company without the prior written consent of Hitachi, which shall not be unreasonably withheld. This Branding Agreement shall be fully and freely assignable by Hitachi.

7.4 Notwithstanding anything in this Branding Agreement to the contrary, Company may transfer its rights under this Branding Agreement between itself and its Subsidiaries and any of its legal successors-in-interest formed pursuant to a reorganization or other corporate restructuring of Company, provided that after such reorganization or corporate restructuring, such successor-in-interest agrees in writing to be bound by the terms of this Branding Agreement and Company and its Subsidiaries have substantially the same assets as Company and its Subsidiaries had immediately prior to such reorganization or corporate restructuring.

## 8. <u>TERMINATION</u>

8.1 Hitachi may terminate this Agreement by giving written notice of termination to Company at any time upon or after:

(a) the filing by Company of a petition in bankruptcy or insolvency which petition or proceeding is not dismissed within sixty (60)

days;

(b) any adjudication that Company is bankrupt or insolvent;

(c) the filing by Company of any petition or answer seeking reorganization, readjustment or arrangement of its business under any law relating to bankruptcy or insolvency which petition or proceeding is not dismissed within sixty (60) days;

(d) the appointment of a receiver for all or substantially all of the property of Company;

(e) the institution of any proceeding for the liquidation winding up of Company's business which petition or proceeding is not dismissed within sixty (60) days.

8.2 The licenses granted in Article 2 may also be terminated in accordance with the following:

(a) Company shall be deemed in material breach of this Branding Agreement if there is a Material Quality Breach as to any of the Licensed Products marked with the Licensed Marks by Company.

(b) If Company materially breaches this Branding Agreement, Hitachi shall provide written notice to Company, advising that Company has committed a material breach, describing the material breach, the specific license grant(s) in Article 2 to which the material breach relates, and, in the case of a Material Quality Breach, the specific Licensed Products giving rise to the material breach, and providing Company with thirty (30) Business Days to cure. Should Company not cure the material breach within such thirty (30)-Business Day period, the specific license grant(s) identified by Hitachi, for the specific Licensed Product identified by Hitachi in the case of a Material Quality Breach, shall immediately terminate, and Company shall (i) cease all further use of the Licensed Marks on the goods that were subject to the terminated license grant(s) (the "**Terminated Products**"); (ii) cease all further shipment, distribution and sale of the Terminated Products bearing the Licensed Marks; and (iii) cease all further advertising, marketing and promotion of the Terminated Products using the Licensed Marks.

8.3 All rights and licensed granted to any Subsidiary of Company, including without limitation via sublicense from Company, shall automatically terminate if such Subsidiary ceases to meet the definition of "Subsidiary," as defined in this Branding Agreement.

8.4 Upon the termination or suspension of any of the licenses granted in Article 2 either pursuant to this Article 8 or as a result of the expiration of the license, Company shall immediately (i) cease shipping or distributing the products covered by such license grant that bear the Licensed Marks; and (ii) cease all further use of the Licensed Marks for such products. Company shall have no obligation to recall or stop the shipment or sales by its dealers or customers of any products covered by the terminated or suspended license and that bear the Licensed Marks shipped by Company prior to the termination or expiration or suspension of any license. Company may continue after any termination or expiration or suspension of any license to ship goods from which the Licensed Marks have been removed or obliterated (for example, by covering the Licensed Marks with a sticker).

8.5 To the extent that upon termination or expiration of any of the license grants in Section 2.1 or Section 2.2, Company has in inventory any Licensed Products or Repair Parts bearing any Licensed Marks, or Company Products that incorporate or use Electronic Labels, which were manufactured or remanufactured: (i) prior to Closing; or (ii) pursuant to and in compliance with this Branding Agreement and that have not been determined to have been the

basis of a Material Quality Breach and were not Suspended Products at the termination or expiration of the license grants and their material deficiencies have not been cured, Company shall be entitled to sell and distribute its inventory of such products provided it uses the same channels of trade and distribution that were used by Company prior to the Closing or that Company uses for comparable products distributed under Company's own marks.

8.6 Notwithstanding anything to the contrary in this Agreement, Sections 1, 3.1(f), 3.1(g), 10, 11, 12.2, 12.3, 14, 15 and this Section 8 shall, to the extent applicable, survive any expiration or termination of this Branding Agreement or of any licenses granted herein.

8.7 Except as set forth in Section 12.3, the Parties acknowledge that each shall have available to it all remedies at law or in equity (including specific performance) for breach of this Branding Agreement in addition to or as a substitute for all remedies that they may have under this Branding Agreement.

## 9. UNAUTHORIZED USE BY OTHERS

9.1 During the term of the licenses granted in Article 2, Company shall notify Hitachi in writing of any unauthorized use of the Licensed Marks by others promptly as such unauthorized use comes to its attention. Upon notification Hitachi may, at its discretion, choose to take action including but not limited to bringing, prosecuting or settling any claim involving the Licensed Marks. Should Hitachi decide to take such action, Hitachi shall bear all costs and expenses in connection therewith and shall retain exclusively any damage awards or settlement amounts recovered. If Hitachi believes that Company should be joined as a party to such action, Hitachi and Company will discuss in good faith Company's cooperation and cost recovery for its participation in any such action.

9.2 Upon obtaining written confirmation from Company that a Company Reseller no longer distributes any Hitachi-branded Product, Hitachi shall have the right to contact such Company Reseller and advise it that it must cease all use of the Licensed Hitachi Marks and any other indicia suggesting an affiliation or association with Hitachi (except as may be separately authorized under any agreement directly between Hitachi and such Company Reseller). Hitachi may advise such resellers that failure to comply with Hitachi's demands violates Hitachi's rights. Should such resellers continue to use the Licensed Hitachi Marks or other prohibited indicia, Hitachi may pursue all legal action against such resellers that Hitachi in its sole discretion deems appropriate and necessary. Company agrees to reasonably cooperate with Hitachi with respect to such actions, which may include providing the names and addresses of particular resellers of Company Products.

#### 10. ACKNOWLEDGMENTS AND COVENANTS

10.1 Company understands, acknowledges and agrees that, as between Company and Hitachi, all rights in the Licensed Marks (including all registrations thereof) throughout the world are the sole and exclusive property of Hitachi, free and clear from any claim or retention of rights thereto on Company's part and that Hitachi reserves all rights in the Licensed Marks, other than those expressly granted herein. Company shall do nothing inconsistent with such ownership, and it shall forbear from attacking the validity of the Licensed Marks, or any

registrations thereof, or Hitachi's title thereto. Company acknowledges that nothing in this Branding Agreement shall give Company any right, title or interest in the Licensed Marks at any time other than the right of Company to use the Licensed Marks in accordance with the licenses granted in Article 2 and the rights granted in Section 8.5.

10.2 Company shall not knowingly take or omit to take, or authorize the taking or omission of, any action that could reasonably be expected to dilute or adversely affect Hitachi's claims to ownership of the Licensed Marks or the validity, enforceability, registration, or application for registration of the Licensed Marks.

10.3 All use of the Licensed Marks by Company shall inure to the benefit of Hitachi. During the term of this Branding Agreement and thereafter, Company shall not apply to register the Licensed Marks, any phonetic equivalent thereof, or any mark or term confusingly similar thereto as, or as part of, a trademark, service mark, trade name, business name, domain name or other indication of source or origin in any jurisdiction worldwide.

10.4 Company shall ensure that its Subsidiaries do not take any action that Company is prohibited from taking as specified in this Branding Agreement and Company shall not authorize any third party to take any action that Company is expressly prohibited from taking under the terms of this Branding Agreement.

10.5 Nothing herein shall restrict or otherwise affect the right of Hitachi to obtain relief against Company or any third party for any acts of trademark infringement, unfair competition or dilution.

## 11. **INDEMNIFICATION**

11.1 Company shall pay, indemnify, reimburse and hold harmless Hitachi and its Subsidiaries (for purposes of this Article 11, "**Hitachi Protected Parties**"), from and against any and all Losses to the extent incurred by any of them, as a result of, arising from or with respect to: (a) the sale, distribution, promotion or marketing of the Licensed Products under the Licensed Marks by Company, or by Company's Resellers other than to any Hitachi Protected Party; (ii) any actions or omissions by Company with respect thereto; and (iii) the use by Company of the Licensed Marks in a manner prohibited by this Branding Agreement; provided, however, that Company's obligations under this Section 11.2 shall not include Losses to the extent arising out of any breaches by Hitachi of any Transaction Document with respect to which Hitachi is obligated to indemnify Buyer or Buyer Parent under any Transaction Document.

11.2 The provisions of Sections 9.4 and 9.6 of the Purchase Agreement are incorporated, *mutatis mutandis*, into this Article 11 by reference, with respect to indemnification rights and obligations of the Parties arising under Section 11.1 of this Branding Agreement.

## 12. <u>REPRESENTATIONS AND WARRANTIES; LIMITATIONS OF LIABILITY</u>

12.1 Each Party represents and warrants that it has the corporate power and authority to enter this Branding Agreement, and to carry out the terms and obligations set forth in this Branding Agreement, and that the persons executing this Branding Agreement on its behalf have the authority to act for and bind the Party.

12.2 EXCEPT AS SPECIFICALLY PROVIDED IN THIS ARTICLE 12, NEITHER PARTY MAKES ANY OTHER WARRANTIES, EITHER EXPRESS OR IMPLIED, AS TO ANY OTHER MATTER WHATSOEVER, AND EACH PARTY HEREBY EXPRESSLY DISCLAIMS ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR ANY PARTICULAR PURPOSE, ACCURACY, NON-INFRINGEMENT, AND TITLE, AND ANY WARRANTIES THAT MAY ARISE FROM COURSE OF DEALING, COURSE OF PERFORMANCE OR USAGE OF TRADE.

12.3 Except as expressly set forth herein, neither Party shall be liable for any indirect, incidental, consequential, special or punitive damages arising out of or relating to this Branding Agreement, irrespective of whether such Party has been advised of the possibility of any such damages. The foregoing limitation of liability shall not apply with respect to any liability arising under, out of or in connection with Section 11 or any infringement, misappropriation or other violation of any Intellectual Property Rights of a Party.

## 13. NOTICES AND OTHER COMMUNICATIONS

All notices and other communications required or permitted hereunder shall be in writing and shall be mailed by Federal Express or other internationally recognized express carrier, registered or certified mail, postage prepaid, or otherwise delivered by hand or by messenger, addressed to a Party at the addresses set forth below, or at such other address as a Party may substitute by written notice provided to the other Party in such manner. Such notices shall be deemed to have been served when delivered.

## If to Hitachi:

Hitachi, Ltd. Business Development Office 6-6 Marunouchi 1-chome Chiyoda-ku Tokyo 100-8280, Japan Attention: General Manager Facsimile: 81-3-4564-6260

## If to Company:

Viviti Technologies Ltd. c/o Western Digital Corporation 3355 Michelson Drive, Suite 100 Irvine, California 92612 Attention: General Counsel Facsimile: (949)672-9612 E-mail: <u>Michael.Ray@wdc.com</u>

## 14. ADR, GOVERNING LAW AND JURISDICTION

14.1 The Parties agree that, in the event of a dispute arising under this Branding Agreement, the Parties shall agree to attend a non-binding mediation prior to bringing any action

in a court of law. The mediation shall be facilitated by a neutral third-party, selected by the Parties. A representative from each Party with authority to settle on behalf of that party shall attend the mediation in person. The purpose of the mediation is to resolve the disputed issue(s) without the need for formal legal proceedings.

14.2 This Branding Agreement shall be governed by and construed in accordance with the laws of the State of Delaware as applied to contracts entered into and fully performed in such state.

14.3 The Parties agree that the Delaware Court of Chancery (and if jurisdiction in the Delaware Court of Chancery shall be unavailable, any Delaware State court and the Federal court of the United States of America sitting in the State of Delaware) shall have exclusive jurisdiction of interpreting or enforcing this Branding Agreement, provided, however, that, notwithstanding the foregoing, either Party may (i) seek injunctive or other preliminary relief, (ii) enforce its Intellectual Property Rights, and (iii) seek recognition and enforcement of any order or judgment, in any court of competent jurisdiction. The parties expressly waive their right to a trial by jury.

## 15. <u>MISCELLANEOUS</u>

15.1 Nothing contained in this Branding Agreement shall be construed as conferring any right to use in advertising, publicity or other promotional activities any name, trade name, trademark, or other designation of either Party hereto (including any contraction, abbreviation, or simulation of any of the foregoing) except as specifically provided herein; and each Party hereto agrees not to use the existence of this Branding Agreement or any provision thereof in any advertising or sales promotional activity, without the express written approval of the other Party.

15.2 The Parties acknowledge that this Branding Agreement is not and does not purport to be a full recitation of Hitachi's rights in and to the Licensed Marks, which rights may not be limited to the terms of this Branding Agreement.

15.3 Except for the Purchase Agreement and other Transaction Documents, (a) this Branding Agreement contains the complete agreement between, and contains all of the promises and undertakings made to each other by, the Parties regarding the subject matter of this Agreement, and (b) any and all prior agreements, representations, negotiations, and undertakings between the Parties, oral or written, express or implied, with respect to the subject matter hereof, are hereby superseded and merged by and into this Branding Agreement, effective as of the Closing Date. This Agreement may not be revised or modified without the mutual written consent of the Parties.

15.4 If any paragraph, provision or clause of this Branding Agreement shall be found or held to be invalid or unenforceable by a court or other decision-making body of competent jurisdiction, in a judgment from which no further appeal can be taken, the remainder of this Branding Agreement shall remain valid and enforceable, and to the extent required in the pursuit of this Branding Agreement, the Parties shall negotiate in good faith a substitute, valid and enforceable provision that reflects the Parties' intent in entering this Branding Agreement.

15.5 This Branding Agreement may be executed in identical counterparts, each of which shall constitute an original and all of which shall constitute one and the same agreement.

[Remainder of this page is intentionally left blank]

# **IN WITNESS WHEREOF**, the Parties have caused their duly authorized officers to execute this Agreement on the dates indicated below.

For and on behalf of Hitachi, Ltd.	
By:	Date:
Name:	
Title:	
For and on behalf of Viviti Technologies Ltd.	
By:	Date:
Name:	
Title:	
	[Signature page for Branding Agreement]

# <u>EXHIBIT H</u>

# EXHIBIT G TO THE STOCK PURCHASE AGREEMENT Form of Secondment Agreement

[see attached]

Form of Secondment Agreement

# FORM OF

# SECONDMENT AGREEMENT

This SECONDMENT AGREEMENT (this "<u>Agreement</u>") is made and entered into by and between [\_\_\_\_], a company incorporated under the laws of Japan (the "<u>Seconding Entity</u>"), and [\_\_\_\_], a company incorporated under the laws of Japan (the "<u>Receiving Entity</u>" and together with the Seconding Entity, the "<u>Parties</u>" and each a "<u>Party</u>"), with respect to the terms and conditions for seconding employees (the "<u>Secondee</u>") from the Seconding Entity to the Receiving Entity pursuant to that certain stock purchase agreement (as amended, the "Stock Purchase Agreement"), dated as of March 7, 2011, entered into by and among Western Digital Corporation, a corporation incorporated under the laws of Delaware, Western Digital Ireland, Ltd., a corporation organized under the laws of Japan and Viviti Technologies Ltd., a company incorporated under the laws of the Republic of Singapore, as follows.

### 1. **D**EFINITIONS; INTERPRETATION

1.1 <u>Definitions</u>. Bracketed terms used in this Agreement are defined or referenced in <u>Annex A</u> attached hereto.

# 2. Secondment

# 2.1 <u>Secondee and Secondment Period</u>.

(a) <u>Secondment</u>. On the date hereof, the Seconding Entity shall use its reasonable best efforts to second to the Receiving Entity the employees referred to on <u>Schedule 2.1</u> (each such employee, a "<u>Secondee</u>").

# (b) <u>Secondment Period</u>.

(i) The initial period of secondment for each Secondee shall be from the date hereof until March 31, 2014 and thereafter shall, subject to the Secondee's consent, automatically renew for successive two-year periods (the initial and any renewal terms, collectively, the "Secondment Period"), [except that either Party may elect not to renew the Secondment Period by giving written notice of non-renewal no later than three months prior to the end of the then-current two-year period (whether an initial or renewal period)]<sup>1</sup>; provided, that the Parties may agree otherwise in writing with respect to any Secondee.

(ii) At any time during the Secondment Period, the Receiving Entity may terminate the secondment of any Secondee (A) by written agreement of the Seconding Entity and the Receiving Entity, and (B) upon written notice from the Receiving Entity to the Seconding Entity that such Secondee has done or omitted to do anything which would entitle the

<sup>&</sup>lt;sup>1</sup> The bracketed language shall only apply to the secondment of Non-R&D/R&D Related Secondees described in Section 6.12(e)(ii) of the Stock Purchase Agreement.

Receiving Entity to terminate such Secondee's employment (*chokai kaiko*) pursuant to the employment rules, regulations and policies of the Receiving Party if such Secondee were a direct employee of the Receiving Party. Upon written notice from (X) the Seconding Entity to the Receiving Entity stating that such Secondee has voluntarily retired or resigned, or (Y) the Receiving Entity to the Seconding Entity that such Secondee has done or omitted to do such things as described in clause (B) of this paragraph, the secondment of such Secondee hereunder shall terminate upon the effectiveness of such retirement, resignation, or receipt of the notice that such Secondee has so acted or omitted to act, as applicable.

2.2 <u>Secondment Principles</u>. During the Secondment Period, each Secondee shall remain an employee of the Seconding Entity but shall (a) be made available on a full-time basis to the Receiving Entity, (b) be managed by the Receiving Entity, (c) be subject to the employment rules, regulations and policies of the Receiving Entity with respect to working terms and conditions such as working hours, break time and any holidays, unless otherwise set forth herein, (d) perform such duties and provide such services at such times and at such places as the Receiving Entity may require, and (e) act in accordance with and subject to the instructions of the Receiving Entity. The Parties acknowledge and agree that no Secondee shall be under the control or supervision of the Seconding Entity during the Secondment Period and agree that the Seconding Entity shall not be responsible to the Receiving Entity for the quality, nature or sufficiency of the services performed by its Secondee during such period; provided, however, that the Receiving Entity shall neither relocate any Secondee nor materially change any Secondee's duties, working hours or any other term or condition of his or her service without obtaining the Seconding Entity's written approval with respect to any relocation or material change, which approval shall not be unreasonably withheld.

# 2.3 <u>Working Terms and Conditions</u>.

(a) <u>Secondment Costs and Expenses</u>. The basis for calculation of employment cost, allotment, and method of payment to the Secondee related to the performance of services by the Secondee during the Secondment Period shall be subject to the terms in Annex B<sup>2</sup>

(b) <u>Leave</u>. The Secondee may obtain leave pursuant to the employment rules, regulations and policies of the Seconding Entity, provided that, the Secondee shall submit a prior leave application form to its superior at the Receiving Entity. In case the annual paid leave requested by the Secondee would interfere with the operation of the Receiving Entity, it shall be entitled to order the Secondee to take such annual paid leave at different period.

(c) <u>Receiving Entity Non-Disclosure Rules</u>. Each Secondee shall be bound by any and all confidentiality rules, policies, agreements or arrangements (the "<u>Confidentiality Rules</u>") of the Receiving Entity. With respect to each Secondee, the Seconding Entity hereby waives its rights under any and all Confidentiality Rules of the Seconding Entity to the extent such Confidentiality Rules of the Seconding Entity are inconsistent with the Confidentiality Rules of the Receiving Entity shall take adequate measures such as

<sup>&</sup>lt;sup>2</sup> Please note that there are three annexes for the Secondment Agreement between Chuo Shoji Ltd. and HGST Japan since there are two secondees who each have different tables for allocation of cost.

execution of a non-disclosure agreement, control or return of materials or media that include or record the Confidential Information (as defined in <u>Section</u> <u>6.1</u>), to the extent practically possible, during and after the Secondment Period, in order to prevent divulgation of the Confidential Information which the Secondee may be aware of in the course of performing the service to the Receiving Entity.

# (d) <u>Employee Intellectual Property Rules</u>.

(i) Without limiting the generality of <u>Section 2.2</u>, the Seconding Entity acknowledges that ownership of any and all intellectual property rights in any inventions, trade secrets, or work product generated by such Secondee in the course of and in connection with such Secondee's duties during the Secondment Period (the "<u>Employee IPR</u>") shall be treated in accordance with the rules, policies, agreements or arrangements for ownership of Employee IPR (the "<u>IPR Rules</u>") of the Receiving Entity.

(ii) The Seconding Entity waives and shall not exercise any rights, (A) to require any Secondee to assign or transfer any Employee IPR to any person or entity, or (B) to apply for the registration thereof, in each case, under the Seconding Entity's IPR Rules, in a manner that would be inconsistent with the Receiving Entity's IPR Rules. Notwithstanding the preceding sentence, if the Seconding Entity acquires any interest in any Employee IPR, then, pursuant to the Receiving Party's IPR Rules that such Employee IPR should be owned by the Receiving Party, the Seconding Entity shall promptly assign such Seconding Entity's interest in the Employee IPR to the Receiving Entity.

(e) <u>Changes to the Seconding Entity's Working Terms and Conditions</u>. Notwithstanding anything in this Agreement to the contrary but to the extent permitted by the Stock Purchase Agreement, the Seconding Entity and its Affiliates shall, in their sole discretion and from time to time, have the right to terminate or modify any of its rules, regulations and policies relating to the Secondee and the right to adopt new plans or policies, at any time, related to any Compensation and Benefit Arrangements. For the avoidance of doubt and notwithstanding anything herein to the contrary, the Secondee may participate in any Compensation and Benefit Arrangements, provided that, it satisfies all of the eligibility requirements, as applicable, set forth in the rules related to such Compensation and Benefit Arrangements of the applicable Party pursuant to <u>Annex B</u>.

2.4 <u>Reimbursement</u>. All costs and expenses and other amounts incurred by the Seconding Entity with respect to any Secondee shall accrue on a calendar month basis (except as provided in <u>Section 4.2</u>) and shall be reimbursed by the Receiving Entity pursuant to <u>Annex B</u>. The Receiving Entity shall make such reimbursement to the Seconding Entity by wire transfer of immediately available funds by the end of each calendar month in which it receives an invoice from the Seconding Entity on or before the tenth (10th) day of such calendar month. In the event that the Receiving Entity receives an invoice from the Seconding Entity of any calendar month, the Receiving Entity shall make such reimbursement to the Seconding Entity by the end of the following calendar month. All invoices shall be denominated in Japanese yen and shall be accompanied by reasonable documentation evidencing the breakdown for such invoiced amounts in reasonable detail. If the Secondment Period with respect to any Secondee starts or ends on a day other than on the first or last day (respectively) of a calendar

month, the amount of any costs and expenses to be reimbursed by the Receiving Entity shall be prorated based on the actual number of calendar days worked in such month. For the avoidance of doubt, reimbursable amounts hereunder shall mean the amount incurred by a company pursuant to the rules, regulations and policies related to the Compensation and Benefit Arrangements as applicable to the Secondee, and actual payroll and other administrative costs and expenses. The Receiving Entity shall pay the gross amount of any present or future sales, use, excise, value-added, ad valorem or any other tax applicable to the furnishing by the Seconding Entity of benefits to the Secondee (other than income taxes), or the Seconding Entity shall be provided with a tax exemption certificate acceptable to applicable governmental authorities.

2.5 <u>Relocation Expenses</u>. If the Receiving Entity determines that a Secondee is required to relocate in order to perform his or her duties for the Receiving Entity, the Receiving Entity shall pay all costs and expenses associated with the relocation in accordance with the Receiving Entity's rules, regulations and policies.

2.6 <u>Transfer</u>. During the Secondment Period, the Seconding Entity and the Receiving Entity may discuss and negotiate the transfer of employment of any Secondee to the Receiving Entity or any of its affiliates.

2.7 <u>Unions; Employees</u>. The secondment of Secondee to the Receiving Entity shall be made upon adequate mutual consultation between the Seconding Entity and its labor union except as otherwise agreed in the collective agreement therebetween. The Parties shall use reasonable best efforts to complete procedures necessary or required in relation to the labor union to enable the secondment of any Secondee to the Receiving Party.

### 3. CERTAIN NOTICES

During the Secondment Period, the Seconding Entity shall notify the Receiving Entity promptly on becoming aware of (a) any threatened labor strike, walkout, slowdown, stoppage, picketing, or other material dispute with respect to Secondee, (b) any notice from a Secondee to the Seconding Entity of an election to terminate employment, or (c) any allegation by a Secondee or a labor union representing the Secondee of any unfair labor practices, discrimination, or breach of contract by the Seconding Entity.

### 4. **RETIREMENT BENEFITS**

4.1 <u>Retirement Benefits</u>. In the event that a Secondee retires (whether voluntarily or involuntarily) from the Seconding Entity during his or her Secondment Period, such Secondee's retirement benefits, including both lump-sum payment and pension (hereinafter the same), shall be calculated based on such Secondee's aggregated service years with the Seconding Entity and the Receiving Entity through the effective date of retirement, determined in accordance with the rules, regulations and policies of the Seconding Entity as set forth in <u>Annex B</u>. Notwithstanding the provisions in Section 2.4, all retirement benefit costs and expenses incurred by the Seconding Entity, with respect to any Secondee during his or her Secondment Period, shall be reimbursed to the Seconding Entity by the Receiving Entity once every six months. Any retirement benefit payments to be made to such Secondee shall be borne on a pro rata basis by the Seconding Entity

and the Receiving Entity, based on such Secondee's periods of service with the Seconding Entity and the Receiving Entity (with his or her Secondment Period counted as time with the Receiving Entity); <u>provided</u>, <u>however</u>, any retirement benefit costs and expenses already reimbursed to the Seconding Entity by the Receiving Entity, if any, shall be deducted from any retirement benefit payment borne by the Receiving Entity.

4.2 <u>Pre-Secondment Period Accrued Benefits</u>. Any and all liabilities relating to any Compensation and Benefit Arrangements maintained or contributed to, or any benefits under applicable laws required to be maintained or contributed to, by the Seconding Entity for the benefit of any Secondee, calculated with respect to any time period ending prior to the commencement of the Secondment Period, shall be for the account of and borne by the Seconding Entity regardless of whether any such benefit is to be paid to any Secondee at any time after the commencement of the Secondment Period. Without prejudice to the generality of the foregoing, any benefits under the Compensation and Benefit Arrangements to be paid during the Secondment Period which vary based on length of employment or services shall be calculated giving effect to, and credit for, time served by the Secondee with the Seconding Entity.

### 5. CLAIMS AND LIABILITIES

5.1 <u>Consequential Damages Waiver</u>. EXCEPT WITH RESPECT TO THE PARTIES' RESPECTIVE INDEMNITY OBLIGATIONS UNDER <u>SECTION 7</u>, TO THE EXTENT PERMITTED BY APPLICABLE LAW, IN NO EVENT SHALL EITHER PARTY BE LIABLE TO THE OTHER PARTY UNDER ANY LEGAL THEORY FOR ANY INDIRECT, SPECIAL, INCIDENTAL, CONSEQUENTIAL OR PUNITIVE DAMAGES, OR ANY DAMAGES FOR LOSS OF PROFITS, REVENUE OR BUSINESS, EVEN IF SUCH PARTY HAS PREVIOUSLY BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES. FOR PURPOSES OF THIS <u>SECTION 5.1</u>, "PARTIES" AND "PARTY" SHALL INCLUDE ANY AFFILIATES.

5.2 <u>Disclaimer</u>. EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THIS AGREEMENT, NEITHER PARTY MAKES (AND EACH PARTY HEREBY EXPRESSLY DISCLAIMS) ANY REPRESENTATIONS OR WARRANTIES, WHETHER EXPRESS, IMPLIED OR STATUTORY, INCLUDING, WITHOUT LIMITATION, IMPLIED WARRANTIES OF FITNESS FOR A PARTICULAR PURPOSE, AND ANY WARRANTIES THAT MAY ARISE FROM COURSE OF PERFORMANCE, COURSE OF DEALING OR USAGE OF TRADE. FOR PURPOSES OF THIS <u>SECTION 5.2</u>, "PARTIES" AND "PARTY" SHALL INCLUDE ANY AFFILIATES.

### 6. Confidential Information

6.1 "<u>Confidential Information</u>" means all information disclosed, pursuant to or related to this Agreement, by a Party (the "<u>Discloser</u>") in writing, orally or in any other form concerning the businesses, affairs, properties, employees, finances, products, technologies and operations of the Discloser to the other Party (the "<u>Recipient</u>") and any notes, analyses, compilations, studies, forecasts, interpretations or other documents or records that contain any such information. Confidential Information does not include information or material that (a) is now, or hereafter

becomes, through no act or failure to act on the part of the Recipient, generally known or available; (b) is or was known by the Recipient at or before the time such information or material was received from the Discloser; (c) is furnished to the Recipient by a third party that is not under an obligation of confidentiality to the Discloser with respect to such information or material; or (d) is independently developed by the Recipient.

6.2 The Recipient shall hold all Confidential Information in confidence and shall not disclose to third parties for any purpose whatsoever other than as necessary in order to fulfill its obligations or exercise its rights under this Agreement. The Recipient shall take reasonable measures to protect the confidentiality of the Discloser's Confidential Information in a manner that is at least protective as the measures it uses to maintain the confidentiality of its own Confidential Information of similar importance. Notwithstanding the foregoing, the Recipient may disclose the Discloser's Confidential Information (a) to employees and consultants that have a need to know such information, provided that each such employee and consultant is under a duty of nondisclosure that is substantially comparable with the confidentiality and nondisclosure provisions herein, and (b) to the extent the Recipient is legally compelled to disclose such Confidential Information, provided that if permitted by applicable law and regulations the Recipient shall give advance notice of such compelled disclosure to the Discloser, and shall cooperate with the Discloser in connection with any efforts to prevent or limit the scope of such disclosure and use of the Confidential Information.

# 7. INDEMNIFICATION

### 7.1 <u>Indemnification</u>.

(A) <u>Indemnification by the Receiving Entity</u>. The Receiving Entity shall indemnify, defend and hold harmless the Seconding Entity and its officers, directors and employees from and against any and all claims related to (a) the actions or omissions of any Secondee in the course and scope of his or her services to the Receiving Entity during his or her Secondment Period, and (b) any breach by the Receiving Entity of this Agreement.

(B) <u>Indemnification by the Seconding Entity</u>. The Seconding Entity shall indemnify, defend and hold harmless the Receiving Entity and its officers, directors and employees from and against any and all claims related to (a) the actions or omissions of the Seconding Entity with respect to any Secondee, and (b) any breach by the Seconding Entity of this Agreement.

7.2 <u>Indemnification Procedures</u>. If an indemnified party (an "<u>Indemnified Party</u>") becomes aware of a claim for which indemnification may be sought hereunder, then such Indemnified Party shall give notice thereof (a "<u>Claim Notice</u>") to the indemnifying party (an "<u>Indemnifying Party</u>") as promptly as practicable; provided, however, that the failure of an Indemnified Party to give a timely Claim Notice hereunder shall not affect its rights to indemnification hereunder. If the Indemnifying Party notifies in writing to the Indemnified Party within 20 Business Days after receipt of a Claim Notice that the Indemnifying Party is obligated under the terms of its indemnity hereunder in connection with the claim set forth in the Claim Notice, then the Indemnifying Party shall be entitled, at its own cost, risk and

expense, to defend or negotiate the settlement of such claim through its representative. Notwithstanding the foregoing, the Indemnified Party shall have the right to defend and negotiate the settlement of such claim through its representative at its sole discretion. Either Party appointing a representative to defend and settle a claim made by a third party shall, from time to time, advise of the status of the defense and consult in good faith with respect to the defense with, the other Party that has not appointed its representative. Neither Party shall make settlement in connection with a claim with a third party without prior consent of the other Party (such consent shall not be unreasonably withheld). The Indemnifying Party shall indemnify the Indemnified Party in accordance with any settlement or all other resolution, or judgment or order by the court (by the appellate court if appealed) under this Section 7.2.

### 8. TERM AND TERMINATION

8.1 <u>Term</u>. This Agreement shall be effective as of the date hereof, and shall continue in full force and effect until the expiration of the Secondment Period for all of the Secondee (the "<u>Term</u>").

### 8.2 <u>Effect of Termination</u>.

(a) <u>Continuing Liability</u>. Termination of this Agreement for any reason shall not (a) release either Party from any liability, obligation or agreement which has already accrued at the time of termination, or (b) be deemed a waiver or release of, or otherwise be deemed to prejudice or adversely affect, any rights, remedies or claims, whether for damages or otherwise, which a Party may have under this Agreement, at law or otherwise, or which may arise out of or in connection with such termination.

(b) <u>Survival</u>. The provisions of <u>Sections 2.3(a)</u>, 2.4, 4.1, and 4.2 with respect to any payment obligations which accrued prior to, and remain outstanding as of, the expiration of the Term, and <u>Sections 1.1, 2.3(c)</u>, 2.3(d), 5, 6, 7, 9, 11, and 12, and this <u>Item (b)</u> shall survive any termination of this Agreement.

### 9. Notices

9.1 <u>Addresses</u>. Any notice, claim or demand in connection with this Agreement or with any litigation under this Agreement (each a "<u>Notice</u>") shall be deemed to be sufficiently given if delivered or sent to the recipient at its fax number or address set out in <u>Schedule 9.1</u> or any other fax number or address notified to the sender by the recipient for the purposes of this Agreement.

9.2 Form. Any Notice shall be deemed to have been received on the next Business Day in the place to which it is sent, if sent by fax, or three Business Days from the date deposited with a courier, if sent by international courier, or five Business Days from the date of posting, if sent by post.

9.3 <u>Contact Person</u>. The person named by each Party in <u>Schedule 9.1</u> shall be the primary point of contact at that Party for all matters relating to this Agreement. Each Party agrees that it will exercise its rights and send and receive all notices under this Agreement through such person or to any alternative person specified by a Party by giving not less than five Business Days' written notice to the other Party.

### 10. General

10.1 <u>Whole Agreement</u>. This Agreement contains the whole agreement between the Parties relating to the subject matter of this Agreement at the date hereof to the exclusion of any terms implied by applicable law which may be excluded by contract and supersedes any previous written or oral agreement between the Parties in relation to the matters dealt with in this Agreement except as otherwise set forth in the Stock Purchase Agreement.

10.2 <u>No Partnership</u>. Nothing in this Agreement shall be deemed to constitute a partnership between the Parties and no Party shall be deemed to be the representative of the other Party for any purpose.

10.3 <u>Waiver</u>. No failure of a Party to exercise, and no delay by it in exercising, any right, power or remedy in connection with this Agreement (each, a "<u>Right</u>") shall operate as a waiver of that Right, nor shall any single or partial exercise of any Right preclude any other or further exercise of such Right or the exercise of any other Right. Any express waiver with regard to any breach of this Agreement shall not be deemed to be a waiver of any subsequent breach.

10.4 <u>Variation</u>. No variation of this Agreement shall be effective unless in writing and signed by or on behalf of each of the Parties.

10.5 <u>Assignment</u>. Except as otherwise provided in this Agreement, the Parties may not assign or transfer, or delegate all or any part of their rights or obligations under this Agreement or any benefit arising under or out of this Agreement.

10.6 <u>Further Assurances</u>. At any time after the date hereof, each Party shall, at its own cost, execute such documents and do such acts and things for the purpose of giving to the other Party the full benefit of the provisions of this Agreement.

10.7 <u>Invalidity</u>. If any provision in this Agreement shall be held to be illegal, invalid or unenforceable, in whole or in part, under any enactment or rule of law, such provision or part shall to that extent be deemed not to form part of this Agreement but the legality, validity and enforceability of the remainder of this Agreement shall not be affected.

10.8 <u>Counterparts</u>. This Agreement may be entered into in any number of counterparts, each of which shall be deemed an original, and all of which taken together shall constitute one and the same instrument.

10.9 <u>Costs</u>. Each Party shall bear all costs incurred by it in connection with the preparation, negotiation and entry into this Agreement and the documents to be entered into pursuant to this Agreement.

10.10 Language. This Agreement shall be in the Japanese language, and any translation of this Agreement into any other language other than Japanese by either Party shall be for convenience purposes only. In the event of any conflict between this Agreement and a convenience translation, this Japanese version shall prevail.

10.11 <u>Governing Law</u>. This Agreement and the legal relations among the Parties shall be governed by and construed in accordance with the laws of Japan.

### 11. NEGOTIATION PROCEDURES; CONSENT TO JURISDICTION, SERVICE OF PROCESS AND VENUE

### 11.1 <u>Negotiation Procedure</u>.

(a) <u>Good Faith Discussions</u>. The Parties intend that all disputes, controversies or differences between the Parties arising out of or in relation to or in connection with this Agreement shall be settled by the Parties amicably through good faith discussions upon the written request of either Party.

(b) <u>Negotiation Procedures</u>. Prior to filing suit, instituting a judicial or administrative proceeding, or seeking other judicial or governmental resolution (each, a "<u>Proceeding</u>") in connection with any dispute between the Parties or any of their subsidiaries arising out of or in relation to or in connection with this Agreement, the Parties will attempt to resolve such dispute by good faith negotiations. Such negotiations shall proceed as follows:

(i) Any Party may send a written notice to another Party requesting such negotiations. Promptly following receipt of such notice by the receiving Party, each Party shall cause the individual designated by it as having general responsibility for this Agreement to meet in person with the individual so designated by the other Party to discuss the dispute.

(ii) If the dispute is not resolved within thirty (30) days after the first meeting between such individuals (or, if earlier, within forty five (45) days of the notice referred to in clause (a) above), then, upon the written request of any Party, each Party shall cause the individual designated by it as having general responsibility for the overall relationship defined by this Agreement to meet in person with the individual so designated by the other Party to discuss the dispute.

(iii) If the dispute is not resolved within fifteen (15) days after the first meeting between such individuals (or, if earlier, within thirty (30) days of the notice referred to in clause (ii) above), then, upon the written request of either Party, each Party shall nominate one corporate officer or director, which corporate officers or directors shall meet in person and attempt in good faith to negotiate a resolution to the dispute.

Except and only to the limited extent provided in <u>Item (c)</u>, neither Party shall file suit, institute a Proceeding or seek other judicial or governmental resolution of the dispute until at least thirty (30) days after the first meeting between the corporate officers or directors described in clause (iii) above (or if earlier forty five (45) days after the notice referred to in such clause (iii)) but after the expiration of such periods, either Party may file suit, institute a Proceeding or seek other judicial or governmental resolution. For purposes of this Agreement, the procedures set forth in this Item (b) shall be referred to as the "<u>Negotiation Procedures</u>".

(c) <u>Institution of Proceedings</u>. Notwithstanding the provisions of <u>Item (b)</u>, any Party may institute a Proceeding at any time seeking a temporary restraining order including preliminary injunction, if necessary in the sole judgment of that Party to avoid material harm to

its property, rights or other interests, before commencing, or at any time during the course of, the Negotiation Procedures described in <u>Item (b)</u>. In addition, any Party may file an action prior to the commencement of or at any time during or after the Negotiation Procedures in <u>Item (b)</u> if in the sole judgment of that Party it is necessary to prevent the expiration of a statute of limitations or filing period or the loss of any other substantive or procedural right.

11.2 <u>Consent to Jurisdiction, Service of Process and Venue</u>. Each of the Parties irrevocably and unconditionally submits to the exclusive jurisdiction of the Tokyo District Courts for the purposes of any action, suit, proceeding, hearing, order, charge or complaint (each, an "<u>Action</u>") or other proceeding arising out of or in relation to or in connection with this Agreement (and agrees that no such Action or proceeding arising out of or in relation to or in relation to or in connection with this Agreement shall be brought by it or any of the subsidiaries except in such courts).

[Remainder of page intentionally blank.]

By:	
Name:	
Title:	

By:	
Name:	
Title:	

Signature Page to Secondment Agreement

### ANNEX A

#### DEFINITIONS

"Action" has the meaning specified in Section 11.2.

"Agreement" has the meaning set forth in the preamble.

"Affiliate" means any person or entity now or hereafter controlling, controlled by or under common control with another person or entity.

"<u>Compensation and Benefit Arrangements</u>" means any plan, program, policy, practice or arrangement relating to compensation, insurance, bonuses, vacation, profit-sharing, stock options, disability, pension, retirement, allowances, welfare, healthcare, unemployment or any other benefits to employees or former employees, their beneficiaries or dependents.

"Business Day" means a day, other than Saturday, Sunday or public holidays in Japan.

"Claim Notice" has the meaning specified in Section 7.2.

"Confidential Information" has the meaning specified in Section 6.

"Confidentiality Rules" has the meaning specified in Section 2.3.

"Discloser" has the meaning specified in Section 6.1.

"Employee IPR" has the meaning specified in Section 2.3(d)(i).

"Secondment Period" has the meaning specified in Section 2.1(b)(i).

"Indemnified Party" has the meaning specified in Section 7.2.

"Indemnifying Party" has the meaning specified in Section 7.2.

"IPR Rules" has the meaning specified in Section 2.3(d)(i).

"Negotiation Procedures" has the meaning specified in Section 11.1(b).

"Notice" has the meaning specified in Section 9.1.

"<u>Parties</u>" or "<u>Party</u>" has the meaning set forth in the preamble.

"Proceeding" has the meaning specified in Section 11.1(b).

"Receiving Entity" has the meaning set forth in the preamble.

"Recipient" has the meaning specified in Section 6.

"Right" has the meaning specified in Section 10.3.

"Secondee" has the meaning specified in Section 2.1(a).

"<u>Seconding Entity</u>" has the meaning set forth in the preamble.

"<u>Stock Purchase Agreement</u>" has the meaning specified in the preamble.

"Term" has the meaning specified in Section 8.1.

# <u>EXHIBIT I</u>

# EXHIBIT H TO THE STOCK PURCHASE AGREEMENT Form of Company Waiver and Release

[see attached]

Form of Company Waiver and Release

# SEPARATION AGREEMENT AND GENERAL RELEASE

### I, XXXX XXXX, agree to the following:

1. Benefits Payable. In exchange for signing this Separation Agreement and General Release ("Agreement"), I will receive the following benefits:

(a) Cash Payment: A lump sum payment of \$XXXX.XX, minus legally required deductions; and

(b) **COBRA Premiums:** A lump sum payment of **\$XXXX.XX**, minus legally required deductions, as a subsidy to pay my COBRA premiums for continuing medical, dental and vision coverage, based on my current elections. I understand that I must actively enroll in COBRA coverage and pay related premiums to maintain such insurance coverage.

Together, these benefits will be referred to as the "RA Payment," and are also subject to the terms and conditions set forth in the Hitachi Global Storage Technologies Resource Action Plan ("Hitachi RA Plan"), to which I agree to be bound.

2. **Employment Termination and Transition**. I agree that my employment with Hitachi Global Storage Technologies, Inc. ("Company") has ended or will end on **XXXXXX**, or on a later date as determined by the Company in its sole discretion. I understand that my eligibility for the RA Payment is dependent upon my remaining employed with the Company until the date determined by the Company. I will act reasonably and in good faith to ensure a smooth transition of my job responsibilities and, as requested, assist the Company in any other transitional activities within the scope of my past job duties, including without limitation, providing assistance, cooperation and information to Company management to allow an effective transition of my responsibilities.

3. **Released Parties.** I understand that this Agreement releases legal claims against the Company, its current and former parents (including without limitation, Hitachi Global Storage Technologies Netherlands B.V., Viviti Technologies Ltd., Hitachi, Ltd. and Western Digital Corporation), their subsidiaries, related companies, partnerships, joint ventures, or other affiliates, and, with respect to each of them, their predecessors and successors; and, with respect to each such entity, all of its past, present, and future employees, officers, directors, stockholders, owners, representatives, assigns, attorneys, agents, insurers, employee benefit programs (and the trustees, administrators, fiduciaries, and insurers of such programs), and any other persons acting by, through, under or in concert with any of the persons or entities listed in this paragraph (altogether, the "Released Parties").

4. **Release of Legal Claims.** I, on my own behalf, on behalf of my heirs, family members, executors, and assigns, hereby fully and forever release each Released Party from any claim, obligation, duty or cause of action relating to any matter of any kind ("Claims"), whether presently known or unknown, suspected or unsuspected, that I may possess arising from any acts, facts or omissions that have occurred up to and including the date I signed this Agreement, including without limitation:

1

Employee Initials\_\_\_\_\_

(a) any and all Claims relating to or arising from my employment relationship with the Company and the termination of that relationship;

(b) any and all Claims under the law of any jurisdiction including, without limitation, wrongful discharge of employment, constructive discharge from employment, termination in violation of public policy, discrimination (including age discrimination), whistleblower claims, retaliation, failure to accommodate a disability, breach of contract, both express and implied, breach of a covenant of good faith and fair dealing, both express and implied, promissory estoppel, negligent or intentional infliction of emotional distress, negligent or intentional misrepresentation, negligent or intentional interference with contract or prospective economic advantage, unfair business practices, defamation, libel, slander, negligence, personal injury, assault, battery, invasion of privacy, false imprisonment, and conversion;

(c) any and all Claims for violation of any federal, state or municipal statute, including, without limitation, Title VII of the Civil Rights Act of 1964, the Civil Rights Act of 1991, the Age Discrimination in Employment Act of 1967, the Americans with Disabilities Act of 1990, the Family and Medical Leave Act of 1993, the Worker Adjustment and Retraining Notification Act of 1988 ("WARN Act"), the Fair Labor Standards Act, the Employee Retirement Income Security Act of 1974, the California Fair Employment and Housing Act, the California Family Rights Act, the California Labor Code, the California WARN Act and any other similar state or local law;

- (d) any and all Claims for violation of the federal, or any state, constitution;
- (e) any and all Claims arising out of any other laws and regulations relating to employment or employment discrimination;

(f) any Claim for any loss, cost, damage, or expense arising out of any dispute over the non-withholding or other tax treatment of any of the proceeds that I have received under this Agreement; and

(g) any and all Claims for attorneys' fees and costs.

I agree that the release of claims set forth in this section shall operate as a complete general release as to the Claims released. This release does not apply to any obligations incurred under this Agreement, nor does it apply to legal claims that the law does not permit me to waive by signing this Agreement.

**5. Pursuit of Claims.** I acknowledge that I have not filed, initiated, or prosecuted (or caused to be filed, initiated, or prosecuted) or assigned any Claim, lawsuit, complaint, charge, action, compliance review, investigation, or proceeding with respect to any Claim this Agreement purports to waive.

6. Acknowledgement of Release of ADEA Claims. I acknowledge that I am waiving and releasing any rights I may have under the Age Discrimination in Employment Act of 1967 ("ADEA") and that this waiver and release is knowing and voluntary. The Company and I agree that this waiver and release does not apply to any rights or claims that may arise under the ADEA after the date I signed this Agreement. I acknowledge that the RA Payment is being provided in consideration for this Agreement and is in addition to anything of value to which I was already entitled.

2

Employee Initials\_\_\_\_\_

7. **Civil Code Section 1542.** I represent that I am not aware of any claim that I might have other than the Claims that are released by this Agreement. I acknowledge that I have been advised by legal counsel to the extent I desired and that I am familiar with the provisions of California Civil Code Section 1542, which provides as follows:

A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY HIM OR HER MUST HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR.

I, being aware of the above code section, expressly waive any rights I might have under such code section, as well as similar rights under any other statute or common law principles intended to protect against the waiver of unknown claims.

8. **Consideration of Release and Statistical Information.** I acknowledge that the Company is advising me to take this Agreement home, read it and carefully consider all of its terms before signing it.

(a) **Consideration Period for Employees Under 40**: If I am under the age of 40, I have at least 45 calendar days ("Consideration Period") after my termination date to consider this Agreement and to review all other documentation associated with the Hitachi RA Plan provided to me by the Company. I understand that I may take the full 45 business days to consider this information.

(b) Consideration Period for Employees 40 or Over: If I am forty years of age or older, I have at least 45 calendar days after receiving a copy of this Agreement ("Consideration Period") to consider this Agreement, the attached statistical data about terminated employees and all other documentation associated with the Hitachi RA Plan provided to me by the Company. I understand that I may take the full 45 calendar days to consider this information.

If I sign this Agreement before expiration of the applicable Consideration Period, I hereby certify that I knowingly and voluntarily waive the right to any remaining portion of the applicable Consideration Period, for reasons personal to me, with no pressure by any Company representative to do so. I have carefully considered this Agreement, have obtained satisfactory answers to any questions I had, and have determined that the Company has complied with all disclosure requirements necessary to obtain a valid ADEA and OWBPA release of claims. I have read this Agreement carefully, I fully understand what it means, and I am entering into it voluntarily.

**9. Revocation of Release.** If I am 40 years of age or older, I understand that I may change my mind about (i.e., revoke) my waiver and release under this Agreement within 7 days after I sign it, by delivering a written notice of revocation to the Human Resources Department responsible for my most recent work location. Revocation of my waiver does not alter or change the termination of my employment with the Company. If I am under 40 years old, I may not revoke this Agreement once I have signed it.

**10. This Agreement to be Kept Confidential**. I agree not to disclose the terms, amount, or existence of this Agreement to anyone other than a member of my immediate family, attorney, or

3

Employee Initials\_\_\_\_\_

other professional advisor and, even as to such a person, only if the person agrees to honor this confidentiality requirement. Such a person's violation of this confidentiality requirement will be treated as a violation of this Agreement by me. This Section does not prohibit my disclosure of the terms, amount, or existence of this Agreement to the extent necessary to enforce this Agreement, nor does it prohibit disclosures protected or required by law. I acknowledge that the Company would be irreparably harmed if this Section is violated.

**11. Confidential Information, Nonsolicitation and Nondisparagement**. I agree to remain bound by any Company or Released Party agreement or policy relating to confidential information, employee inventions, nonsolicitation, noncompetition, or similar matters to which I am now subject ("Surviving Agreements"). I agree not to criticize or disparage any Released Party, or the products, processes, policies, practices or research of any Released Party. Nothing in this paragraph or paragraph 10 shall prohibit me from complying with any lawful subpoena or court order or taking any other actions affirmatively authorized by law.

**12. Return of Property.** I have returned to the Company all files, documents, records, copies of the foregoing, computers, computer peripherals, cellular phones or other electronic mobile devices, Company-provided credit cards, keys, building passes, security passes, access or identification cards, and any other property of the Company in my possession or control. I agree not to incur any expenses, obligations, or liabilities on behalf of the Company.

**13. Non-Admission of Liability.** This Agreement is not an admission of guilt or wrongdoing by any Released Party. I agree that I have voluntarily executed this Agreement on my own behalf, and also on behalf of any heirs, agents, representatives, successors and assigns that I may have now or in the future.

**14. Information Concerning Severance Program**. This Agreement is offered as part of the Hitachi RA Plan. The Company separately has provided me with a copy of the Hitachi Global Storage Technologies Resource Action Plan and Summary Plan Description.

**15. Entire Agreement and Modification**. This Agreement, the Hitachi RA Plan, and any Surviving Agreements are the entire agreement between me and the Company regarding the subject matter of this Agreement. I acknowledge that the Company has made no representations or promises to me (such as that my former position will remain vacant), other than those contained in or referred to by this Agreement. All prior representations, understandings and agreements concerning the subject matter of this Agreement are hereby superseded. Once effective, this Agreement may not be modified in any manner, nor may any provision or any legal remedy with respect to it be waived, except by a writing signed by both me and the Company's most senior Human Resources executive. If one party successfully asserts that any provision in this Agreement is void, the rest of the Agreement shall remain valid and enforceable unless the other party to this Agreement elects to cancel it. If this Agreement is cancelled, I agree to repay to the Company the RA Payment that I received for signing it.

[Remainder of Page Intentionally Left Blank]

Employee Initials\_\_\_\_\_

4

PLEASE TAKE THIS AGREEMENT HOME AND READ IT CAREFULLY BEFORE SIGNING IT. THIS AGREEMENT INCLUDES A RELEASE OF KNOWN AND UNKNOWN CLAIMS, WHICH MEANS THAT WITH VERY LIMITED EXCEPTIONS YOU ARE WAIVING YOUR RIGHT TO TAKE LEGAL ACTION AGAINST THE COMPANY AND OTHER PARTIES FOR EMPLOYMENT RELATED DISPUTES. YOU ARE HEREBY ADVISED TO TAKE ADVANTAGE OF THE CONSIDERATION PERIOD IN PARAGRAPH 8 AND TO CONSULT AN ATTORNEY OF YOUR CHOICE (AT YOUR OWN COST) BEFORE SIGNING THIS AGREEMENT.

# PLEASE INITIAL EACH PAGE OF THIS AGREEMENT AND SIGN IT IMMEDIATELY BELOW.

### EMPLOYEE

Employee Signature

Employee Serial Number

HITACHI GLOBAL STORAGE TECHNOLOGIES

Director, Human Resources

Print Name		
Date		
Date		
Employee Initia	ıls	

5

### March [•], 2012

### Re: Separation Agreement and General Release

Reference is hereby made to that certain Stock Purchase Agreement (as may be amended from time to time, the "<u>SPA</u>"), dated as of March 7, 2011, by and among Western Digital Corporation (the "<u>Buyer Parent</u>"), Western Digital Ireland, Ltd. (the "<u>Buyer</u>"), Hitachi, Ltd. (the "<u>Seller</u>"), and Viviti Technologies Ltd. (the "<u>Company</u>") and the Separation Agreement and General Release (the "<u>Release</u>"), by and between the

Hitachi Global Storage Technologies, Inc.("<u>HGST</u>") and [**Employee Name**] ("<u>Employee</u>"), dated as of [•], 2012. Terms not defined herein shall have the meanings assigned to such terms in the SPA or Release, as appropriate.

For the purpose of clarification, nothing in the Release shall constitute a waiver or release by each Employee of any rights to any claims pursuant to any applicable directors' and officers' liability insurance policies or for indemnification, reimbursement, offset or recovery provided to such Employee under the SPA, applicable indemnification agreements or under the formation and corporate governance documents of HGST, the Company and each of their subsidiaries or other affiliates. Each Employee may rely on this letter in releasing their signature page to the Release to HGST.

Nothing in this letter is intended to modify the SPA or the Release in any respect except as expressly set forth herein.

Sincerely,

HITACHI GLOBAL STORAGE TECHNOLOGIES, INC.

By:

Name: Title:

cc: Western Digital Corporation

[Side Letter – Release]

# EXHIBIT J

# EXHIBIT I TO THE STOCK PURCHASE AGREEMENT Form of Reverse Services Agreement

[see attached]

Form of Reverse Services Agreement

### **REVERSE SERVICES AGREEMENT**

THIS REVERSE SERVICES AGREEMENT (this "<u>Agreement</u>") is entered into as of March \_\_\_, 2012 (the "<u>Effective Date</u>"), by and among Hitachi, Ltd., a company incorporated under the laws of Japan ("<u>Seller</u>"), and Viviti Technologies Ltd., a company incorporated under the laws of the Republic of Singapore and, as of the Closing, a wholly -owned subsidiary of Buyer Parent ("<u>Company</u>"). Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the Purchase Agreement (as defined below).

#### RECITALS

WHEREAS, on March 7, 2011, Buyer Parent, Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect wholly-owned subsidiary of the Buyer Parent ("<u>Buyer</u>"), Seller and Company entered into a Stock Purchase Agreement, as amended by a First Amendment to Stock Purchase Agreement, dated May 27, 2011, a Second Amendment to Stock Purchase Agreement, dated November 23, 2011, a Third Amendment to Stock Purchase Agreement, dated January 30, 2012, a Fourth Amendment to Stock Purchase Agreement, dated March 6, 2012, and a Sixth Amendment to Stock Purchase Agreement, dated March 6, 2012, and a Sixth Amendment to Stock Purchase Agreement, dated March 6, 2012 (as amended, through the date hereof, the "<u>Purchase Agreement</u>") and pursuant to, and subject to the terms thereof, Buyer acquired from Seller all of Seller's right, title and interest in and to the Stock;

WHEREAS, Company and the Subsidiaries provided certain services to Seller and its Affiliates prior to the Closing; and

WHEREAS, Company shall provide, and, as applicable, shall cause the Subsidiaries, Representatives and Authorized Third Parties (each as defined below) to provide, Seller and its Affiliates certain services reasonably necessary for the operation of Seller and its Affiliates following the Closing, pursuant to and in accordance with the terms of this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein and other good and valid consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

### ARTICLE I

### SERVICES

1.1 <u>Services</u>.

(a) Subject to the terms and conditions of this Agreement, Company shall provide, and, as applicable, shall cause the Subsidiaries, Representatives and Authorized Third Parties to provide, to Seller and its Affiliates the services described on Exhibit A attached hereto in Japan (and such other locations specified on Exhibit A attached hereto) (the "<u>Services</u>") for the period from the Effective Date until the termination or expiration of this Agreement pursuant to Article III below (the "<u>Services Period</u>") solely to the extent that (1) such Services are provided by Company or the Subsidiaries, Representatives and Authorized Third Parties to Seller or any of its Affiliates in Japan (and such other locations specified on Exhibit A attached hereto) as of the Effective Date, and (2) such Services are reasonably necessary to support during the Services Period the operation of Seller and its Affiliates in Japan (and such other locations specified on Exhibit A attached hereto) in all material respects as they were operated as of the Effective Date. Company shall provide, and shall cause the Subsidiaries, Representatives and Authorized Third Parties to provide, the Services to Seller and its Affiliates in substantially the same manner as Company, the Subsidiaries, Representatives and Authorized Third Parties provided such services to Seller and its Affiliates as of the Effective Date. For purposes of this Agreement, "<u>Representatives</u>" means, with respect to Company, its directors, officers, employees, financial advisors, attorneys, accountants, consultants, agents and other authorized representatives, acting in such capacity.

(b) Company shall, and as applicable, shall cause the Subsidiaries, Representatives and Authorized Third Parties to, (i) materially comply with all applicable Laws relating to the performance of the Services; and (ii) materially comply with any reasonable confidentiality, security, privacy or other policies of Seller and its Affiliates relating to the performance of the Services which have been provided to Company reasonably in advance.

(c) To the extent permitted by applicable Law, Company agrees to pass through to Seller and its Affiliates any rights Company may have with respect to Authorized Third Parties in connection with any failure by such Authorized Third Parties to materially comply with all applicable Laws relating to the performance of the Services or to materially comply with any reasonable confidentiality, security, privacy or other policies of Seller or any of its Affiliates, or Company or the Subsidiaries, relating to the performance of the Services.

(d) Seller shall use, and shall cause its Affiliates to use, the Services only for substantially the same purpose and in substantially the same manner and amount as the Services were used by Seller and its Affiliates as of the Effective Date; <u>provided</u>, <u>however</u>, that the scope and amount of the Services may be reduced by Seller as specified herein.

(e) Notwithstanding anything herein to the contrary, the Services shall not include (i) the provision of any funding or financial accommodation and (ii) any service that is directly provided by any third party to Seller or any of its Affiliates under an agreement between such third party and Seller or any such Affiliate.

1.2 <u>Additional Services</u>. Seller may request that Company and/or the Subsidiaries, Representatives and Authorized Third Parties provide additional services required by Seller or its Affiliates which have not been addressed herein and which are reasonably necessary to support the operation of Seller or any of its Affiliates during the Services Period as Seller or any of its Affiliates was operated in all material respects as of the Effective Date. Seller shall request such additional services from Company in writing within thirty (30) calendar days of the Effective Date. Within five (5) Business Days of Company's receipt of Seller's written request for such additional services, the Service Coordinators (as defined below) shall commence negotiations, in good faith, with respect to the scope, duration and price (which may be a market price) of such additional services to be provided during the Services Period. Within five (5) Business Days of agreement on such items, the parties shall work in good faith to set forth the additional agreed-upon services in a new services description, in a format similar to that set forth in Exhibit A. Upon the mutual written agreement to such new services description, the additional agreed -upon services shall be deemed "Services" under this Agreement, and such new schedule shall be deemed incorporated into Exhibit A and shall in all other respects be subject to the terms and conditions of this Agreement. Notwithstanding the foregoing, Company has no obligation to agree to provide any service pursuant to this Section 1.2 to the extent such service (a)(i) had not been provided by Company, the Subsidiaries or any Representative or Authorized Third Party to Seller or any of its Affiliates as of the Effective Date ("<u>Existing Services</u>") and (ii) is not related to the continuation of the Existing Services; or (b) does not require any historical or institutional knowledge or unique capabilities on the part of Company or the Subsidiaries, Representatives or Authorized Third Parties.

1.3 Service Schedules. The parties acknowledge and agree that the descriptions of Services in Exhibit A are general only and do not purport to contain an exhaustive description of the Services to be provided. The parties further acknowledge and agree that it may not be practicable to describe each and every aspect of a particular Service in detail; therefore, each Service that is generally agreed upon by the parties and included in Exhibit A will be provided in accordance with the applicable terms of this Agreement, consistent with the past practices of the parties even where all aspects regarding the provision of a particular Service is not described in detail.

1.4 <u>Service Coordinators</u>. Each of Seller and Company shall nominate a representative to act as the primary contact person with respect to the provision of the Services (each such person, a "<u>Service Coordinator</u>"). The Service Coordinators shall be managerial-level employees of Company and Seller, as applicable. The initial Service Coordinators shall be Toyoki Furuta for Seller and a designee to be provided promptly after the Effective Date for Company, and each of their respective phone numbers, facsimile numbers and email addresses shall be set forth on an update made promptly after the Effective Date to Schedule 1 attached hereto. Each of Seller and Company may, in its sole discretion, change its Service Coordinator from time to time by providing written notice to the other party of such change and the relevant contact information for the new Service Coordinator at least three (3) Business Days prior to such change taking effect. Unless Seller and Company otherwise agree in writing, all communications relating to this Agreement or to the Services shall be directed to the Service Coordinators in accordance with Section 5.3 hereof. At Seller's Service Coordinator request, Company's Service Coordinator shall provide Seller's Service Coordinator with an estimate of Service Fees and Expenses for the Services.

1.5 <u>Insurance</u>. During the Services Period, Company shall maintain adequate\_insurance for the conduct of the Services in form and coverage consistent with Company's current coverage as of the Effective Date.

1.6 <u>Subcontractors</u>. Company may subcontract any of its obligations under this Agreement to third-party service providers which have been approved by Seller in writing ("<u>Authorized Third Parties</u>"), which approval shall not be unreasonably withheld or delayed. The Authorized Third Parties include those parties so identified in Exhibit A.

1.7 <u>Consents</u>. Company shall use commercially reasonably efforts to obtain all\_material licenses, approvals and consents of any third party required by Company to provide the Services (collectively "<u>Required Consents</u>"). If notwithstanding such commercial reasonable efforts, Company is unable to provide any Service because of a failure to obtain such material licenses, approvals or consents, or Company reasonably believes that performance of the Service would infringe or misappropriate a third party's intellectual property rights, the parties shall cooperate to determine the best alternative approach.

# 1.8 Intellectual Property.

(a) <u>Developed IP</u>. Subject to Seller's compliance with the terms and conditions of this Agreement, including payment of all Service Fees and Expenses, Company hereby assigns and agrees to assign all of its right, title and interest in and to the Developed IP to Seller. Seller shall have the sole right to apply for, file, register, or otherwise seek Intellectual Property Rights with respect to Developed IP, including the right to seek Patent protection for inventions that constitute Developed IP, if any. Company will provide (and will use commercially reasonable efforts to cause any inventors of Developed IP in Company's or the Subsidiaries' employ to provide) reasonable information and assistance, at Seller's cost and expense, to effect the assignment of rights pursuant to this Section 1.8(a) . With respect to any Developed Technology, Company reserves and Seller hereby grants and agrees to grant to Company and the Subsidiaries, under the Developed Intellectual Property Rights, a worldwide, non-exclusive, perpetual and irrevocable license to use such Developed Technology in the ordinary course of its business. With respect to any Patents within the Developed Intellectual Property Rights, Company reserves and Seller hereby grants and agrees to grant to Company and the Subsidiaries a worldwide, non-exclusive, perpetual and irrevocable license to make, have made, use, sell, offer for sale, and import any article of manufacture or composition of matter and practice any method or process.

(b) <u>Background IP</u>. To the extent, if any, that any Background Technology is embodied in any Deliverables provided to Seller under this Agreement, subject to Seller's compliance with the terms and conditions of this Agreement, including payment of all Service Fees and Expenses, Company hereby grants and agrees to grant to Seller, under Company's Background Intellectual Property Rights, a worldwide, non-exclusive, perpetual, and irrevocable license to use such Background Technology, solely as embodied in such Deliverables, in the ordinary course of Seller's business.

(c) <u>No Other Rights</u>. Except as expressly set forth in this Agreement, neither party grants any rights in or to its Technology or Intellectual Property Rights pursuant to this Agreement. As between the parties, each party shall be solely responsible to prepare, file, prosecute, maintain, and enforce its Intellectual Property Rights in its discretion and at its own cost. Except as expressly set forth in this Agreement, there shall be no right, license, authority, covenant not to sue, immunity from suit, or other defense, whether by implication, by reason of exhaustion, estoppel, or otherwise pursuant to or as a result of this Agreement or the activities of the parties under this Agreement.

(d) Company will use commercially reasonable efforts to identify to Seller any Patents owned by Company and claiming Background Technology Company expects to be

embodied in Deliverables to be delivered to Seller in connection with the provision of the Services under which Company does not have the right to grant to company a license of the scope set forth in Section 1.8(b) without incurring an obligation to pay a royalty or other consideration to a third party.

(e) As used in this Section 1.8:

(i) "<u>Background Intellectual Property Rights</u>" means Intellectual Property Rights, if any, owned by Company that are (a) embodied in Background Technology, and (b) not Developed Intellectual Property Rights; <u>provided</u>, <u>however</u>, that a Patent will be a Background Intellectual Property Right only if Company has the right to grant to Seller a license of the scope set forth in Section 1.8(b)) under such Patent without incurring any obligation to pay any royalty or other consideration to any third party.

(ii) "<u>Background Technology</u>" means Technology, if any, owned by Company or the Subsidiaries that is (A) embodied in any Deliverable, and (B) not Developed Technology.

Agreement.

(iii) "Deliverables" means the deliverables, if any, to be provided by Company as part of the Services performed pursuant to this

(iv) "<u>Developed Intellectual Property Rights</u>" means Intellectual Property Rights to the extent such Intellectual Property Rights (A) are first created by Company in the course of its performance of the Services pursuant to this Agreement and within the scope and during the term of this Agreement, and (B) are embodied in Developed Technology. For the avoidance of doubt, "Developed Intellectual Property Rights" includes the right to seek Patent protection for inventions that constitute Developed Intellectual Property, if any, but does not include any Patents or Patent applications of Company or the Subsidiaries.

(v) "Developed IP" means, collectively, Developed Technology and Developed Intellectual Property Rights.

(vi) "<u>Developed Technology</u>" means Technology embodied in any Deliverable to the extent such Technology is first developed or created by Company in the course of its performance of the Services pursuant to this Agreement and within the scope and during the term of this Agreement.

(vii) "<u>Intellectual Property Rights</u>" means Patents, copyrights, and rights with respect to trade secrets, whether arising under the laws of the United States, Japan or any other jurisdiction, including, in each case, any rights apply for, register, and enforce any of the foregoing. Notwithstanding the foregoing, "Intellectual Property Rights" does not include any trademark rights or similar rights with respect to indicia of source or origin.

(viii) "Patents " means all classes and types of patents, including utility patents, utility models, design patents, invention certificates, including divisionals, continuations, continuations-in-part, reexaminations, reissues, extensions and renewals, in all jurisdictions of the world.

(ix) "<u>Software or Firmware</u>" means a set of instructions, that either (i) directly provides instructions to the computer hardware, or, (ii) indirectly serves as an input to another piece of software.

(x) "<u>Technology</u>" means inventions, know-how, designs, specifications, Software or Firmware and other copyrightable material, technical information, devices, and other developments and technology.

1.9 <u>Use of Services</u>. Company shall be required to provide, or cause the Subsidiaries, Representatives and Authorized Third Parties to provide, Services in Japan (and such other locations specified on Exhibit A attached hereto) only to Seller and its Affiliates, and only in connection with the operation of Seller and its Affiliates in existence as of the Effective Date. Seller shall not, and shall not permit any of its Affiliates or any third parties under its control to, resell any Services to any Person whatsoever or permit the use of the Services by any Person other than in connection therewith.

1.10 Interruption of Services. Subject to Section 5.14, Company may cease or suspend providing, or have the Subsidiaries, Representatives or Authorized Third Parties cease or suspend providing, as applicable, the Services to Seller and its Affiliates if and to the extent such cessation or suspension is (i) required by applicable Law, (ii) necessary due to regularly scheduled maintenance, alterations, repairs or replacements with respect to the applicable Services or the facilities used to provide such Services, or (iv) necessary due to the temporary shutdown of the operation of the facilities providing any Service whenever Company determines such action is necessary in the exercise of its reasonable judgment.

1.11 <u>Cooperation</u>. Seller shall cooperate, and shall cause its Affiliates to\_cooperate, with Company, the Subsidiaries, Representatives and Authorized Third Parties, and provide such Persons with such information and assistance as such Persons may reasonably require to enable them to provide the Services. Seller shall allow, and shall cause its Affiliates to allow, such Persons and their respective employees, agents and sub-contractors reasonable access to its facilities as necessary for the performance of the Services.

1.12 <u>Right to Change or Eliminate Services</u>. The parties acknowledge that Company and its Affiliates may make changes from time to time in the manner in which the Services are performed, including without limitation, the elimination of Services, if Company (i) makes similar changes in the manner in which similar services are performed for its Affiliates, or eliminates for its Affiliates similar Services, and (ii) furnishes to Seller substantially the same notice (in consent and timing) as Company furnishes to its own Affiliates respecting such changes and/or eliminations.

### 1.13 <u>Consultation Period</u>.

(a) The Service Coordinators shall collectively review, as promptly as reasonably practicable after the Effective Date, the Services listed on Exhibit A to determine if they properly reflect the Existing Services in Japan (and such other locations specified on Exhibit

A attached hereto) that are reasonably necessary, during the Services Period, to support the operation of Seller or its Affiliates as Seller or such Affiliate was operated in all material respects as of the Effective Date ("<u>Needed Services</u>"). In connection with such review, Seller shall have the right to amend Exhibit A to add any Needed Services to Exhibit A at any time within the forty-five (45) day period after the Effective Date (such period, the "<u>Consultation Period</u>"); and

(b) Seller shall not have a right to update Exhibit A with respect to adding any services relating to intellectual property matters.

(b) During the Consultation Period, the Service Coordinators shall discuss in good faith details with respect to the process of invoicing by Company and the Subsidiaries for the Services pursuant to Section 4.2, including determining whether a monthly billing cycle is appropriate for the invoicing of specific Services.

1.14 <u>Excluded Services</u>. Notwithstanding anything herein to the contrary, the following shall not be included as Services pursuant to this Agreement: Any services relating to intellectual property matters other than those listed on Exhibit A as of the Effective Date.

# ARTICLE II

# QUALITY OF SERVICES; LIMITATION OF LIABILITY; IMPROVEMENTS

- 2.1 <u>Quality of Services</u>.
  - (a) Company shall perform the Services, or to cause the Services to be

performed, in a workmanlike manner at the same general level of service, with the same degree of care (which in no event may be less than reasonable care), and in a manner similar in all material respects to the manner in which such Services have been provided by Company and the Subsidiaries, Representatives and Authorized Third Parties to Seller and its Affiliates as of the Effective Date.

(b) In addition to the above and to the extent permitted by applicable Law, Company agrees to pass through to Seller and its Affiliates any warranties provided by Authorized Third Parties providing the Services.

(c) In the event of an alleged breach, default or nonperformance of any obligation under this Agreement by Seller or Company, the other party shall provide prompt written notice to the breaching party setting forth in reasonable detail the nature and extent of the alleged breach, default or nonperformance. The breaching party will then have a period of ten (10) Business Days in which to initiate actions reasonably designed to cure such alleged breach, default or nonperformance, and all such deficiencies shall, in any case, be cured within thirty (30) days following receipt by the breaching party of notice thereof.

2.2 <u>Specific Performance</u>. Company acknowledges that the rights of Seller to\_enforce the covenants and agreements made in this Agreement are special, unique, and of extraordinary character, and that, in the event Company violates or fails or refuses to perform any covenant or agreement made by it herein, Seller would be irreparably damaged and be without adequate remedy at law. Company agrees, therefore, that, in the event it fails or refuses to perform, or otherwise violates, any covenant or agreement made by it herein, Seller shall, in

addition to any remedies available at law, be entitled to seek specific performance of such covenant(s) or agreement(s) and any other equitable remedy. For the avoidance of doubt, the foregoing shall not apply to the extent Company, the Subsidiaries, Representatives or Authorized Third Parties are expressly permitted to cease or suspend the provision of Services pursuant to Section 1.10 or otherwise not comply with other obligations hereunder pursuant to Section 5.14.

2.3 Limitation of Liability. IN NO EVENT SHALL COMPANY, THE SUBSIDIARIES NOR ANY OF ITS OR THEIR RESPECTIVE DIRECTORS, OFFICERS, EMPLOYEES OR AGENTS (INCLUDING COMPANY'S REPRESENTATIVES) BE LIABLE TO SELLER OR ANY OF ITS AFFILIATES FOR ANY INDIRECT, SPECIAL, INCIDENTAL, PUNITIVE OR CONSEQUENTIAL DAMAGES OR FOR ANY LOSS OF PROFITS, LOSS OF REVENUE, LOSS RESULTING FROM INTERRUPTION OF BUSINESS OR LOSS OF DATA ARISING UNDER OR RELATING TO THIS AGREEMENT, WHETHER BASED ON BREACH OF CONTRACT, TORT (INCLUDING NEGLIGENCE), OR OTHERWISE, AND WHETHER OR NOT COMPANY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGE. EXCEPT IN THE CASE OF COMPANY'S GROSS NEGLIGENCE, FRAUD OR WILLFUL MISCONDUCT, IN NO EVENT SHALL THE TOTAL LIABILITY OF COMPANY, THE SUBSIDIARIES AND ANY OF ITS OR THEIR DIRECTORS, OFFICERS, EMPLOYEES AND AGENTS (INCLUDING COMPANY'S REPRESENTATIVES) ARISING OUT OF OR RELATING IN ANY WAY TO THIS AGREEMENT EXCEED THE FEES PAID BY SELLER TO COMPANY HEREUNDER.

2.4 <u>No Other Warranties</u>. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY STATED HEREIN, COMPANY DISCLAIMS ALL EXPRESS AND IMPLIED REPRESENTATIONS AND WARRANTIES IN CONNECTION WITH THE SERVICES, INCLUDING ANY IMPLIED WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE OR NON-INFRINGEMENT.

2.5 Indemnification. Company shall indemnify, hold harmless and reimburse\_Seller and its Affiliates for all Losses based upon, attributable to, arising out of or resulting from Company's gross negligence, fraud or willful misconduct in connection with the Services. Notwithstanding anything herein to the contrary, neither Company, any of the Subsidiaries, nor any Representative shall be liable or held accountable, in damages or otherwise, for any error in judgment or any mistake of fact or Law or for anything which Company does or refrains from doing, other than for Company's gross negligence, fraud or willful misconduct.

2.6 <u>Mitigation</u>. Seller has a duty to mitigate (and cause its Affiliates to\_mitigate) the Losses that would otherwise be recoverable from Company pursuant to this Agreement by taking appropriate and reasonable actions to reduce or limit the amount of any such Losses.

# ARTICLE III

### TERM AND TERMINATION OF THE SERVICES

3.1 <u>Term</u>. Unless earlier terminated pursuant to Section 3.2, with respect to each of the Services (or any portion thereof), the term of this Agreement as it relates thereto will

be for a period beginning on the Effective Date and continuing until the termination by Seller of all the Services to be provided by Company under this Agreement pursuant to Section 3.2(a); <u>provided</u>, <u>however</u>, that, if any Service (or portion thereof) is designated to be provided for a fixed period or until a certain date as set forth in Exhibit A attached hereto, then this Agreement shall be automatically deemed terminated with respect to such Service (or portion thereof) upon completion of such period or on such date.

# 3.2 <u>Termination</u>.

(a) Any of the Services, or any portion thereof, may be terminated by Seller, in its sole discretion, at any time by furnishing forty-five (45) days' prior written notice to Company of Seller's intention to terminate the applicable Service, which written notice shall specify (i) the Service (or portion thereof) being terminated and (ii) the date on which the Service (or portion thereof) shall be terminated; provided, however, that Seller shall be responsible for the payment of any and all Service Fees and Expenses (each as defined below) accrued or incurred for such Service under this Agreement prior to the later of (A) the effective date of the termination and, (B) in the event that Company is contractually or legally required to incur Expenses related to such Service beyond the effective date of the termination, the date that Company is no longer contractually or legally required to incur such Expenses.

(b) Either Seller or Company, with respect to (i) and (ii) below, and Company, with respect to (iii) below, may immediately terminate this Agreement by written notice to the other party upon the occurrence of any of the following events:

(i) the other party (A) enters into proceedings in bankruptcy or insolvency, (B) makes an assignment for the benefit of creditors, (C) files or has filed against it any petition under a bankruptcy law or any other law for relief as a debtor (or similar law in purpose or effect) or (D) enters into liquidation or dissolution proceedings;

(ii) the other party materially breaches any of its obligations hereunder and the breach remains uncured for the applicable period specified in Section 2.1(c); or

(iii) any amount due under this Agreement remains unpaid by Seller for a period of more than fifteen (15) days following Seller's receipt of a notice of delinquency.

3.3 <u>Survival of Certain Obligations</u>. Without prejudice to the survival of other agreements of the parties, the right of Company to receive the applicable payments for expenses for the Services rendered prior to the effective date(s) of termination of such Services under this Agreement shall survive the termination or expiration, in whole or in part, of this Agreement. In addition, Section 1.8, Section 2.3, Section 2.4, Section 2.5, Section 2.6, Section 3.3, Section 4.1, Section 4.2, and Article V shall survive the termination or expiration of this Agreement.

### ARTICLE IV

#### CONSIDERATION

#### 4.1 <u>Consideration</u>.

(a) The fees charged by Company for Services hereunder ("<u>Service Fees</u>") shall be equal to Company's and, as applicable, the Subsidiaries' fully allocated cost for\_such Services, including (a) all compensation, benefit and other costs and expenses incurred by or with respect to employees directly engaged in providing such Services, including (i) in respect of compensation, all applicable bonus compensation, (ii) in respect of benefits, all benefits under Plans, and (iii) in respect of costs and expenses, all costs of materials and for such items as travel incurred in respect of the Services, as well as a reasonable allocation for space, maintenance, and facilities costs allocable to employees engaged in providing the Services, as well as the actual cost of any third party services used in providing the Services, and (b) similar costs with respect to those directly engaged in the supervision of such Services. Without limitation or modification of the foregoing, the cost-allocation memoranda attached hereto as Exhibit B memorialize how the fully allocated costs for the Services described in Items Nos. 2, 4, 5, 7 and 25 set forth on Exhibit A attached hereto are calculated as of the Effective Date. For clarity, each attachment comprising Exhibit B attached hereto references the Item No. set forth on Exhibit A attached hereto to which such cost allocation memorandum corresponds.

(b) In addition to the Service Fees, Seller will reimburse Company for all reasonable, documented, out-of-pocket expenses incurred by Company or the relevant Subsidiary, Representative and/or Authorized Third Party in connection with the provision of the Services ("<u>Expenses</u>").

### 4.2 <u>Invoicing</u>.

(a) Company and/or the relevant Subsidiary will invoice Seller for such expenses monthly in arrears for the Services provided under this Agreement. Each such invoice shall include (i) a brief description of the Service provided by Company, the Subsidiaries, Representatives and Authorized Third Parties during that month, and the Service Fees for such Services Fees and (ii) the amounts of Expenses incurred by Company, the Subsidiaries, Representatives and Authorized Third Parties during that month and reasonable documentation of such expenses. Seller shall pay all amounts due under each invoice (in the currency denominated by Company in such invoice) within forty-five (45) days following receipt of such invoice (the "<u>Due Date</u>") without offset, withholding or deduction of any kind. Seller shall be responsible for the payment of all Taxes payable with respect to the performance, receipt or consumption of the Services or the execution and delivery of this Agreement, other than any Tax based upon the net income of Company or any other Person providing any of the Services.

(b) Interest shall accrue on any unpaid balance at a rate of 1% per month for the period commencing on the Due Date and ending on the date payment is received in full by Company, and Seller shall bear all reasonable costs and expenses (including attorney's fees and court costs) incurred by Company or the Subsidiaries in collecting outstanding balances from Seller not paid on the Due Date.

### ARTICLE V

### MISCELLANEOUS

#### 5.1 <u>Confidentiality</u>.

(a) Each party (the "<u>Receiving Party</u>") agrees that, from the Effective Date until the fifth anniversary of the Effective Date, it shall, and shall cause its Affiliates and Representatives to:

(i) take proper and all reasonable measures to ensure the confidentiality of all Confidential Information (as defined below) of the other party (the "<u>Disclosing Party</u>"), including keeping it separate from information belonging to the Receiving Party;

(ii) use such Confidential Information only for the Proper Use (as defined below);

(iii) permit access to such Confidential Information only to such of its Representatives having a need to know such Confidential Information ("<u>Permitted Disclosees</u>"), <u>provided</u>, that Receiving Party shall cause or have caused its Permitted Disclosees to be bound by and comply with the confidentiality no less restrictive than hereunder by written agreements, and inform each of those Permitted Disclosees of the confidential nature of such Confidential Information and of the obligations on Receiving Party in respect thereof, and Receiving Party shall be responsible for any breach of this Section 5.1 by any of its Permitted Disclosees;

- Proper Use;
- (iv) make copies of the Confidential Information of the Disclosing Party only to the extent that the same are strictly required for the

(v) treat all Confidential Information of the Disclosing Party with the degree of care to avoid disclosure to any third party as is used with respect to Receiving Party's own information of like importance which is to be kept confidential; and

(vi) promptly return all Confidential Information of the Disclosing Party to the Disclosing Party upon its written request or (at the Disclosing Party's option) destroy all such Confidential Information and provide to the Disclosing Party a certificate of such destruction signed by a duly authorized officer of the Receiving Party.

(b) Where any Confidential Information of the Disclosing Party is the subject of any security regulations of any Governmental Entity, Receiving Party shall, and hereby undertakes to, take such measures as may be required by such regulations to protect such Confidential Information. Without prejudice to any obligations imposed on and assumed by the Receiving Party under any security regulations of any Governmental Entity, the obligations of confidentiality herein shall not apply to any Information which the Receiving Party by its written records can show: (i) was in the possession of the Receiving Party before such Information was imparted or disclosed by the Disclosing Party;

the Information;

- (ii) is independently developed by any servant, agent or employee of the Receiving Party without access to or use or knowledge of
- (iii) is in or subsequently comes into the public domain other than by breach by the Receiving Party of its obligations hereunder;

(iv) is received by the Receiving Party without restriction on disclosure or use from a third party which the Receiving Party reasonably and honestly believes is entitled to make such disclosure; or

(v) is approved for release by the written agreement of the Disclosing Party.

The Receiving Party may disclose Confidential Information of the Disclosing Party if required to be disclosed by applicable Law; provided that, if the Receiving Party is to make such disclosure, it shall give the Disclosing Party as much prior notice thereof as is reasonably practicable so that the Disclosing Party may seek such protective orders or other confidentiality protection as the Disclosing Party, in its sole discretion, may elect, and the Receiving Party shall reasonably co-operate with the Disclosing Party in protecting the confidential or proprietary nature of such Confidential Information which is to be so disclosed.

(c) As used in this Section 5.1:

(i) "<u>Confidential Information</u>" shall mean: (A) in respect of Information provided in documentary form or by way of a model or in other tangible or intangible form, Information which at the time of disclosure to the Receiving Party is marked, or otherwise designated, to show expressly or by implication that it is imparted or disclosed in confidence; (B) Information the nature of which, or the circumstances in which it was supplied, implies that it should be treated as confidential notwithstanding the absence of any mark or designation of confidentiality; (C) in respect of Information that is imparted or disclosed or ally or by demonstration or presentation, any Information that the Receiving Party has been expressly informed by the Disclosing Party at the time of disclosure to have been imparted or disclosed in confidence; (D) in respect of Information imparted or disclosed orally or by demonstration or presentation, any note or record of the disclosure; and (E) any copy of any of the foregoing.

(ii) "<u>Information</u>" shall mean (A) with respect to that disclosed by Company, information relating to the Services provided pursuant to this Agreement, by or on behalf of Company, to Seller or any of its Affiliates, in oral or documentary form or by way of models or other tangible or intangible form or by demonstrations or presentations; and (B) with respect to that disclosed by Seller or any of its Affiliates, information relating to Seller's or any of its Affiliates' utilization or receipt of Services provided pursuant to this Agreement, by or on behalf of Seller or any of its Affiliates, to Company, in oral or documentary form or by way of models or other tangible or intangible form or by demonstrations, including all Seller and Affiliate information accessed in connection with the provision of the Services, whether in electronic or other form.

(iii) "<u>Proper Use</u>" shall mean (A) with respect to Seller and its Affiliates, the use of Company's Confidential Information (1) wholly necessarily and exclusively for the purpose of conducting the business of Seller and its Affiliates in connection with the Services; (2) in connection with the enforcement of Seller's rights hereunder; and (3) in connection with the defense by Seller of any claim asserted against Seller hereunder; and (B) with respect to Company, the use of Seller's and its Affiliates' Confidential Information (1) wholly necessarily and exclusively for the purpose of providing or the causing the provision of, the Services; (2) in connection with the enforcement of Company's rights hereunder; and (3) in connection with the defense by Company of any claim asserted against Company hereunder.

# 5.2 Entire Agreement; Amendments and Waivers.

(a) This Agreement and the Purchase Agreement contain the entire agreement of the parties with respect to the subject matter hereof and supersede all prior agreements, written or oral, with respect to their subject matter. The exhibits and schedules to this Agreement are hereby incorporated and made a part hereof and are an integral part of this Agreement.

(b) Any provision of this Agreement, including all exhibits hereto, may be amended or waived, but only if, such amendment or waiver is in writing and is signed, in the case of an amendment, by the parties to this Agreement or, in the case of a waiver, by the party against whom the waiver is to be effective.

(c) No failure or delay by either party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by applicable Law.

5.3 <u>Notices</u>. All notices, requests and other communications to either party hereunder shall be in writing (including facsimile transmission) and shall be given,

if to Company, to:

Viviti Technologies Ltd. c/o Hitachi Global Storage Technologies, Inc. 3403 Yerba Buena Road San Jose, CA 95135 Attention: Christopher Dewees Facsimile: (408) 717-9063 E-mail: Christopher.Dewees@hitachigst.com with a copy (which shall not constitute notice) to:

O'Melveny & Myers LLP 610 Newport Center Drive, Suite 1700 Newport Beach, California 92660 Attention: J. Jay Herron, Esq. Facsimile: (949) 823-6994 E-mail: jherron@omm.com

if to Seller, to:

Hitachi, Ltd., Business Development Office 6-6 Marunouchi 1-chome Chiyoda-ku Tokyo 100-8280, Japan Attention: General Manager Phone: +81-3-4564-5483 Fax: +81-3-4564-6260

with a copy (which shall not constitute notice) to:

Morrison & Foerster LLP Shin-Marunouchi Building, 29th Floor 5-1, Marunouchi 1-chome Chiyoda-ku, Tokyo 100-6529 Japan Attention: Kenneth A. Siegel, Esq. Facsimile: 011-81-3-3214-6512 E-mail: KSiegel@mofo.com

or to such other address or facsimile number as such party may hereafter specify for the purpose by notice to the other party hereto. All such notices, requests and other communications shall be deemed received on the date of receipt by the recipient thereof if received prior to 5:00 p.m. on a Business Day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed to have been received on the next succeeding Business Day in the place of receipt.

5.4 <u>Disputes</u>. In the event of any controversy or dispute arising out of or<u>relating to this Agreement</u>, the Service Coordinators shall in good faith attempt to resolve such dispute. If after twenty (20) days the parties have not reached an agreement with respect to such dispute, either party may file a claim against the other party pursuant to Section 5.5 below.

5.5 <u>Governing Law; Negotiation Procedure; Service of Process; Consent to Jurisdiction; Venue; Waiver of Jury Trial.</u>

(a) <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without regard to its conflicts of law rules.

#### (b) <u>Negotiation Procedure; Service of Process</u>.

(i) The parties intend that all disputes between the parties arising out of this Agreement shall be settled by the parties amicably through good faith discussions upon the written request of either party.

(ii) Prior to filing suit, instituting a Proceeding or seeking other judicial or governmental resolution in connection with any dispute between the parties or any of their subsidiaries arising out of this Agreement or any of the transactions contemplated hereby, the parties will attempt to resolve such dispute by good faith negotiations. Such negotiations shall proceed as follows:

(A) Either party may send a written notice to another party requesting such negotiations. Promptly following receipt of such notice by the receiving party, each party shall cause the individual designated by it as having general responsibility for this Agreement to meet in person with the individual so designated by the other party to discuss the dispute.

(B) If the dispute is not resolved within thirty (30) days after the first meeting between such individuals (or if earlier within forty five (45) days of the notice referred to in clause (i) above), then, upon the written request of either party, each party shall cause the individual designated by it as having general responsibility for the overall relationship defined by this Agreement to meet in person with the individual so designated by the other party to discuss the dispute.

(C) If the dispute is not resolved within fifteen (15) days after the first meeting between such individuals (or if earlier within thirty (30) days of the notice referred to in clause (ii) above), then, upon the written request of either party, Company shall nominate one corporate officer of the rank of senior vice president or higher, and Seller shall nominate one corporate officer of the rank of Board Director or higher, which corporate officers shall meet in person and attempt in good faith to negotiate a resolution to the dispute.

(iii) Except and only to the limited extent provided in Section 5.5(b)(iv), neither party shall file suit, institute a Proceeding or seek other judicial or governmental resolution of the dispute until at thirty (30) days after the first meeting between the corporate officers described in clause (iii) above (or if earlier forty five (45) days after the notice referred to in such clause (iii)) but after the expiration of such periods, either party may file suit, institute a Proceeding or seek other judicial or governmental resolution. For purposes of this Agreement, the procedures set forth in Section 5.5(b)(ii) and this Section 5.5(b)(iii) shall be referred to as the "Negotiation Procedures".

(iv) Notwithstanding the provisions of Sections 5.5(b)(ii) and 5.5(b)(iii), either party may institute a Proceeding at any time seeking a preliminary injunction, temporary restraining order, or other equitable relief, if necessary in the sole judgment of that party to avoid material harm to its property, rights or other interests, before commencing, or at any time during the course of, the dispute procedure described in Sections 5.5(b)(ii) and 5.5(b)(iii). In addition, either party may file an action prior to the commencement of or at any time during or after the dispute resolution procedures in Sections 5.5(b)(ii) and 5.5(b)(iii) if in the sole judgment of that party it is necessary to prevent the expiration of a statute of limitations or filing period or the loss of any other substantive or procedural right.

(c) <u>Consent to Jurisdiction; Venue</u>. Each of the parties irrevocably and unconditionally submits to the exclusive jurisdiction of the Delaware Court of Chancery (and if jurisdiction in the Delaware Court of Chancery shall be unavailable, any Delaware State court and the Federal court of the United States of America sitting in the State of Delaware) for the purposes of any Action or other proceeding arising out of this Agreement or any transaction contemplated hereby (and agrees that no such Action or proceeding relating to this Agreement shall be brought by it or any of its Affiliates except in such courts). Each of the parties further agrees that, to the fullest extent permitted by applicable Law, service of any process, summons, notice or document by U.S. registered mail to such person's respective address set forth in Section 5.3 above shall be effective service of process for any Action or proceeding in the State of Delaware with respect to any matters to which it has submitted to jurisdiction as set forth above in the immediately preceding sentence. Each of the parties irrevocably and unconditionally waives (and agrees not to plead or claim), any objection to the laying of venue of any Action or proceeding arising out of this Agreement or the transactions contemplated hereby in the Delaware Court of Chancery (and if the Delaware Court of Chancery shall be unavailable, in any Delaware State court or the Federal court of the United States of America sitting in the State of Delaware) or that any such Action or proceeding brought in an inconvenient forum.

# (d) <u>Waiver of Jury Trial</u>. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT.

5.6 <u>Books and Records; Inspection</u>. Seller shall, and shall cause its Affiliates to, make available on a timely basis to Company and the Subsidiaries, Representatives and Authorized Third Parties such information and materials reasonably requested by Company to enable such Persons to provide the Services. Seller shall, and shall cause its Affiliates to, provide to such Persons reasonable access to the premises of Seller and its Affiliates, to the extent necessary for the purpose of providing the Services. During the Services Period and for a period of three (3) years following the Services Period or, if applicable Law requires a longer period, such longer period, Company shall maintain a complete and accurate set of files, books and records of all business activities and operations conducted by Company related to the Services provided under the terms of this Agreement, as well as any correspondence related to compliance with any applicable national, state and local laws, rules and regulations. Company will provide Seller, subject to Section 5.1 hereof, such information as Seller may reasonably

request from Company's books and records to the extent relating to the provision of any Service hereunder.

5.7 <u>Severability</u>. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other Governmental Entity to be invalid, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to either party. Upon such a determination, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

# 5.8 <u>Binding Effect; Benefit; Assignment</u>.

(a) The provisions of this Agreement shall be binding upon and shall inure to the benefit of the parties hereto and their respective successors and assigns. No provision of this Agreement is intended to confer any rights, benefits, remedies, obligations or liabilities hereunder upon any Person other than the parties hereto and their respective successors and assigns.

(b) No party may assign, delegate or otherwise transfer any of its rights or obligations under this Agreement without the consent of each other party hereto, except that Seller may transfer or assign its rights and obligations under this Agreement, in whole or from time to time in part, to one or more of its Affiliates at any time; provided that such transfer or assignment shall not relieve Seller of any of its obligations hereunder.

5.9 <u>Definitional and Interpretive Provisions</u>. Section 1.2 of the Purchase Agreement\_shall also apply to this Agreement; <u>provided</u>, <u>however</u>, that for the purposes of this Agreement, "Business Days" shall mean a day, other than Saturday, Sunday or a public holiday in the country in which the applicable Service is performed.

5.10 <u>Counterparts</u>. This Agreement may be signed in any number of counterparts, each of which shall be an original, with the same effect as if the signatures thereto and hereto were upon the same instrument. This Agreement shall become effective when each party hereto shall have received a counterpart hereof signed by the other party hereto. Until and unless each party has received a counterpart hereof signed by the other party hereto, this Agreement shall have no effect and no party shall have any right or obligation hereunder (whether by virtue of any other oral or written agreement or other communication).

5.11 <u>No Third Party Beneficiaries</u>. This Agreement is intended and agreed to be solely for the benefit of the parties, and no third party, other than their respective Affiliates, shall accrue any benefit, claim or right of any kind whatsoever pursuant to, under, by or through this Agreement.

5.12 <u>Relationship of the Parties</u>. It is expressly understood and agreed that in rendering the Services hereunder, each of the parties is acting as an independent contractor and that this Agreement does not make the providing party an employee, agent or other

representative of the other party for any purpose whatsoever. Neither party has the right or authority to enter into any contract, warranty, guarantee or other undertaking in the name or for the account of the other party, or to assume or create any obligation or liability of any kind, express or implied, on behalf of the other party, or to bind the other party in any manner whatsoever, or to hold itself out as having any right, power or authority to create any such obligation or liability on behalf of the other party or to bind the other party in any manner whatsoever (except as to any actions taken by a party at the express written request and direction of the other party). No employee, contractor or subcontractor of either party shall be deemed to be an employee, contractor or subcontractor of the other party, it being fully understood and agreed that no employee of either party is entitled to benefits or compensation from the other party. Each party is wholly responsible for withholding and payment of all applicable national, state and local and other payroll taxes with respect to its own employees, including any contributions from them as required by Law.

5.13 <u>Conflict</u>. In case of conflict between the terms and conditions of this Agreement\_and any exhibit or schedule hereto, the terms and conditions of such exhibit or schedule shall control and govern, insofar as such terms and conditions in the exhibit or schedule relate to the Service that is the subject of such conflict. In the event of any conflict between the terms of the Purchase Agreement, on the one hand, and this Agreement and each exhibit or schedule hereto, on the other hand, the terms of this Agreement shall control and govern.

5.14 <u>Force Majeure</u>. Each party shall be excused from its obligations under this <u>Agreement</u>, other than payment obligations, to the extent that any delay or failure in the performance of such obligations is a result of any cause beyond its reasonable control (and without the fault of such party), including, acts of God, acts of civil or military authority, embargoes, epidemics, war, terrorism, riots, insurrections, fires, explosions, earthquakes, floods, severe weather conditions or changes in Law.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed by their respective authorized officers as of the day and year first above written.

Hitachi, Ltd.

By: \_\_\_

Name: Title:

Viviti Technologies Ltd.

By: \_\_\_\_

Name: Title:

SIGNATURE PAGE TO THE REVERSE SERVICES AGREEMENT

# Execution Version SIXTH AMENDMENT TO STOCK PURCHASE AGREEMENT

This Sixth Amendment to the Stock Purchase Agreement (this "<u>Amendment</u>") is made this 6<sup>th</sup> day of March, 2012, by and among Western Digital Corporation, a Delaware corporation (the "<u>Buyer Parent</u>"), Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect wholly owned subsidiary of the Buyer Parent (the "<u>Buyer</u>"), Hitachi, Ltd., a company incorporated under the laws of Japan (the "<u>Seller</u>"), and Viviti Technologies Ltd., a company incorporated under the laws of the Republic of Singapore and a wholly owned subsidiary of the Seller (the "<u>Company</u>", and collectively with the Seller, the Buyer and the Buyer Parent, the "<u>Parties</u>", and each, a "<u>Party</u>"). Capitalized terms used herein and not otherwise defined shall have the meanings set forth in the Stock Purchase Agreement (as defined below).

WHEREAS, the Parties entered into a Stock Purchase Agreement on March 7, 2011, as amended by a First Amendment to Stock Purchase Agreement, dated May 27, 2011, as further amended by a Second Amendment to Stock Purchase Agreement, dated November 23, 2011, as further amended by a Third Amendment to Stock Purchase Agreement, dated January 30, 2012, as further amended by a Fourth Amendment to Stock Purchase Agreement, dated February 15, 2012, and as further amended by a Fifth Amendment to Stock Purchase Agreement, dated as of the date hereof (together, the "<u>Stock Purchase Agreement</u>");

WHEREAS, immediately prior to the execution of this Amendment, the Parties executed the Fifth Amendment to Stock Purchase Agreement; and

WHEREAS, the Parties desire to amend the Stock Purchase Agreement as reflected herein.

NOW, therefore, in exchange for good and valuable consideration, the receipt of which is hereby acknowledged, the Parties hereby agree as follows:

1. **Section 1.1 of the Stock Purchase Agreement.** The definition of Target Working Capital set forth in Section 1.1 (Definitions) of the Stock Purchase Agreement is hereby amended and restated in its entirety to read as follows: "<u>Target Working Capital</u>" means \$0.00 (zero)."

2. Section 2.2 of the Stock Purchase Agreement. A new Section 2.2(h) is hereby added to the Stock Purchase Agreement as follows:

"(h) <u>Assumed Benefit Plan Credit</u>. The Purchase Price shall be increased by \$84,000,000, which the Parties hereby agree represents the net value, as of the Closing, of the liabilities in respect to unfunded Benefit Plans that will be assumed by the Seller (or its Affiliates) as of the Closing."

3. **Section 2.5 of the Stock Purchase Agreement**. Section 2.5 of the Stock Purchase Agreement is hereby deleted in its entirety and replaced by the following:

"Section 2.5 Closing. Unless this Agreement shall have been terminated pursuant to <u>Article X</u>, and subject to the satisfaction or waiver of all of the conditions set forth in

<u>Article VII</u> and <u>Article VIII</u>, the closing of the transactions contemplated by this Agreement (the "<u>Closing</u>") shall take place at the offices of O'Melveny & Myers LLP, 610 Newport Center Drive, 17th Floor, Newport Beach, California, or such other place (including by electronic transmission) as the Buyer and the Seller shall agree, on March 8, 2012, or at such other time and place as shall be mutually agreed by the Parties (the "<u>Closing Date</u>")."

4. **Section 6.12(d)(i)(B) of the Stock Purchase Agreement – Seller Benefit Plans Adjustment**. Section 6.12(d)(i)(B) of the Stock Purchase Agreement is hereby deleted in its entirety and replaced by the following:

"Assumed PBO" means the Projected Benefit Obligation with respect to the Assumed Benefit Plans as determined according to the actuarial methods and assumptions used in determining the Projected Benefit Obligation set forth in the Financial Statements less \$148,000,000."

5. **Section 6.12(d)(v) of the Stock Purchase Agreement – Seller Benefit Plans Adjustment.** Section 6.12(d)(v) of the Stock Purchase Agreement is hereby amended to include the following language immediately prior to the proviso:

"and (D) substituting (1) an amount equal to what the Estimated Unfunded Assumed PBO Amount attributable to the Hitachi Ltd. Corporate Pension Fund would be if the definition of Assumed PBO in <u>Section 6.12(d)(i)(B)</u> did not contain the reduction of \$148,000,000 for (2) \$148,000,000 in <u>Section 6.12(d)(i)(B)</u>;"

6. **Section 10.1(b) of the Stock Purchase Agreement.** Section 10.1(b) of the Stock Purchase Agreement is hereby deleted in its entirety and replaced by the following:

"(b) by any of the Parties at any time after March 15, 2012 (the "<u>Termination Date</u>"), if the Closing shall not have occurred and the Party seeking termination is not in material violation or breach of its respective representations, warranties, covenants or obligations contained in this Agreement,"

7. **Statement of Estimated Working Capital**. The Parties hereby acknowledge and agree that the statement of estimated Working Capital attached as <u>Exhibit A</u> hereto is the "Statement of Estimated Working Capital" as such term is defined in the Stock Purchase Agreement.

8. **Consideration Spreadsheet**. The Parties hereby acknowledge and agree that the Consideration Spreadsheet delivered by the Company to the Buyer Parent pursuant to <u>Section 2.3(a)</u> shall be amended to reflect the amendment to Section 6.12(d)(i)(B) of the Stock Purchase Agreement set forth in item 4. above.

9. **Satisfaction of Certain Conditions under the Stock Purchase Agreement**. The Parties hereby acknowledge and agree that all of the conditions set forth in Section 7.2 and Section 8.2 of the Stock Purchase Agreement have been satisfied.

10. **Effect on the Stock Purchase Agreement.** This Amendment shall not constitute a waiver, amendment or modification of any provision of the Stock Purchase Agreement not

expressly referred to herein. Except as expressly amended or modified herein, the provisions of the Stock Purchase Agreement are and shall remain in full force and effect and are hereby ratified and confirmed. On and after the date hereof, each reference in the Stock Purchase Agreement to "this Agreement", "herein", "hereof", "hereunder" or words of similar import shall mean and be a reference to the Stock Purchase Agreement as amended hereby, although it shall not alter the dates as of which any provision of the Stock Purchase Agreement speaks. For example, phrases such as "as of the date hereof" and "as of the date of this Agreement" shall continue to refer to March 7, 2011, the date that the Stock Purchase Agreement was originally executed. To the extent that a provision of this Amendment conflicts with or differs from a provision of the Stock Purchase Agreement, such provision of this Amendment shall prevail and govern for all purposes and in all respects.

11. Miscellaneous. Sections 12.4, 12.5, 12.7 and 12.13 of the Stock Purchase Agreement are incorporated herein by reference.

[Signature page follows]

#### **"BUYER PARENT"**

# WESTERN DIGITAL CORPORATION

By: <u>/s/</u> Michael Ray

Michael Ray Senior Vice President, General Counsel and Secretary

#### **"BUYER"**

#### WESTERN DIGITAL IRELAND, LTD.

By: /s/ Michael Ray Michael Ray Vice President

#### "SELLER"

# HITACHI, LTD.

By: /s/ Toyoki Furuta Name: Toyoki Furuta

Title: General Manager Business Development Office

# "COMPANY"

# VIVITI TECHNOLOGIES LTD.

By: /s/ Christopher S. Dewees Name: Christopher S. Dewees Title: SVP

Signature Page to Sixth Amendment to Stock Purchase Agreement

### **INVESTOR RIGHTS AGREEMENT**

This Investor Rights Agreement (the "<u>Agreement</u>") is made as of this 8th day of March, 2012, among Western Digital Corporation, a Delaware corporation (the "<u>Company</u>"), and Hitachi, Ltd., a company incorporated under the laws of Japan (the "<u>Investor</u>").

WHEREAS, the Company, Western Digital Ireland, Ltd., a corporation organized under the laws of the Cayman Islands and an indirect wholly owned subsidiary of the Company (the "<u>Buyer</u>"), the Investor, and Viviti Technologies Ltd., a company incorporated under the laws of the Republic of Singapore and a wholly-owned subsidiary of the Investor ("<u>Viviti</u>"), are parties to that certain Stock Purchase Agreement, dated as of March 7, 2011 (as amended, through the date hereof, the "<u>Stock Purchase Agreement</u>"), pursuant to which the Investor will receive from the Buyer on the Closing Date Twenty Five Million (25,000,000) shares of the Company's Common Stock (the "<u>Shares</u>") as a portion of the consideration for the sale of the Investor's stock in Viviti; and

WHEREAS, in connection with the Stock Purchase Agreement and the transfer of the Shares to the Investor, the parties desire to enter into this Agreement in order to establish certain rights and restrictions relating to the Investor's ownership of the Shares.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained and for other good and valid consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

# ARTICLE 1 DEFINITIONS

Section 1.01 **Definitions**. The following terms shall have the following meanings:

"<u>13D Group</u>" means any partnership, limited partnership, syndicate or other group, as those terms are used within the meaning of Section 13(d)(3) of the Exchange Act.

"<u>Action</u>" means any action, suit, proceeding, hearing, order, charge, complaint, claim, arbitration or investigation at Law or in equity, or before any Governmental Entity.

"Affiliate" means any Person now or hereafter controlling, controlled by or under common control with another Person.

"Agreement" has the meaning set forth in the Preamble.

"<u>Beneficial Owner</u>," "<u>Beneficial Ownership</u>," "<u>Beneficially Own</u>" or "<u>Beneficially Owned</u>" shall refer to the concept of "beneficial ownership" in Rule 13d-3 promulgated under the Exchange Act.

"Board" or "Board of Directors" means the Board of Directors of the Company.

"brokers' transaction" has the meaning ascribed to such term under Rule 144(g) under the Securities Act.

"Business Day." means a day, other than Saturday, Sunday or public holidays in the United States of America.

"Buyer" has the meaning set forth in the Recitals.

"<u>Change of Control</u>" means the existence or occurrence of any of the following: (i) the sale, conveyance or disposition by the Company of more than fifty percent (50%) of the Company's assets; (ii) the consolidation, merger or other business combination of the Company with or into any other entity, unless, immediately after such consolidation, merger or other business combination, shareholders of the Company immediately prior to the consummation of the transaction continue to own Equity Securities representing, directly or indirectly, more than fifty percent (50%) of the aggregate voting rights of such new or surviving entity; or (iii) the acquisition by any Person, whether singly or as part of a 13D Group, as a result of one transaction or a series of transactions over time, of Equity Securities representing, directly or indirectly, more than fifty percent (50%) of the aggregate voting rights of the Company. For purposes of this definition, a sale of more than fifty percent (50%) of the Company and its Affiliates or the sale of equity interests of one or more of the Company's Affiliates with an aggregate value in excess of fifty percent (50%) of the aggregate value of the Company and its Affiliates is sold.

"Closing" has the meaning set forth in the Stock Purchase Agreement.

"Closing Date" has the meaning set forth in the Stock Purchase Agreement.

"Common Stock" means the Common Stock of the Company, par value 0.01 United States dollars.

"<u>Company</u>" has the meaning set forth in the Preamble.

"Company Indemnitees" has the meaning set forth in Section 4.06(b).

"<u>Company Supported Distribution</u>" means a public underwritten offering by the Company of Registrable Securities that is designated by the Investor as a "Company Supported Distribution" in the applicable Shelf Take-Down Notice or Demand Notice.

"<u>Competitors</u>" has the meaning set forth in <u>Section 3.03(b)(i)</u>.

"Competitor List Letter" has the meaning set forth in Section 3.03(b)(i).

"Competitor Transferees" has the meaning set forth in Section 3.03(b)(i).

"<u>Demand Notice</u>" has the meaning set forth in <u>Section 4.02(a)</u>.

"Demand Registration" has the meaning set forth in Section 4.02(a).

"Demand Registration Statement" has the meaning set forth in Section 4.02(a).

"Election Meetings" has the meaning set forth in Section 2.01.

"<u>Equity Securities</u>" of the Company means any capital stock or other equity interests of the Company, any securities convertible into, exercisable for or exchangeable for capital stock or other equity interests of the Company, and any other rights, warrants or options to acquire any of the foregoing securities.

"<u>Exchange Act</u>" means the Securities Exchange Act of 1934, as amended, and any successor federal statute, and the rules and regulations thereunder, all as the same shall be in effect from time to time.

"Existing Shelf Registration Statement" has the meaning set forth in Section 4.01(a).

"FINRA" means the Financial Industry Regulatory Authority.

"FIRPTA Certificate" has the meaning set forth in Section 3.03(e).

"<u>Governmental Entity</u>" means any supranational, foreign, domestic, federal, territorial, state, county, city, township or other local governmental authority, or any regulatory, self-regulatory, administrative or other agency, instrumentality, court, government organization, quasigovernmental organization, mediator, arbitrator or arbitral forum (whether public or private), commission, tribunal thereof, or any political or other subdivision, department or branch of any of the foregoing, or any private body exercising any tax, regulatory or governmental or quasi-governmental authority.

"Indemnified Party" has the meaning set forth in Section 4.06(c).

"<u>Indemnifying Party</u>" has the meaning set forth in <u>Section 4.06(c)</u>.

"<u>Investor</u>" has the meaning set forth in the Preamble.

"Investor Designee Termination Event" has the meaning set forth in Section 2.06.

"Investor Designee" and "Investor Designees" have the meanings set forth in Section 2.01.

"Investor Indemnitees" has the meaning set forth in Section 4.06(a).

"<u>Law</u>" means any foreign or domestic constitutional provision, act, statute or other law, ordinance, rule, regulation or interpretation of any Governmental Entity and any binding and enforceable decree, injunction, judgment, order, ruling, assessment, writ, doctrine, assessment or arbitration award or similar form of decision or determination issued by a Governmental Entity.

"<u>Lockup Period</u>" means, with respect to the Shares Beneficially Owned by the Investor, the period commencing on the date of this Agreement and ending on the day that is one (1) year from the date of this Agreement.

"Losses" shall have the meaning set forth in Section 4.06(a).

"<u>Other Securities</u>" means the Common Stock or other securities of the Company which the Company is registering pursuant to a Registration Statement covered by <u>ARTICLE 4</u>.

"Permitted Acquisition" has the meaning set forth in Section 3.01(a).

"<u>Permitted Transfer</u>" has the meaning set forth in <u>Section 3.03(d)</u>.

"<u>Person</u>" means any individual, sole proprietorship, partnership, limited liability company, corporation, association, joint stock company, trust, joint venture, unincorporated organization, any other business organization or entity, or Governmental Entity.

"Piggyback Notice" has the meaning set forth in Section 4.03(a).

"Piggyback Registration" has the meaning set forth in Section 4.03(a).

"<u>Prospectus</u>" means the prospectus included in any Registration Statement, as amended or supplemented by any prospectus supplement and by all other amendments thereto, including post-effective amendments, and all material incorporated by reference into such prospectus.

"<u>Registrable Securities</u>" means the Shares acquired by the Investor pursuant to the Stock Purchase Agreement, as well as any shares of Common Stock or other securities issued as (or issuable upon the conversion or exercise of any warrant, right or other security which is issued as) a dividend or other distribution with respect to, or in exchange generally for, or in replacement generally of, such Shares or other Registrable Securities and any securities issued in exchange for such Shares or other Registrable Securities in any merger, reorganization, consolidation, share exchange, recapitalization, restructuring or other comparable transaction of the Company. As to any particular Registrable Securities, once issued such securities shall cease to be Registrable Securities when (i) a Registration Statement with respect to the sale by the Investor has been declared or deemed effective by the SEC and such securities have been disposed of pursuant to such effective Registration Statement, (ii) such securities have been otherwise Transferred (other than pursuant to <u>Section 3.03(d)(i)</u> and in accordance with <u>Section 6.04</u>), (iii) such securities shall have ceased to be outstanding or (iv) such securities have been or could all be sold in a single transaction without volume or other limitations pursuant to Rule 144 (or any similar provisions then in force) under the Securities Act.

"Registration Expenses" has the meaning set forth in Section 4.04(a).

"<u>Registration Statement</u>" means any registration statement of the Company under the Securities Act which permits the public offering of any of the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus, amendments and supplements to such registration statement, including post-effective amendments, all exhibits and all material incorporated by reference or deemed to be incorporated by reference in such registration statement.

"SEC" means the U.S. Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended, and any successor federal statute, and the rules and regulations thereunder, all as the same shall be in effect from time to time.

"Shares" has the meaning set forth in the Preamble.

"Shelf Date" has the meaning set forth in Section 4.01(a).

"Shelf Registration Statement" has the meaning set forth in Section 4.01(a).

"Shelf Take-Down Notice" has the meaning set forth in Section 4.01(b).

"Significant Subsidiary," means a significant subsidiary (as defined under the Exchange Act) of the Company.

"<u>Standstill Period</u>" means the period commencing on the Closing Date and continuing until the earlier to occur of (i) a Change of Control or (ii) the date which is 90 days following the termination of the Investor's rights pursuant to <u>Section 2.06</u>.

"Stock Purchase Agreement" has the meaning set forth in the Recitals.

"Suspension Period" has the meaning set forth in Section 4.05(a)(ii).

"<u>Transfer</u>" means (i) sell, assign, give, pledge, encumber, hypothecate, mortgage, exchange or otherwise dispose, (ii) grant to any Person any option, right or warrant to purchase or otherwise receive, or (iii) enter into any swap or other agreement that transfers, in whole or in part, any of the economic consequences or other rights of ownership.

"Viviti" has the meaning set forth in the Recitals.

# ARTICLE 2 BOARD REPRESENTATION

Section 2.01 **Investor Designee Appointment and Nomination Right**. The Investor shall have the right, but not the obligation, to designate two nominees to serve as directors of the Company (each, an "<u>Investor Designee</u>" and, together, the "<u>Investor Designees</u>"). In the event the Investor determines to designate the initial Investor Designees, the Investor shall notify the Company in writing of the names of the initial Investor Designees. Promptly following receipt by the Company of all documentation reasonably requested by the Company in connection with the appointment of the initial Investor Designees, the Company shall increase the size of the Board by two, and fill the resulting vacancies with the initial Investor Designees in accordance with the Company's Bylaws. Thereafter, the Company shall (a) include the Investor Designees in its slate of nominees for election to the Board of Directors at each annual or special meeting of stockholders of the Company following the Closing at which directors are to be elected and at which the seats held by the Investor Designees are subject to election (such annual or special meetings, the "<u>Election Meetings</u>") and (b) recommend that the Company's stockholders vote in favor of the Investor Designees, support the Investor Designees for election in a manner no less favorable than the manner in which the Company supports its other nominees,

and otherwise use commercially reasonable efforts to cause the election of the Investor Designees to the Board of Directors at each of the Election Meetings. The foregoing appointment and nomination rights will be subject to the Investor Designees satisfying the Company's Board Qualifications (as defined in Section 2.03); <u>provided</u> that, if an Investor Designee does not meet the Board Qualifications, (i) the Company will not nominate a replacement candidate in place of the rejected Investor Designee (unless the Investor does not nominate a replacement candidate pursuant to its rights in the following clause (ii) within the time period stated in such clause), and (ii) the Investor shall have the right (if exercised as promptly as reasonably practicable and in any event within 30 days) to nominate a replacement candidate in place of the rejected Investor Designee that meets the Board Qualifications is put forward by the Investor.

Section 2.02 <u>Vacancies</u>. At any time prior to an Investor Designee Termination Event, if an Investor Designee who has been duly elected to the Board resigns from the Board, is removed (with or without cause) pursuant to applicable Law or the Company's Bylaws, fails to satisfy the Board Qualifications, dies or otherwise cannot or is not willing to stand for reelection or to continue to serve as a member of the Board, the Company shall use commercially reasonable efforts to cause the vacancy to be filled by a new Investor Designee prior to or concurrent with any further meeting or action by the Board.

Section 2.03 **Board Qualifications**. Each Investor Designee shall, at the time of nomination and at all times thereafter until such individual's service on the Board of Directors ceases, (a) meet any applicable requirements under applicable Law, stock exchange rules or the Company's corporate governance policies to be a member of the Board of Directors, (b) be an executive officer or former executive officer of the Investor, (c) not be an officer or director of any Competitor or Competitor Transferee, and (d) prior to being nominated, agree to comply with the requirements of this Section 2.03 (the "<u>Board</u> <u>Qualifications</u>"). The Company shall not revise or amend the Board Qualifications in a manner that has the intent or effect of adversely affecting the nomination or election of an Investor Designee (by for instance, adding requirements that all directors meet citizenship or independence requirements that would disqualify Persons known by the Company to be the Investor's probable designees).

Section 2.04 **Compensation, Indemnification and Insurance**. Investor Designees shall be entitled to the same retainer, equity compensation or other fees or compensation, including travel and expense reimbursement, paid to the non-employee directors of the Company for their services as a director, including any service on any committee of the Board. For so long as an Investor Designee continues to serve as a director and for a period of six (6) years thereafter, the Company shall, to the extent permitted by applicable Laws, indemnify such Investor Designees and shall maintain in full force and effect directors' and officers' liability insurance in reasonable amounts from established and reputable insurers to the same extent it now indemnifies and provides insurance for the non-executive members of the Board of Directors. In all directors' and officers' insurance policies, each Investor Designee shall be covered as an insured in such a manner as to provide the Investor Designee with rights and benefits under such insurance policies no less favorable than provided to the other non-executive directors of the Company.

Section 2.05 <u>**Committees**</u>. Unless otherwise agreed to by the Board of Directors, the Investor Designees shall not be appointed to or otherwise gain membership on any of the Board committees.

#### Section 2.06 Termination of Investor Designee Rights. Notwithstanding the foregoing:

(a) the Investor's rights under this <u>ARTICLE 2</u> with respect to one of the Investor Designees shall terminate automatically at the end of the second full calendar year following the Closing Date; and

(b) all of the Investor's rights under this <u>ARTICLE 2</u> shall terminate automatically (in the case of (i) and (ii)) or following written notice from the Company (in the case of (iii)) upon the earliest to occur of:

Agreement;

(i) the Investor ceasing to Beneficially Own at least fifty percent (50%) of the Shares received pursuant to the Stock Purchase

(ii) if the Investor has first sold at least ten percent (10%) of the Shares received pursuant to the Stock Purchase Agreement, the Investor ceasing to Beneficially Own at least five percent (5%) of the total issued and outstanding Common Stock; or

(iii) any (A) breach by the Investor of the provisions of <u>ARTICLE 3</u> of this Agreement or (B) material breach by the Investor of the Non-Competition Agreement between the parties entered into upon the Closing of the transactions contemplated by the Stock Purchase Agreement, provided that in the case of (B), if such breach is a Remediable Breach (as such term is defined in the Non-Competition Agreement) and such breach is cured pursuant to the dispute resolution procedures set forth in Section 4 thereof, then such breach shall not be an Investor Designee Termination Event (as defined below) hereunder.

(each of the events described in subsections (a) and (b) of this Section 2.06 are referred to as an "<u>Investor Designee Termination Event</u>"). The Investor shall cause any applicable Investor Designee to tender his resignation from the Board of Directors promptly upon the occurrence of an Investor Designee Termination Event.

Section 2.07 <u>Non-Transferability</u>. The Investor may not Transfer to any Person all or any portion of its rights under this <u>ARTICLE 2</u> under any circumstances, notwithstanding the Transfer of all or any portion of the Shares.

#### ARTICLE 3 INVESTOR RESTRICTIONS

Section 3.01 <u>Standstill</u>. During the Standstill Period and unless otherwise approved by the Board of Directors (excluding any Investor Designees), the Investor will not, and will cause each of its Affiliates, directors, officers or employees not to, directly or indirectly, acting alone or as part of a 13D Group:

(a) acquire or agree, offer, seek or propose, whether by purchase, tender or exchange offer, by joining any 13D Group or otherwise, to acquire ownership of any, (x) of the businesses or material assets of the Company or any Significant Subsidiary (except for any transaction in the ordinary course of business), (y) any Equity Securities or any equity securities of any Significant Subsidiary, or (z) rights or options to acquire such ownership other than (i) the delivery of the Shares pursuant to the Stock Purchase Agreement, (ii) the acquisition of the Company's securities as a result of any stock splits, stock dividends or other distributions or recapitalizations or offerings made available by the Company to holders of Common Stock, including rights offerings, (iii) any acquisition of the Company's securities approved by the Board of Directors (excluding any Investor Designees), or (iv) any acquisition of the Company's securities pursuant to a Permitted Transfer (each event listed in clauses (i) through (iv), a "<u>Permitted Acquisition</u>");

(b) engage in any "solicitation" (within the meaning of the Exchange Act) of proxies or consents relating to the election of directors with respect to the Company, or become a "participant" in any "election contest" (both within the meaning of the Exchange Act) seeking to elect directors not nominated by the Board of Directors, other than the Investor Designees, or call, or seek or propose to call, any meeting of the Company's shareholders in connection therewith;

(c) in any manner, agree, attempt, seek or propose to deposit any securities of the Company or any rights to acquire (whether currently, upon lapse of time, following the satisfaction of any conditions, upon the occurrence of any event or any combination of the foregoing) any Equity Securities of the Company in any voting trust or similar arrangement;

(d) form or join in the formation of a 13D Group (other than a 13D Group consisting only of the Investor and its Affiliates) with respect to any Equity Securities or equity securities of any Significant Subsidiary, or grant to any Person any proxy with respect to the exercise of voting rights with respect to the Shares; or

(e) publicly announce any intention, plan or arrangement or finance (or arrange financing for) any Person in connection with any of the

### foregoing.

# Section 3.02 Permitted Actions.

(a) The restrictions set forth in <u>Section 3.01(a)</u>–Section 3.01(e) shall cease to have effect if any of the following occurs (<u>provided</u>, that if any event described in this <u>Section 3.02</u> occurs and, during the following 12 months, none of the transactions described below has been consummated, then the restrictions set forth in <u>Section 3.01</u> shall thereafter resume and continue to apply in accordance with their terms):

(i) in the event that the Company enters into a definitive agreement for a merger, consolidation or other business combination transaction as a result of which the stockholders of the Company immediately prior to the consummation of such transaction would not own (including Beneficial Ownership) more than fifty percent (50%) of the aggregate voting rights of the surviving entity;

(ii) in the event that a tender offer or exchange offer for more than 50% of the Common Stock is commenced by any Person (and not involving any breach of <u>Section 3.01</u>) which tender offer or exchange offer, if consummated, would result in a Change of Control, and the Board of Directors recommends that the stockholders of the Company tender their shares in response to such offer within ten (10) Business Days after the commencement thereof or such longer period as shall then be permitted under U.S. federal securities Laws; or

(iii) in the event that the Company makes any public announcement indicating that it is actively pursuing a Change of Control, and such announcement is not disavowed by the Company pursuant to a public announcement made within two Business Days of such first announcement.

(b) Notwithstanding the foregoing, this <u>Section 3.02</u> shall not restrict or otherwise apply to the activities of any Investor Designee in such Person's capacity as a director of the Company, acting in good faith and in satisfaction of such Person's duties to the Company in such capacity.

#### Section 3.03 Dispositions.

(a) <u>Lockup Period</u>. The Investor agrees that during the Lockup Period, without the prior written consent of the Company, the Investor shall not, and shall not authorize, permit or direct its subsidiaries or Affiliates to, directly or indirectly, Transfer any of the Shares.

# (b) Competitor Transferees.

(i) During the term of this Agreement, the Investor agrees that it shall not, and shall not allow any of its Affiliates to, Transfer, directly or indirectly, any of the Shares knowingly to any Person identified in that certain competitor list letter (the "<u>Competitor List Letter</u>"), dated as of the Closing Date, and provided to the Investor at the Closing, as such letter may be amended from time to time in accordance with <u>Section</u> <u>3.03(b)(ii)</u> (collectively, "<u>Competitors</u>"), or to any Affiliate of any such Person (Competitors and their respective Affiliates collectively, "<u>Competitor Transfer</u>"), and any such Transfer shall be null and void; <u>provided</u>, <u>however</u>, that the foregoing shall not prohibit any sale of Shares through brokers' transactions to a Person who the Investor has no reason to believe is a competitor (and, for the avoidance of doubt, Investor shall have no duty of inquiry in connection with such brokers' transactions).

(ii) The Competitor List Letter identifies the Competitors as of the date hereof. The Company may amend the Competitor List Letter following the date hereof to add or remove Competitors from such Competitor List Letter, each such amendment to be effective upon delivery of written notice thereof to the Investor, <u>provided</u> that (x) any Person so added to the Competitor List Letter as a Competitor must be a material direct competitor of the Company in one of its principal lines of business, as determined in good faith by the Company, (y) there shall not be more than ten (10) Competitors in total identified on the Competitor List Letter at any time, and (z) the Company may not amend

the Competitor List Letter (A) prior to the six month anniversary of the Closing, or (B) more than twice per each twelve-month period thereafter.

(c) <u>5% Threshold</u>. During the term of this Agreement, the Investor agrees that it shall not, and shall not allow any of its Affiliates to, Transfer, directly or indirectly, any of the Shares knowingly to any member of a 13D Group and any such Transfer shall be null and void; <u>provided</u>, <u>however</u>, that the foregoing shall not prohibit any sale of Shares through brokers' transactions to a Person who the Investor has no reason to believe is a member of a 13D Group (and, for the avoidance of doubt, Investor shall have no duty of inquiry in connection with such brokers' transactions).

(d) <u>Permitted Transfers</u>. Notwithstanding the foregoing, the following Transfers of the Shares shall be permitted at any time (each a "<u>Permitted Transfer</u>"):

(i) by the Investor to any of its Affiliates and by any Affiliate of the Investor to any other Affiliate of the Investor, <u>provided</u> that prior to and as a condition to any such Transfer, (A) the Company is furnished with written notice of the name and address of such Affiliate and the Shares Transferred, and (B) such Affiliate agrees in writing to be bound by and subject to the terms and conditions of this Agreement; and <u>provided</u>, <u>further</u>, that, with respect to any Transfer of registration rights under <u>ARTICLE 4</u> in connection with any Permitted Transfer under this <u>Section</u> <u>3.03(d)(i)</u> prior to and as a condition to any such Transfer, such Affiliate agrees to designate the Investor as its exclusive representative, agent and attorney-in-fact to exercise all of its rights thereunder pursuant to a written agreement in form reasonably satisfactory to the Company; or

(ii) by the Investor to a third party pursuant to a tender offer, exchange offer, merger, consolidation or other transaction (A) which is recommended to the stockholders of the Company by the Board of Directors; or (B) in the case of a merger or other business combination transaction, which has been approved by the stockholders of the Company.

(e) <u>Real Property Interests</u>. In connection with a subsequent disposition of Shares, the Investor may request a certification, in accordance with applicable Treasury Regulations, to the effect that (i) any interests in the Company do not constitute "U.S. real property interests" within the meaning of section 897(c)(1) of the Internal Revenue Code of 1986, as amended and (ii) the Company is not, and has not been during the shorter of (A) the 5 years preceding the date of the certification or (B) the Investor's holding period for the Shares, a United States real property holding corporation within the meaning of section 897(c)(2) of the Code (a "<u>FIRPTA Certificate</u>"). Upon the request of a FIRPTA Certificate by the Investor, the Company agrees to execute and deliver such FIRPTA Certificate within 10 days of such request, unless the Company determines, after reasonable diligence, that it cannot execute the certification because it cannot certify as to the information contained in clauses (i) or (ii). The Company's obligation under this <u>Section 3.03(e)</u> shall continue until such time as the Investor has disposed of all of the Shares received pursuant to the Stock Purchase Agreement.

# ARTICLE 4 REGISTRATION RIGHTS

#### Section 4.01 Shelf Registration.

(a) On or before the expiration of the Lockup Period (the "<u>Shelf Date</u>"), so long as the Company is eligible to do so, the Company shall file with the SEC a Registration Statement providing for registration and resale, on a continuous or delayed basis pursuant to Rule 415 under the Securities Act, as such rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the SEC, of all of the Registrable Securities, <u>provided</u> that such obligation shall be satisfied if the Company shall have in effect an automatically effective shelf registration statement on Form S-3ASR (or any comparable or successor form or forms then in effect) (an "<u>Existing Shelf Registration Statement</u>") as of the Shelf Date (any such registration statement, a "<u>Shelf Registration Statement</u>") that covers resale of the Registrable Securities; <u>provided</u>, <u>further</u>, that for the avoidance of doubt, the existence of an Existing Shelf Registration Statement shall not have any effect on the restrictions set forth in <u>Section 3.03</u>. The Shelf Registration Statement shall be on Form S-3 (or any comparable or successor form or forms then in effect) under the Securities Act; <u>provided</u>, <u>however</u>, that if the Company is a well-known seasoned issuer (as defined in Rule 405 under the Securities Act) at the time of filing of the Shelf Registration Statement with the SEC, such Shelf Registration Statement shall be designated by the Company as an automatic shelf registration statement (as defined in Rule 405 under the Securities Act). The Company shall use its commercially reasonable efforts to keep the Shelf Registration Statement is not on Form S-3ASR, the Company shall use commercially reasonable efforts to cause the Shelf Registration Statement to become effective, as promptly as practicable, but in no event later than one hundred twenty (120) days following the filing of the Shelf Registration Statement.

(b) The Investor agrees that if it wishes to sell Registrable Securities pursuant to a Shelf Registration Statement and related Prospectus, it will do so in accordance with this <u>Section 4.01(b)</u> and <u>Section 4.05</u>. In the event the Investor wishes to sell Registrable Securities pursuant to a Shelf Registration Statement and related Prospectus, whether in an underwritten offering or otherwise that would require action by the Company pursuant to <u>Section 4.01(b)(i)</u>, the Investor agrees to notify the Company of such intent (a "<u>Shelf Take-Down Notice</u>") and shall deliver a Shelf Take-Down Notice at least twenty (20) Business Days prior to any intended distribution of Registrable Securities under the Shelf Registration Statement, it being agreed that if the Investor intends to distribute any Registrable Securities by means of an underwritten offering it shall promptly so advise the Company and the Company shall reasonably cooperate with the Investor to facilitate such distribution, including the actions required pursuant to <u>Section 4.05(a)(viii)</u> and, if a Company Supported Distribution is requested, <u>Section 4.05(a)(xiv</u>). From and after the date the Shelf Registration Statement is declared or deemed effective, the Company shall, as promptly as practicable after the date of the Shelf Take-Down Notice:

(i) if required by applicable Law, file with the SEC a post-effective amendment to the Shelf Registration Statement or prepare and, if required by applicable Law, file a supplement to the related Prospectus or a supplement or amendment to any

document incorporated therein by reference or file any other required document so that the Investor is named as a selling security holder in the Shelf Registration Statement and the related Prospectus in such a manner as to permit the Investor to deliver or be deemed to have delivered such Prospectus to purchasers of Registrable Securities in accordance with applicable Law and, if the Company shall file a post-effective amendment to the Shelf Registration Statement, use commercially reasonable efforts to cause such post-effective amendment to be declared or deemed effective under the Securities Act as promptly as practicable;

(ii) provide the Investor copies of any documents filed pursuant to <u>Section 4.01(b)(i)</u>; and

(iii) notify the Investor as promptly as practicable after the effectiveness under the Securities Act of any post-effective amendment filed pursuant to <u>Section 4.01(b)(i)</u>; <u>provided,however</u>, that if such Shelf Take-Down Notice is delivered during a Suspension Period, the Company shall so inform the Investor and shall take the actions set forth in clauses (i) and (ii) above promptly upon expiration of the Suspension Period in accordance with <u>Section 4.05</u>; <u>provided</u>, <u>further</u>, that the Investor shall not be entitled to deliver to the Company more than one (1) Shelf Take-Down Notice in any twelve (12) month period and each Shelf Take-Down Notice may only be made if the sale of the Registrable Securities covered thereby is reasonably expected to result in aggregate gross cash proceeds in excess of Fifty Million Dollars (\$50,000,000) (without regard to any underwriting discount or commission) and, <u>provided</u>, <u>further</u>, that the Investor shall not be entitled to request more than three (3) Company Supported Distributions in the aggregate. A Shelf Take-Down Notice may not be made without the Company's prior written consent (not to be unreasonably withheld, delayed or conditioned) if the sale of the Registrable Securities covered thereby is reasonably expected to exceed the greater of

(i) the value of twelve million five hundred thousand (12,500,000) Shares at the time of the sale or (ii) Five Hundred Million Dollars (\$500,000,000).

(c) If any of the Registrable Securities to be sold pursuant to a Shelf Registration Statement are to be sold in a firm commitment underwritten offering which underwritten offering was initially requested by the Investor pursuant to a Shelf Take-Down Notice, and the managing underwriter of such underwritten offering advises the Investor that it is their good faith opinion that the total number or dollar amount of Registrable Securities proposed to be sold in such offering, together with any Other Securities proposed to be included by the Company or holders thereof which are entitled to include securities in such Registration Statement, exceeds the total number or dollar amount of such securities that can be sold without having an adverse effect on the price, timing or distribution of the Registrable Securities to be so included, together with all such Other Securities that in the opinion of such managing underwriter can be sold without so adversely affecting such offering, and such number of Registrable Securities and Other Securities shall be allocated for inclusion as follows:

(i) first, the Registrable Securities for which inclusion in such underwritten offering was requested by the Investor; and

(ii) second, among the Company and any holders of Other Securities, pro rata, based on the number of Other Securities proposed to be included in such underwritten offering by the Company and the number of Other Securities Beneficially Owned by each such holder of Other Securities;

(d) The Investor shall have the right to notify the Company that it has determined that the Shelf Take-Down Notice be abandoned or withdrawn, in which event the Company shall promptly abandon or withdraw all activities undertaken in connection with such offering with respect to Registrable Securities, and such withdrawn Shelf Take-Down Notice shall not count against the limit of Shelf Take-Down Notices or Company Supported Distributions, as applicable; provided, however, that the Company shall not be required to pay for expenses of any registration proceeding begun pursuant to Section 4.01(a), which has been subsequently abandoned or withdrawn pursuant to this Section 4.01(d) at the request of the Investor, and shall be reimbursed by the Investor for reasonable and documented out-of-pocket expenses (including legal fees and printing expenses) so incurred, unless the withdrawal is based upon material adverse information concerning the Company that the Company has not publicly disclosed in compliance with applicable securities Laws at least five (5) Business Days prior to the Company's receipt of such withdrawal request.

(e) In the event that the SEC sets forth a limitation on the securities that may be registered on a particular Shelf Registration Statement, the Company may reduce the number of securities to be registered on such Shelf Registration Statement to such number of securities as allowed by the SEC.

#### Section 4.02 Demand Registration.

(a) At any time following the expiration of the Lockup Period, if the Company is unable to file, cause to be effective or maintain the effectiveness of a Shelf Registration Statement as required under <u>Section 4.01</u>, the Investor shall have the right, by delivering a written notice to the Company (a "<u>Demand Notice</u>"), to require the Company to register under and in accordance with the provisions of the Securities Act the number of Registrable Securities Beneficially Owned by the Investor and requested by such Demand Notice to be so registered (a "<u>Demand Registration</u>"); provided, however, that the Company shall not be required to effect more than three (3) Demand Registrations for underwritten offerings pursuant to this <u>Section 4.02(a)</u>; provided, further, that the Investor shall not be entitled to deliver to the Company more than two (2) Demand Registrations in any twelve (12) month period; and provided, further, that a Demand Registration may not be made until at least one hundred and twenty (120) days after the date of a prior Demand Registration, and, in any event, a Demand Notice may only be made if the sale of the Registrable Securities requested to be registered by the Investor is reasonably expected to result in aggregate gross cash proceeds in excess of Fifty Million Dollars (\$50,000,000) (without regard to any underwriting discount or commission); and provided, further, that the Investor shall not be entitled to request more than three (3) Company Supported Distributions in the aggregate (including underwritten Demand Registrations). A Demand Registration may not exceed the greater of (i) the value of twelve million five hundred thousand (12,500,000) Shares or (ii) Five Hundred Million Dollars (\$500,000,000) without the Company's prior written consent (not to be unreasonably withheld, delayed or conditioned). A Demand Notice shall also specify the expected method or methods of disposition of the applicable

Registrable Securities. Following receipt of a Demand Notice, the Company shall use commercially reasonable efforts to file, as promptly as reasonably practicable, but not later than ninety (90) Business Days after receipt by the Company of such Demand Notice, a Registration Statement relating to the offer and sale of the Registrable Securities requested to be included therein by the Investor in accordance with the methods of distribution elected (a "Demand Registration Statement") and shall use commercially reasonable efforts to cause such Registration Statement to be declared effective under the Securities Act as promptly as practicable after the filing thereof, but in no event later than one hundred twenty (120) days following the date of filing the Registration Statement, it being agreed that if the Investor intends to distribute any Registrable Securities by means of an underwritten offering it shall promptly so advise the Company and the Company shall cooperate with the Investor to facilitate such distribution, including the actions required pursuant to <u>Section 4.05(a)(viii)</u> and, if a Company Supported Distribution is requested, <u>Section 4.05(a)(xiv</u>).

(b) If any of the Registrable Securities registered pursuant to a Demand Registration are to be sold in a firm commitment underwritten offering, and the managing underwriter of such underwritten offering advises the Investor in writing that it is their good faith opinion that the total number or dollar amount of Registrable Securities proposed to be sold in such offering, together with any Other Securities proposed to be included by the Company or holders thereof which are entitled to include securities in such Registration Statement, exceeds the total number or dollar amount of such securities that can be sold without having an adverse effect on the price, timing or distribution of the Registrable Securities to be so included together with all such Other Securities, then there shall be included in such firm commitment underwritten offering the number or dollar amount of Registrable Securities and such Other Securities that in the opinion of such managing underwriter can be sold without so adversely affecting such offering, and such number of Registrable Securities shall be allocated for inclusion as follows:

(i) first, the Registrable Securities for which inclusion in such underwritten offering was requested by the Investor; and

(ii) second, among the Company and any holders of Other Securities, pro rata, based on the number of Other Securities proposed to be included in such underwritten offering by the Company and the number of Other Securities Beneficially Owned by each such holder of Other Securities;

(c) In the event of a Demand Registration, the Company shall be required to maintain the continuous effectiveness of the applicable Registration Statement for a period of at least thirty (30) days after the effective date thereof or such shorter period in which all Registrable Securities included in such Registration Statement have actually been sold.

(d) The Investor shall have the right to notify the Company that it has determined that the Registration Statement relating to a Demand Registration be abandoned or withdrawn with respect to Registrable Securities, in which event the Company shall promptly abandon or withdraw such Registration Statement with respect to Registrable Securities and such abandoned or withdrawn registration shall not count against the limit of Demand Registrations or Company Supported Distributions, as applicable; <u>provided</u>, <u>however</u>, that the Company shall not be

required to pay for expenses of any registration proceeding begun pursuant to <u>Section 4.02(a)</u>, which has been subsequently abandoned or withdrawn pursuant to this <u>Section 4.02(d)</u> at the request of the Investor, and shall be reimbursed by the Investor for reasonable and documented out-of-pocket expenses (including legal fees and printing expenses) so incurred, unless the withdrawal is based upon material adverse information concerning the Company that the Company has not publicly disclosed at least five (5) Business Days prior to the Company's receipt of such withdrawal request.

(e) Notwithstanding anything contained herein to the contrary, with the prior written consent of the Investor (which consent shall not be unreasonably withheld, conditioned or delayed), the Company shall be entitled to coordinate (but not in violation of <u>Section 4.02</u>) any offerings under this <u>Section 4.02</u> with any offerings to be effected pursuant to similar agreements with the holders of Other Securities, including, if practicable, by filing one Registration Statement for all Other Securities.

(f) The Investor may not make a Demand Registration in the event that a Shelf Registration Statement is effective and covers the number of Registrable Securities that the Investor wishes to sell.

(g) In the event that the SEC sets forth a limitation on the number of securities to be registered in a particular Demand Registration, the Company may reduce the number of securities to be registered in such Demand Registration to such number of securities as allowed by the SEC.

# Section 4.03 Piggyback Registration.

(a) At any time following the expiration of the Lockup Period, if the Company proposes to file a registration statement under the Securities Act with respect to an offering (i) by the Company for its own account (other than a registration statement (A) on Form S-4, Form S-8 or any successor forms thereto, (B) filed solely in connection with any employee benefit, dividend reinvestment, or any other similar plan or (C) for the purpose of effecting a rights offering afforded to all holders of the Shares) or (ii) for the account of any of its security holders, the Company will give the Investor written notice of such filing at least ten (10) Business Days' prior to the anticipated filing date (the "<u>Piggyback Notice</u>"). The Piggyback Notice shall offer the Investor the opportunity to include in such registration statement the number of Registrable Securities (for purposes of this <u>Section 4.03</u>, "Registrable Securities" shall be deemed to mean solely securities of the same type as those proposed to be offered for the account of the Company or its security holders) as they may request (a "<u>Piggyback Registration</u>"). Subject to <u>Section 4.03(b)</u>, the Company shall include in each such Piggyback Registration all Registrable Securities with respect to which the Company has received a written request from the Investor for inclusion therein within five (5) Business Days after notice has been given to the Investor. The Company shall be required to maintain the effectiveness of the Registration Statement for a Piggyback Registration for a period of at least thirty (30) days after the effective date thereof or such shorter period in which all Registrable Securities included in such Registration Statement have actually been sold.

(b) If any of the securities to be registered pursuant to the registration giving rise to the Investor's rights under this <u>Section 4.03</u> are to be sold in an underwritten offering, the Investor shall be permitted to include all Registrable Securities requested to be included in such registration in such offering on the same terms and conditions as the securities of the Company or its security holders included therein; <u>provided</u>, <u>however</u>, that if such offering involves a firm commitment underwritten offering and the managing underwriter of such underwritten offering, together with all Other Securities that the total number or dollar amount of Registrable Securities proposed to be sold in such offering, exceeds the total number or dollar amount of such registration intend to include in such offering, exceeds the total number or dollar amount of such securities that can be sold without having an adverse effect on the price, timing or distribution of the Registrable Securities to be so included together with all such Other Securities, then there shall be included in such firm commitment underwritten offering such offering, and other Securities that in the opinion of such managing underwriter can be sold without so adversely affecting such offering, and such number of Registrable Securities shall be allocated for inclusion as follows:

(i) first, all Other Securities being sold by the Company or by any Person (other than the Investor) exercising a contractual right to demand registration pursuant to which such registration statement was filed;

- (ii) second, to the Investor, and
- (iii) third, among any other holders of Other Securities.

(c) The Company shall have the right to terminate or withdraw any registration initiated by it under this <u>Section 4.03</u> prior to the effectiveness of the related Registration Statement and shall have no obligation to register any Registrable Securities in connection with such registration, except to the extent provided herein. The Registration Expenses of such withdrawn Piggyback Registration shall be borne by the Company in accordance with <u>Section 4.04</u>. The Investor shall have the right to withdraw at least two (2) Business Days prior to the planned effective date of the related Registration Statement; <u>provided</u>, <u>however</u>, that the Company shall not be required to pay for expenses relating to the proposed inclusion of the Investor's Registrable Securities in such Piggyback Registration, and shall be reimbursed by the Investor for reasonable and documented out-of-pocket expenses (including legal fees and printing expenses) so incurred, unless the withdrawal is based upon material adverse information concerning the Company that the Company has not publicly disclosed in compliance with applicable securities Laws at least five (5) Business Days prior to the Company's receipt of such withdrawal request.

(d) In the event that the SEC sets forth a limitation on the number of securities that may be registered in a particular Piggyback Registration, the Company may reduce the number of securities to be registered in such Piggyback Registration to such number of securities as allowed by the SEC.

#### Section 4.04 Registration Expenses.

(a) Expenses of the Company. Except to the extent otherwise provided herein, in connection with registrations pursuant to Section 4.01, Section 4.02, or Section 4.03, the Company shall pay all of the registration expenses incurred in connection with the registration thereunder (the "Registration Expenses"), including, without limitation, all: (i) reasonable registration and filing fees, (ii) Financial Industry Regulatory Authority, Inc. fees, (iii) printing expenses, (iv) fees and disbursements of the Company's counsel, (v) blue sky fees and expenses, (vi) expenses of the Company's independent accountants in connection with any regular or special reviews or audits incident to or required by any such registration, (vii) expenses incurred in connection with making road show presentations and holding meetings with potential investors and (viii) up to fifty thousand dollars (\$50,000) of reasonable fees and disbursements of one firm of attorneys acting as counsel of the Investor.

(b) <u>Expenses of the Investor</u>. The Investor shall be responsible for (i) any allocable underwriting fees, discounts or commissions, (ii) any allocable commissions of brokers and dealers, (iii) fees and disbursements of the Investor's counsel other than as provided in <u>Section 4.04(a)</u>, and (iv) capital gains, income and transfer taxes, if any, relating to the sale of <u>Registrable Securities of the Investor</u>.

#### Section 4.05 Registration Procedures.

(a) In connection with the registration of any Registrable Securities pursuant to this Agreement, the Company will keep the Investor advised in writing as to the initiation of each such registration and the Company will:

(i) Use commercially reasonable efforts to keep each Registration Statement continuously effective during the period such Registration Statement is required to remain effective pursuant to the terms of this Agreement; upon the occurrence of any event that would cause the Registration Statement or the Prospectus contained therein (A) to contain a material misstatement or omission or (B) not to be effective and usable for resale of Registrable Securities during the period such Registration Statement is required to remain effective pursuant to the terms of this Agreement, the Company shall file promptly an appropriate amendment to the Registration Statement, a supplement to the Prospectus or a report filed with the SEC pursuant to Section 13(a), 13(c), 14 or 15(d) of the Exchange Act, in the case of clause (A), correcting any such misstatement or omission, and, in the case of either clause (A) or (B), the Company shall use commercially reasonable efforts to cause such amendment to be declared or deemed effective and the Registration Statement and the related Prospectus to become usable for their intended purposes as soon as practicable thereafter.

(ii) Notwithstanding anything to the contrary contained herein, the Company may delay filing or suspend the effectiveness of a Registration Statement and the Investor's right to sell thereunder (each such period, a "<u>Suspension Period</u>") if (A) the Company is pursuing an acquisition, merger, reorganization, disposition or similar transaction and the Company determines in good faith that the Company's ability to pursue or consummate such a transaction would be materially adversely affected by any

required disclosure of such transaction in the registration statement, or (B) the Company has experienced some other material non-public event the disclosure of which at such time could reasonably be expected to materially adversely affect the Company; <u>provided</u> that the Company may not take any action pursuant to this <u>Section 4.05(a)</u> for a period of time in excess of 120 days in the aggregate in any twelve (12) month period.

(iii) Prepare and file with the SEC such amendments and post-effective amendments to each Registration Statement as may be necessary to keep such Registration Statement effective during the period provided herein.

(iv) Advise the Investor, promptly (which notice pursuant to clauses (B) through (D) below shall be accompanied by an instruction to suspend the use of the Prospectus until the Company shall have remedied the basis for such suspension and promptly thereafter notified the Investor of such remediation):

(A) when the Prospectus or any Prospectus supplement or post-effective amendment is proposed to be or has been filed, and, with respect to the Registration Statement or any post-effective amendment thereto, when the same has become effective;

(B) of any request by the SEC or any other Governmental Entity received by the Company for amendments to the Registration Statement or amendments or supplements to the Prospectus or for additional information relating thereto;

(C) of the issuance by the SEC of any stop order received by the Company suspending the effectiveness of the Registration Statement under the Securities Act or of the suspension by any state securities commission of the qualification of the Registrable Securities for offering or sale in any jurisdiction, or the threatening or initiation of any proceeding for any of the preceding purposes;

(D) of the receipt by the Company of any notification with respect to the suspension of the qualification or exemption from qualification of any of the Registrable Securities for sale in any jurisdiction, or the initiation or threatening of any proceeding for such purpose; or

(E) of the existence of any fact or the happening of any event, during the pendency of a distribution of Registrable Securities pursuant to a Registration Statement, that makes any statement of a material fact made in such Registration Statement, the Prospectus, any amendment or supplement thereto, or any document incorporated by reference therein untrue, or that requires the making of any additions to or changes in the Registration Statement or the Prospectus in order to make the statements therein not misleading.

(v) Unless any Registrable Securities shall be in book-entry form only, cooperate with the Investor to facilitate the timely preparation and delivery of certificates representing Registrable Securities to be sold and not bearing any restrictive legends (unless required by applicable securities Laws), and enable such Registrable Securities to be in such denominations and registered in such names as the Investor may request at least two (2) Business Days before any sale of Registrable Securities.

(vi) Use commercially reasonable efforts to promptly register or qualify any Registrable Securities under such other securities or blue sky laws of such jurisdictions within the United States as any Investor reasonably requests and which may be reasonably necessary or advisable to enable such Investor to consummate the disposition in such jurisdictions of the Registrable Securities owned by such Investor, keep such registrations or qualifications in effect for so long as the applicable Registration Statement is required to remain in effect and do any and all other acts and things which may be reasonably necessary or advisable to enable such Investor to consummate the disposition in such jurisdictions of the Registrable Securities owned by such Investor; <u>provided</u>, <u>however</u>, that the Company will not be required to (A) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this Agreement, (B) subject itself to taxation in any jurisdiction where it would not otherwise be subject to taxation but for this Agreement or (C) consent to general service of process in any jurisdiction where it would not otherwise be subject to such service but for this Agreement.

(vii) Use commercially reasonable efforts to promptly cause any Registrable Securities covered by a Registration Statement to be registered with or approved by such other Governmental Entity within the United States as may be necessary to enable the Investor to consummate the disposition of such Registrable Securities in accordance with the intended methods of disposition set forth in such Registration Statement.

(viii) In the event that the Investor advises the Company that the Investor intends to distribute any Registrable Securities by means of an underwritten offering, whether pursuant to <u>Section 4.01</u> or <u>Section 4.02</u>, enter into an underwriting agreement in customary form, scope and substance (including customary representations, warranties, covenants and indemnifications) and take all such other actions reasonably requested by the Investor or by the managing underwriter, if any, to expedite or facilitate the underwritten disposition of such Registrable Securities and deliver such documents and certificates as may be reasonably requested by the Investor, its counsel and the managing underwriter, if any.

(ix) Use its commercially reasonable efforts to prevent, or obtain the withdrawal of, any stop order or other order suspending the use of any Prospectus.

(x) Deliver to the Investor and each underwriter, if any, without charge, as many copies of the applicable Prospectus and any amendment or supplement thereto as the Investor or underwriter may reasonably request.

(xi) Cooperate with the Investor and the underwriters, if any, of such Registrable Securities and their respective counsel in connection with any filings required by Law to be made with FINRA.

(xii) Obtain opinions of counsel to the Company and updates thereof addressed to the Investor and the underwriters or initial purchasers, if any, covering matters as are customarily requested in opinions covering secondary resale offerings of companies of comparable size, maturities and lines of business as the Company.

(xiii) Obtain "comfort" letters and updates thereof from the Company's independent certified public accountants, such letters covering matters as are customarily requested in comfort letters covering secondary resale offerings of companies of comparable size, maturities and lines of business as the Company.

(xiv) Only in the case of a Company Supported Distribution, as requested by the managing underwriter in any such underwritten offering, provide reasonable assistance with the marketing of any such offering, including causing members of the Company's management team to participate in a reasonable and customary number of conference calls, investor meetings and due diligence sessions, in each case and, to the extent to be in-person, to take place in the continental United States; <u>provided</u>, that any such requested assistance shall not be required if it would, in the Company's reasonable judgment, interfere with the normal business operations of the Company in any substantial respect.

(b) The Investor agrees by acquisition of a Registrable Security that the Investor shall not be entitled to sell any of such Registrable Securities pursuant to a Registration Statement, or to receive a Prospectus relating thereto, unless the Investor has furnished the Company with the information set forth in the next sentence at least five (5) Business Days prior to the filing of the applicable Registration Statement or Prospectus. The Company may require the Investor pursuant to a Registration Statement to furnish to the Company such customary information regarding the Investor and the distribution of such Shares as the Company may reasonably require for inclusion in such Registration Statement. The Investor agrees promptly to furnish to the Company all information required to be disclosed in order to make the information previously furnished to the Company by the Investor not misleading. Any sale of any Registrable Securities by the Investor shall constitute a representation and warranty by the Investor that the information relating to the Investor and its plan of distribution is as set forth in the Prospectus delivered in connection with such disposition, that such Prospectus does not as of the time of such sale contain any untrue statement of a material fact provided by the Investor and that such Prospectus does not as of the time of such sale omit to state any material fact provided by the Investor necessary to make the statements in such Prospectus, in light of the circumstances under which they were made, not misleading. The Company may exclude from such Registration Statement the Registrable Securities of the Investor fails to furnish such information regarding to the Investor or its plan of distribution without the approval of the Investor in writing.

Notwithstanding any other provision of this Agreement, the Investor shall also provide the Company as a condition to including Registrable Securities in a Registration Statement, such information as is reasonably requested by the Company in response to the Company's customary questionnaire seeking the information required by the Securities Act and the rules and regulations promulgated thereunder.

(c) The Investor shall not use any free writing prospectus (as defined in Rule 405 under the Securities Act) in connection with the sale of Registrable Securities.

(d) Any single offering of Registrable Securities pursuant to any Shelf Registration Statement or any Demand Registration that is reasonably expected to result in aggregate cash proceeds in excess of One Hundred Million Dollars (\$100,000,000) shall be made pursuant to an underwritten offering. The Investor shall determine the managing underwriters for any offering initiated by the Investor, subject to the consent of the Company (which shall not be unreasonably withheld, conditioned or delayed). The Company shall determine the managing underwriters in any Piggyback Registration, subject to the consent of the Company (which shall not be unreasonably withheld, conditioned or delayed).

#### Section 4.06 Indemnification.

(a) The Company shall indemnify and hold harmless, to the fullest extent permitted by Law, (1) the Investor if the Registrable Securities are covered by a Registration Statement or Prospectus, (2) each of the Investor's Affiliates, officers, directors, shareholders, employees, advisors, agents, (3) each underwriter (including the Investor if deemed to be an underwriter pursuant to any SEC comments or policies), if any, and (4) each Person who controls (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) such underwriter (collectively, "Investor Indemnitees"), from and against all losses, claims, damages, liabilities, penalties, judgments, suits, costs and expenses (including reasonable legal fees and disbursements, which shall be reimbursed periodically as incurred) (collectively, "Losses") in connection with any sale of Registrable Securities pursuant to a Registration Statement under this Agreement arising out of or based upon (i) any untrue or alleged untrue statement of a material fact contained in any such Registration Statement or any Prospectus (including preliminary or final) relating to the registration of such Registrable Securities or any amendment or supplement thereto or any document incorporated by reference therein or any omission or (ii) or any alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading, and will reimburse to each of the Persons listed above, for any legal or any other expenses reasonably incurred in connection with investigating and defending any such Losses; provided, however, that the Company shall not be liable to such Investor Indemnitee in any such case to the extent that any such Loss, claim, damage, liability or expense arises out of or is based upon (A) an untrue statement or alleged untrue statement or omission or alleged omission made in such Registration Statement, including any such preliminary or final Prospectus contained therein or any such amendments or supplements thereto, or contained in any free writing prospectus (as such term is defined in Rule 405 under the Securities Act) prepared by the Company or authorized by it in writing for use by such Investor Indemnitee (or any amendment or supplement thereto), in reliance upon and in conformity with information regarding such Investor Indemnitee or its plan of distribution or ownership interests which was furnished in

writing to the Company expressly for use in connection with such Registration Statement, including any such preliminary or final Prospectus contained therein or any such amendments or supplements thereto (B) offers or sales effected by or on behalf of such Investor Indemnitee "by means of" (as defined in Rule 159A under the Securities Act) a "free writing prospectus" (as defined in Rule 405 under the Securities Act) or (C) the failure of any Investor Indemnitee to deliver or make available to a purchaser of Registrable Securities a copy of any Registration Statement, including any preliminary or final Prospectus contained therein or any amendments or supplements thereto (if the same was required by applicable Law to be delivered or made available); provided that the Company shall have delivered or made available to such Investor Indemnitee such Registration Statement, including such preliminary or final Prospectus contained therein and any amendments or supplements thereto.

(b) In connection with any Registration Statement in which the Investor is participating by registering Registrable Securities, the Investor agrees to indemnify and hold harmless, to the fullest extent permitted by Law, the Company, its Affiliates, the officers, directors, shareholders, advisors, agents, representatives or other employees of the Company, each Person who controls (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) the Company, each underwriter, if any, and each Person who controls (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) such underwriter (collectively, "Company Indemnitees"), from and against all Losses, as incurred, arising out of or based on any untrue or alleged untrue statement of a material fact contained in any such Registration Statement or preliminary or final Prospectus relating to the registration of such Registrable Securities or any amendment or supplement thereto or any document incorporated by reference therein, or any omission or alleged omission to state therein a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances in which they were made, not misleading, in each case solely to the extent that such untrue or alleged untrue statement or supplements therein or any such amendments or supplements thereto or contained in any free writing prospectus (as such term is defined in Rule 405 under the Securities Act) in reliance upon and in conformity with written information furnished to the Company by such Selling Investor expressly for inclusion in such document; provided, however, that in no event shall the liability of the Investor hereunder be greater in amount than the dollar amount of the net proceeds received by the Investor upon the sale of the Registrable Securities under the Registration Statement giving rise to such indemnification obligation.

(c) If any Person shall be entitled to indemnity hereunder (an "<u>Indemnified Party</u>"), such Indemnified Party shall give prompt notice to the party from which such indemnity is sought (the "<u>Indemnifying Party</u>") of any claim or of the commencement of any Action with respect to which such Indemnified Party has actual notice and seeks indemnification or contribution pursuant hereto; <u>provided</u>, <u>however</u>, that the delay or failure to so notify the Indemnifying Party shall not relieve the Indemnifying Party from any obligation or liability except to the extent that the Indemnifying Party has been actually prejudiced by such delay or failure. The Indemnifying Party shall have the right, exercisable by giving written notice (including an acknowledgement of its obligation to indemnify the Indemnified Party therefor on the terms set forth herein) to an Indemnified Party promptly after the receipt of written notice from such Indemnified Party of such claim or Action, to assume, at the Indemnifying Party's

expense, the defense of any such Action, with counsel reasonably satisfactory to such Indemnified Party; <u>provided</u>, <u>however</u>, that an Indemnified Party shall have the right to employ separate counsel in any such Action and to participate in the defense thereof, but the fees and expenses of such counsel shall be at the expense of such Indemnified Party unless: (i) the Indemnifying Party agrees to pay such fees and expenses; (ii) the Indemnifying Party fails promptly to assume, or in the event of a conflict of interest cannot assume, the defense of such Action or fails to employ counsel reasonably satisfactory to such Indemnified Party, in which case the Indemnified Party shall also have the right to employ counsel and to assume the defense of such Action or (iii) in the Indemnified Party's reasonable judgment a conflict of interest between such Indemnified Party and Indemnifying Party may exist in respect of such Action; <u>provided</u>, <u>further</u>, that the Indemnifying Party shall not, in connection with any one such Action or separate but substantially similar or related Actions in the same jurisdiction, arising out of the same general allegations or circumstances, be liable for the fees and expenses of more than one firm of attorneys (together with appropriate local counsel) at any time for all of the Indemnified Parties, or for fees and expenses that are not reasonable.

(d) Neither party shall settle, compromise, discharge or consent to an entry of judgment with respect to a claim or liability subject to indemnification under this <u>Section 4.06</u> without the other party's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed); <u>provided</u> that the Indemnifying Party may agree without the prior written consent of the Indemnified Party solely to any settlement, compromise, discharge or consent to an entry of judgment, in each case that relates only to money damages and by its terms obligates the Indemnifying Party to pay the full amount of the liability in connection with such claim and which unconditionally releases the Indemnified Party from all liability in connection with such claim.

(e) If the indemnification provided for in this Section 4.06 is unavailable to hold harmless each of the Indemnified Parties against any losses, claims, damages, liabilities and expenses to which such parties may become subject under the Securities Act, then the Indemnifying Party shall, in lieu of indemnifying each party entitled to indemnification hereunder, contribute to the amount paid or payable by such party as a result of such losses, claims, damages, liabilities or expenses in such proportion as is appropriate to reflect (i) the relative benefits received by the Indemnifying Party, on the one hand, and the Indemnified Parties, on the other hand, from the offering or (ii) if the allocation provided by clause (i) above is not permitted by applicable Law, in such proportion as is appropriate to reflect not only the relative benefits referred to in clause (i) above but also the relative fault of the Indemnifying Party, on the one hand, and such Indemnified Parties, on the other hand, in connection with the statements or omissions or alleged statements or omissions that resulted in such losses, claims, damages, liabilities or expenses. The relative fault of such parties shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact, or omission or alleged omission to state a material fact, relates to information supplied by or concerning the Indemnifying Party on the one hand, or by such Indemnified Party on the other, and such party's relative intent, knowledge, access to information and opportunity to have corrected or prevented such statement or omission. No Person guilty of fraudulent misrepresentation (within the meaning of the Securities Act) shall be entitled to contribution from any Person that is not guilty of such fraudulent misrepresentation.

#### Section 4.07 Miscellaneous.

(a) With a view to making available the benefits of certain rules and regulations of the SEC which may at any time permit the sale of the Registrable Securities to the public without registration, the Company agrees, so long as there are outstanding Registrable Securities, to use its commercially reasonable efforts to file with the SEC in a timely manner all reports and other documents as the SEC may prescribe under Section 13(a) or 15(d) of the Exchange Act at any time while the Company is subject to such reporting requirements of the Exchange Act; and

(b) Subject to the provisions hereof, in the event that the Company proposes to enter into an underwritten public offering, to the extent requested by the managing underwriters, and provided that the Company and all executive officers (as defined under the Exchange Act) and directors of the Company are also so bound, the Investor agrees to enter into a customary agreement with the managing underwriters not to effect any sale or distribution of equity securities of the Company, or any securities convertible, exchangeable or exercisable for or into such securities, without the consent of the managing underwriters, during the period beginning upon receipt of notice hereunder that the Company intends to conduct an offering of its securities in accordance with the terms hereof and ending ninety (90) Business Days following the effective date of such offering, except pursuant to such offering in accordance with the terms hereof; <u>provided</u>, <u>however</u>, that if any executive officer or director is released by such managing underwriters from its lockup obligations herein, then the Investor shall be so released on a pro rata basis (with the percentage of the Investor's Registrable Securities so released being equal to the percentage of shares so released for the executive officer or director having the highest percentage of released shares among all of the executive officers or directors). The Company may impose stop-transfer restrictions with respect to the securities subject to the foregoing restriction until the end of the required stand-off period and shall lift such stop-transfer restrictions immediately upon the end of such period.

(c) The registration rights granted to the Investor under this Agreement shall terminate on the date on which the Investor no longer owns Registrable Securities.

(d) Except for this Agreement, the Company is not party to any agreement granting any holder or prospective holder of any securities of the Company registration rights with respect to such securities. From and after the date hereof, the Company shall not, without the prior written consent of the Investor, enter into any agreement granting any holder or prospective holder of any securities of the Company registration rights with respect to such securities unless such new registration rights, including with respect to underwriters' "cutbacks" and "standoff" obligations, do not conflict with, the registration rights granted to Investor hereunder.

#### ARTICLE 5 TERMINATION

Section 5.01 **Termination**. Other than the termination provisions applicable to particular Sections of this Agreement that are specifically provided elsewhere in this Agreement, this Agreement shall terminate (a) at any time upon the mutual written agreement of the

Company and the Investor and (b) at such time as the Investor ceases to Beneficially Own any Registrable Securities.

## ARTICLE 6 MISCELLANEOUS

Section 6.01 <u>Amendment and Modification</u>. This Agreement may not be amended, modified or supplemented except by written agreement of the Company and the Investor.

Section 6.02 **Titles and Subtitles; Interpretation**. Unless otherwise indicated herein, with respect to any reference made in this Agreement to a Section (or Article, Subsection, Paragraph, Subparagraph or Clause), such reference shall be to a section (or article, subsection, paragraph, subparagraph or clause) of, or an exhibit or schedule to, this Agreement. The table of contents and any article, section, subsection, paragraph or subparagraph headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Any reference made in this Agreement to a statute or statutory provision shall mean such statute or statutory provision as it has been amended through the date as of which the particular portion of the Agreement, and to any then applicable rules or regulations promulgated thereunder. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed, as the context indicates, to be followed by the words "but (is/are) not limited to." The words "herein," "hereof," "hereunder" and words of like import shall refer to this Agreement as a whole, unless the context clearly indicates to the contrary. Words used herein, regardless of the number and gender specifically used, shall be deemed and construed to include any other number, singular or plural, and any other gender, masculine, feminine or neuter, as the context indicates is appropriate. Where specific language is used to clarify or illustrate by example a general statement contained herein, such specific language shall not be deemed to modify, limit or restrict the construction of the general statement which is being clarified or illustrated.

Section 6.03 **Extension; Waiver.** At any time prior to the Closing Date, a party to this Agreement may (a) extend the time for the performance of any of the obligations or acts of the other party, (b) waive any inaccuracies in the representations and warranties of the other party contained herein or in any document delivered pursuant hereto, (c) waive compliance with any of the agreements of the other party contained herein or (d) waive any condition to its obligations hereunder. Any agreement on the part of a party to any such extension or waiver shall be valid only if set forth in a written instrument signed by such party. Except as otherwise expressly provided herein, no failure to exercise, delay in exercising, or single or partial exercise of any right, power or remedy by any party, and no course of dealing among the parties, shall constitute a waiver of any such right, power or remedy.

Section 6.04 **<u>Binding Nature</u>**; <u>Assignment</u>. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned, Transferred or delegated by the Investor without prior written consent of the Company, and any attempt to make any such assignment, Transfer or delegation without such consent shall be null and void; <u>provided</u> that the Investor may assign its

#### registration rights under <u>ARTICLE 4</u> in connection with any Permitted Transfer under <u>Section 3.03(d)(i)</u>.

Section 6.05 <u>Severability</u>. Any provision of this Agreement that is invalid, illegal or unenforceable in any jurisdiction shall, as to that jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability, without affecting in any way the remaining provisions hereof in such jurisdiction or rendering that or any other provision of this Agreement invalid, illegal or unenforceable in any other jurisdiction.

Section 6.06 <u>Notices and Addresses</u>. All notices, requests, consents, waivers and other communications hereunder shall be in writing and shall be deemed given: (a) when delivered if delivered personally (including by courier); (b) on the third day after mailing, if mailed, postage prepaid, by registered or certified mail (return receipt requested); (c) on the day after mailing if sent by a nationally recognized overnight delivery service that maintains records of the time, place, and recipient of delivery; or (d) upon receipt of a confirmed transmission, if sent by telex, telecopy or facsimile transmission or e-mail, in each case to the other parties at the following addresses, facsimile numbers or e-mail addresses or to such other addresses as may be furnished in writing by one party to the others:

(a) if to the Investor to:

Hitachi, Ltd., Business Development Office 6-6 Marunouchi 1-chome Chiyoda-ku Tokyo 100-8280, Japan Attention: General Manager Facsimile: 81-3-4564-6260

with a copy (which shall not constitute notice) to:

Morrison & Foerster LLP Shin-Marunouchi Building, 29<sup>th</sup> Floor 5-1, Marunouchi 1-chome Chiyoda-ku, Tokyo 100-6529 Japan Attention: Kenneth A. Siegel, Esq. Facsimile: 011-81-3-3214-6512 E-mail: <u>KSiegel@mofo.com</u>

(b) if to the Company:

Western Digital Corporation 3355 Michelson Drive, Suite 100 Irvine, California 92612 Attention: General Counsel Facsimile: (949) 672-9612 E-mail: <u>Michael.Ray@wdc.com</u>

with a copy (which shall not constitute notice) to:

O'Melveny & Myers LLP 610 Newport Center Drive, Suite 1700 Newport Beach, California 92660 Attention: J. Jay Herron, Esq. Facsimile: (949) 823-6994 E-mail: jherron@omm.com

Section 6.07 <u>Governing Law</u>. This Agreement and the legal relations among the Parties shall be governed by and construed in accordance with the Laws of the State of Delaware without giving effect to any Law or rule that would cause the Laws of any jurisdiction other than the State of Delaware to be applied.

Section 6.08 <u>**Complete Agreement</u>**. This document and the documents referred to herein contain the complete agreement between the parties and supersede any prior understandings, agreements or representations by or between the parties, written or oral, which may have related to the subject matter hereof in any way.</u>

Section 6.09 <u>No Third-Party Beneficiaries</u>. This Agreement is intended and agreed to be solely for the benefit of the parties hereto, and no third party shall accrue any benefit, claim or right of any kind whatsoever pursuant to, under, by or through this Agreement, except as otherwise contemplated by <u>Section 6.04</u>.

Section 6.10 <u>Counterparts and Signatures</u>. This Agreement may be executed in several counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument.

Section 6.11 **<u>Further Assurances</u>**. Each party shall cooperate and take such action as may be reasonably requested by another party in order to carry out the provisions and purposes of this Agreement and the transactions contemplated hereby.

Section 6.12 <u>Specific Performance</u>. The parties acknowledge and agree that irreparable damage would be caused in the event any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached and money

damages may not be an adequate remedy for any such failure to perform or breach. Accordingly, the parties agree that, in addition to any other remedy to which each party may be entitled at Law or in equity or under this Agreement, each shall be entitled to an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof, and each party expressly waives the defense that a remedy in damages will be adequate.

Section 6.13 <u>Consent to Jurisdiction; Service of Process; Venue</u>. Each of the parties to this Agreement irrevocably and unconditionally submits to the exclusive jurisdiction of the Delaware Court of Chancery (and if jurisdiction in the Delaware Court of Chancery shall be unavailable, any Delaware State court and the Federal court of the United States of America sitting in the State of Delaware) for the purposes of any action or other proceeding arising out of this Agreement or any transaction contemplated hereby. Each of the parties further agree that, to the fullest extent permitted by applicable Law, service of any process, summons, notice or document by U.S. registered mail to such Person's respective address set forth in <u>Section 6.06</u> above shall be effective service of process for any action or proceeding in the State of Delaware with respect to any matters to which it has submitted to jurisdiction as set forth above in the immediately preceding sentence. Each of the parties irrevocably and unconditionally waives (and agrees not to plead or claim), any objection to the laying of venue of any action or proceeding arising out of this Agreement or the transactions contemplated hereby in the Delaware Court of Chancery (and if the Delaware Court of Chancery shall be unavailable, in any Delaware State court or the Federal court of the United States of America sitting in the State of Delaware) or that any such action or proceeding brought in any such court has been brought in an inconvenient forum.

Section 6.14 <u>Waiver of Jury Trial</u>. Each of the parties to this Agreement hereby waives, to the fullest extent permitted by applicable Law, any right it may have to a trial by jury in respect of any suit, action or other proceeding directly or indirectly arising out of, under or in connection with this Agreement. Each party (a) certifies that no representative, agent or attorney of any other party has represented, expressly or otherwise, that such party would not, in the event of any action or proceeding, seek to enforce the foregoing waiver and (b) acknowledges that it and the other parties have been induced to enter into this Agreement, by, among other things, the mutual waiver and certifications in this <u>Section 6.14</u>.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the parties hereto caused this Agreement to be duly executed by their respective authorized officers on the day and year first above written.

Western Digital Corporation

By: /s/ Michael C. Ray

Name: Michael C. Ray Title: Senior Vice President, General Counsel and Secretary

Hitachi, Ltd.

By: /s/ Hiroaki Nakanishi

Name: Hiroaki Nakanishi Title: Representative Executive Officer President

S-1